

### **Bangor University**

DOCTOR OF PHILOSOPHY

Climate change and Welsh catchments: Implications for hydrological regime, water quality and water abstraction

Dallison, Richard

Award date: 2021

Awarding institution: Bangor University

Link to publication

#### **General rights**

Copyright and moral rights for the publications made accessible in the public portal are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognise and abide by the legal requirements associated with these rights.

• Users may download and print one copy of any publication from the public portal for the purpose of private study or research.

You may not further distribute the material or use it for any profit-making activity or commercial gain
You may freely distribute the URL identifying the publication in the public portal ?

Take down policy

If you believe that this document breaches copyright please contact us providing details, and we will remove access to the work immediately and investigate your claim.



**Bangor University** 

DOCTOR OF PHILOSOPHY

Climate change and Welsh catchments: Implications for hydrological regime, water quality and water abstraction

Dallison, Richard

Award date: 2021

Awarding institution: Bangor University

Link to publication

#### General rights

Copyright and moral rights for the publications made accessible in the public portal are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognise and abide by the legal requirements associated with these rights.

• Users may download and print one copy of any publication from the public portal for the purpose of private study or research.

You may not further distribute the material or use it for any profit-making activity or commercial gain
You may freely distribute the URL identifying the publication in the public portal ?

Take down policy If you believe that this document breaches copyright please contact us providing details, and we will remove access to the work immediately and investigate your claim.



# Climate change and Welsh catchments: Implications for hydrological regime, water quality and water abstraction

A thesis submitted to Bangor University by Richard John Harvey Dallison

In candidature for the degree of **Doctor of Philosophy** 

Supervised by Dr. Sopan Patil & Dr. Prysor Williams School of Natural Sciences, Bangor University

Submitted: 27<sup>th</sup> October 2020 | Defended: 10<sup>th</sup> December 2020 Corrected: 8<sup>th</sup> March 2021 | Accepted: 10<sup>th</sup> March 2021

### Research completed as part of the Dŵr Uisce Project

Funded by the European Regional Development Fund (ERDF) through the Interreg Ireland-Wales Co-operation Programme 2014-2020.







### ACKNOWLEDGEMENTS

#### Dedicated to the memory of my sister, Rachael Dallison

"Time flies. Suns rise and shadows fall. Let time go by. Love is forever"

Firstly, I would like to thank all of the staff in the School of Natural Sciences who have helped to shape myself and the research contained within this thesis, during my time as a student at Bangor. I am particularly deeply indebted to both Dr Sopan Patil and Dr Prysor Williams for their continued supervision, support, guidance, and patience during the PhD process. Sopan, firstly thank you for encouraging me to undertake this opportunity to begin with, and for your continued encouragement throughout. Without your expertise in hydrology and hydrological modelling, the thesis quite simply would not be complete. Furthermore, your championing and support of opportunities such as attendance and participation at conferences and workshops, as well as paper submissions to journals, has benefitted both me and the research immensely. Prysor, your valuable advice, constructive feedback, continued encouragement, and especially keen eye for detail, have all helped to improve the research completed and communicated hugely. Thank you both especially for your time in reading the steady stream of paper and chapter drafts in the final few months of the PhD process, your feedback really has been invaluable. Additionally, without the efforts of Mr Ian Harris in sourcing further computing power, this research would not have been completed on time. I am therefore very grateful for both this and you support, advice and expertise during all of my time at Bangor.

The whole Dŵr Uisce team, both past and present, in Ireland and in Wales, have all also had a hand in shaping this work, and I am grateful for every comment, question and discussion over the course of the research, I look forward to continuing to work with you all. I must also acknowledge the European Regional Development Fund through the Interreg Ireland-Wales Co-operation Programme for the funding of this research. I am also very appreciative of Dŵr Cymru Welsh Water for the provision of both data and knowledge for this research, as well as all other organisation that have provided data.

The PhD process would also have been very different without the support, kindness, laughter, and camaraderie of many fantastic friends, to all of whom I am very grateful. A special thank you though, must go especially to Rebecca Jones, Rob Brown, Luke Prosser, and Luke Tyler, without whom the PhD process would have been very lonely. Rebecca, without you in particular I most likely would not have reached this stage, so I will be eternally grateful to you.

Finally, thank you to my Mum who has always encouraged me to follow my ambitions and aspirations and believed in me to achieve anything. Your continued love and support has kept me going through difficult times and always helps me to find the best route forward. I would not be where I am today without you and your unwavering belief in me.

### **EXECUTIVE SUMMARY**

Historical and future climate change is likely to be one of the key drivers of alterations seen in catchments globally. The implications of changing precipitation patterns in particular range from increased seasonality of flows, to changing frequency, duration, and severity of extreme events, to variations in pollutant sourcing and transport. These changes display regional differences, but it is also reasonable to expect catchment level variation. Factors such as topography, land use/land cover and soil type all cause variation of regional trends. Such alterations could have consequences for all water users, but flow regime changes in particular will directly impact those abstracting water from rivers. In this work, hydrological regime change is investigated for five catchments in Wales (Clwyd, Conwy, Dyfi, Teifi, Tywi), both historically and under a future worst-case scenario of global greenhouse gas emissions. For the future period water quality is also considered. Subsequently, impacts on abstraction for public water supply (PWS) and hydroelectric power (HEP) are also quantified.

The first study investigates historical trends in average and extreme precipitation, air temperature and streamflow, before correlating these factors with actual total abstraction data for PWS (as a proxy for overall water demand). A strong warming trend in autumn average temperatures in all catchments is observed (Sen's slope range: 0.38-0.41, p <0.05), with a north-south divide detected in streamflow and extreme temperature trends. A positive correlation between abstraction volume and temperature in four catchments is established (Spearman's  $\rho$  range: 0.094-0.403, p <0.01; Pearson's r range 0.073-0.369, p <0.01). The study provides new insight into the relationship between hydroclimatic factors and total PWS demand and highlights local variation in broader regional changes.

The second study uses the Soil and Water Assessment Tool (SWAT) to model future average and extreme streamflow and water quality. The 12-model ensemble UK Climate Projections 2018 dataset, based on Representative Concentration Pathway 8.5 (RCP8.5) conditions, is used for future daily weather input (2021-2080). A small decline in mean annual flow is observed (-4% to -13%), with much larger variations seasonally. Spring and autumn streamflows are most affected, increasing and decreasing by up to 41% and 52%, respectively; these changes corresponded with increased frequency of high flow events in spring, and low flow events in autumn. Water quality generally declines in all catchments and seasons, with increased concentrations of suspended sediment, nitrogen, and phosphorous, and decreased dissolved oxygen. Results suggest a double threat for water abstractors, especially for PWS, of both declining flows and water quality, in summer and autumn in particular.

The implications of projected hydrological regime changes are investigated in the third study, with a focus on PWS and HEP. A key PWS abstraction in the Tywi is studied using the Water Evaluation And Planning (WEAP) system. The temperature-demand relationship established in the first study is used to project future total water demand under RCP8.5 conditions; static and reducing demand scenarios are also considered. Under all scenarios there is an increase in occurrences of insufficient streamflow to satisfy demand. Alterations in HEP operations are analysed for sixteen sites in the Conwy, and nine in the Tywi. Results suggest a decrease in total annual abstraction volume, resulting in a loss of generation potential, this is especially pronounced up to the mid-2050s. Further work is needed to quantify the financial and environmental implications of observed trends; additional work on water quality changes would also strengthen PWS planning.

## CONTENTS

DECLAF	RATION AND CONSENT
ACKNO	WLEDGEMENTS i
EXECUT	۲IVE SUMMARY
LIST OF	FIGURES i
LIST OF	TABLES xi
ABBRE	VIATIONSxi
СНАРТИ	ER 1: INTRODUCTION
1.1	Study context
1.2	Study justification
1.3	Research aim and objectives
1.4	Thesis structure
1.5	The Dŵr Uisce Project10
СНАРТІ	ER 2: LITERATURE REVIEW1
2.1	Global warming and climatic change1
2.1	.1 Historical changes
2.1	.2 Future projections
2.2	Climatic trends in the UK and Wales20
2.2	.1 Historical changes
2.2	.2 Future projections
2.3	Changing hydrology in the UK and Wales2
2.3	.1 Historical hydrological trends
2.3	.2 Future projections under climate change
2.4	Water quality implications of hydroclimatic change
2.5	Future water resource use implications for Wales
2.6	Summary40
СНАРТІ	ER 3: METHODOLOGICAL OVERVIEW4
3.1	Overarching methodology4
3.2	Study catchments
3.3	Trend analysis
3.4	Study 1: Historical analysis
3.4	.1 Hydroclimatic data
3.4	.2 Hydroclimatic analysis
3.4	.3 Water abstraction relationships
3.5	Study 2: Hydrological modelling5

3.5	1 Model selection	
3.5	2 Model setup, calibration and validation	61
3.5	3 Future model runs	65
3.5	4 Data analysis	
3.6	Study 3: Future water availability analysis	69
3.6	1 Study area and analysis framework	
3.6	2 Hydroelectric power	
3.6	3 Public water supply	71
СНАРТ	R 4: STUDY I. Influence of historical climate patterns on st	reamflow and
water d	emand in Wales, UK	74
4.1	Abstract	75
4.2	Introduction	76
4.3	Data and methods	79
4.3	1 Study catchments	79
4.3	2 Data	
4.3	3 Trend analysis	
4.3	4 Correlation analysis of hydroclimatic and abstraction data	
4.4	Results	84
4.4	1 Trend and breakpoint analysis	
4.4.	2 Correlation analysis of hydroclimatic and abstraction data	
4.5	Discussion	92
4.5	1 Hydroclimatic trends	92
4.5	2 Weather and demand for water	96
4.5	3 Study implications	
4.6	Conclusions	98
СНАРТ	R 5: STUDY II. Modelling the impact of future climate change	on streamflow
and wa	ter quality in Wales, UK	100
5.1	Abstract	101
5.2	Introduction	
5.3	Data and methods	
5.3	1 Study catchments	
5.3	2 Future climate change data	
5.3	3 Hydrological modelling	
5.3	4 Trend analysis	
5.4	Results	116
5.4	1 Climate projections	
5.4	2 Streamflow	
5.4	3 Water quality	

5.5	Discussion	
5.5	5.1 Streamflow trends	
5.5	5.2 Water quality trends	
5.5	5.3 Study implications	
5.6	Conclusions	
СНАРТ	ER 6: STUDY III. Impacts of climate change on	future water availability for
hydrop	power and public water supply in Wales, UK	
6.1	Abstract	
6.2	Introduction	
6.3	Data and methods	
6.3	3.1 Study catchments	
6.3	3.2 Future streamflow and climate projections	
6.3	3.3 Hydroelectric power assessment	
6.3	3.4 Public water supply assessment	141
6.3	3.5 Trend analysis	
6.4	Results	
6.4	4.1 Hydroelectric power	144
6.4	Public water supply	
6.5	Discussion	
6.5	5.1 Hydroelectric power	
6.5	5.2 Public water supply	
6.5	5.3 Study implications	
6.6	Conclusions	
СНАРТ	ER 7: DISCUSSION	
7.1	Hydroclimatic trends and implications	
7.2	Future water quality implications for abstractors	
7.3	Summary	
СНАРТ	ER 8: CONCLUSIONS	
8.1	Final conclusions	
8.2	Recommendations for further work	
8.3	Final summary	
REFER	ENCES	
APPEN	IDICES	
Appe	endix 1: SWAT model calibration charts	
Арре	endix 2: Ensemble projections for seasonal and annua	al streamflow204

### LIST OF FIGURES

Figure 1.1. UK annual average rainfall for the period 1981-2010 (Met Office, 2020)6
Figure 1.2. Thesis study catchments7

Figure 2.1. Historical atmospheric concentrations of CO <sub>2</sub> , CH <sub>4</sub> and N <sub>2</sub> O (IPCC, 2013)12
Figure 2.2. Observed and projected (2017-2100) global mean surface temperature. Adapted from IPCC (2018)
Figure 2.3. Historical over-land annual precipitation anomalies (IPCC, 2013)14
Figure 2.4. CO <sub>2</sub> projetions for each Representative Concentration Pathway (IPCC, 2013)18
Figure 2.5. Projections of future global average surface temperature (IPCC, 2013)19
Figure 2.6. Projections of future UK average temperature (Lowe et al., 2018)24
Figure 2.7. Projections of future UK average precipitation (Lowe et al., 2018)24
Figure 2.8. UK abstraction demand in the 2050s and 2080s as a percentage of available discharge during low flow conditions (ASC, 2016b)

Figure 4.1. Study I catchments showing, (A) stream order and gauging station locations, (B) elevation, and (C) land use/land cover
Figure 4.2. Warming trend in historical average autumn temperatures in all catchments85
Figure 4.3. Declining histrocal trend in the frequency of the hotest days annually for the Dyfi, Teifi and Tywi
Figure 4.4. Declining histrocal trend in the frequency of the coldest summer days for the Dyfi, Teifi and Tywi
Figure 4.5. Historcal annual average streamflows for all study catchments
<b>Figure 4.6.</b> Normalised daily hydroclimatic data compared with normalised daily water supply abstraction volume for January 2012 to September 2015 <b>91</b>

<b>Figure 4.7.</b> Normalised monthly hydroclimatic data compared with normalised monthly water supply abstraction volume for January 2012 to September 2015 <b>92</b>
<b>Figure 4.8.</b> Annual North Atlantic Oscillation index 1825-2019 relative to the 1951-1980 average index value, adapted from Osborn (2011) <b>95</b>
<b>Figure 5.1.</b> Study II catchments showing, (A) stream order, gauging stations and water quality monitoring locations, and (B) catchment elevation <b>107</b>
Figure 5.2. Study II catchments showing, (A) land use/land cover, and (B) soil types108
Figure 5.3. Pseudo-meteorological stations used in SWAT future weather input110
Figure 5.4. Future projections (2021-2080) for seasonal and annual daily mean precipitation and temperature
Figure 5.5. Seasonal and annual average percentage change from 1990-2010 baseline, for a 5-year moving average of future (2021-2080) streamflow119
<b>Figure 5.6.</b> Overview of the direction, magnitude and significance of annual and seasonal trends in projected future (2021-2080) average streamflow <b>120</b>
<b>Figure 5.7.</b> Overview of the direction, magnitude and significance of annual and seasonal trends in projected future (2021-2080) extreme streamflows <b>120</b>
<b>Figure 5.8.</b> Flow duration curve comparison of 1990-2010 and 2060-2080 for, (A) Conwy and Dyfi catchments and, (B) Clwyd, Teifi and Tywi catchment <b>122</b>
<b>Figure 5.9.</b> Overview of the direction, magnitude and significance of annual and seasonal trends in projected future (2021-2080) water quality concentrations
Figure 6.1. Study III catchments, showing land use/land cover classification
<b>Figure 6.2.</b> Projections for future seasonal and annual mean temperature, precipitation and streamflow for the Conwy and Tywi, as well as percentage change from 1990-2010 baseline for a 5-year moving average of future streamflow <b>139</b>
Figure 6.3. Studied abstraction locations for hydroelectric power and public water supply; stream order and catchment elevation also shown
<b>Figure 6.4.</b> Future monthly total water demand at the Tywi public water supply abstraction under three future water demand scenarios: increased, static and decreased <b>143</b>
<b>Figure 6.5.</b> Overview map of the direction, magnitude and significance of annual trends in number of days A <sub>start</sub> achieved for the periods 2021-2079 and 2021-2054 <b>145</b>
<b>Figure 6.6.</b> Overview map of the direction, magnitude and significance of annual trends in number of days A <sub>max</sub> reached for the periods 2021-2079 and 2021-2054 <b>146</b>
<b>Figure 6.7.</b> Overview map of the direction, magnitude and significance of annual trends in mean A <sub>daily</sub> for the periods 2021-2079 and 2021-2054 <b>147</b>
<b>Figure 6.8.</b> Overview map of the direction, magnitude and significance of annual trends in total abstraction for the periods 2021-2079 and 2021-2054 <b>148</b>

**Figure 6.9.** Comparison of the direction, magnitude and significance of annual and seasonal trends in future hydroelectric power characteristics for the period 2021-2079. Based on the average of all Zone 2 (Z2) and Zone 3 (Z3) abstractions in each catchment......**149** 

Figure A1.1. Observed and simulated streamflow for Clwyd catchment calibration2	202
Figure A1.2. Observed and simulated streamflow for Conwy catchment calibration2	202
Figure A1.3. Observed and simulated streamflow for Dyfi catchment calibration	202
Figure A1.4. Observed and simulated streamflow for Teifi catchment calibration	203
Figure A1.5. Observed and simulated streamflow for Tywi catchment calibration	203

Figure A2.1. Ensemble projections of seasonal and annual streamflow for the Clwyd204
Figure A2.2. Ensemble projections of seasonal and annual streamflow for the Conwy205
Figure A2.3. Ensemble projections of seasonal and annual streamflow for the Dyfi206
Figure A2.4. Ensemble projections of seasonal and annual streamflow for the Teifi207
Figure A2.5. Ensemble projections of seasonal and annual streamflow for the Tywi208

### LIST OF TABLES

<b>Table 2.1.</b> Overview of the characteristics of the four Representative Concentration Pathways(RCPs), adapted from Moss <i>et al.</i> (2010) <b>17</b>
<b>Table 2.2.</b> Breakdown of reasons for Welsh streams not achieving good ecological status inthe Water Framework Directive classifications (NRW, 2019a)33
Table 3.1. Summary of comparative study catchment details
Table 3.2.         Number of abstraction locations per use descriptor for each catchment
Table 3.3. Details of gauging stations used throughout the thesis (NRFA, 2020)
Table 3.4. SWAT calibration parameters    62
Table 3.5. Kling-Gupta efficiency calibration and validation statistics for streamflow in each catchment
Table 3.6. Water quality sample details for SWAT validation         65
Table 3.7. Catchment SWAT model characteristics and run period information
Table 4.1. Key Study I catchments details       81
Table 4.2. Mann-Kendall trend analysis and Sen's slope indicator results for average autumnair temperatures trends (1981-2015)
Table 4.3. Pettitt and Standard Normal Homogeneity Test breakpoint analysis statistics for average autumn temperatures
Table 4.4. Signifcant Mann-Kendall trend analysis and Sen's slope indicator results forextreme temperature events (1982-2015)
Table 4.5. Signifcant Mann-Kendall trend analysis and Sen's slope indicator results forextreme streamflow events (1982-2015)
<b>Table 4.6.</b> Pearson's and Spearman's rank correlation coefficient results for daily and monthlyabstraction data with hydroclimatic factors (2012-2015)
Table 5.1. Key Study II catchments details         109
Table 5.2.         Catchment SWAT model details, with calibration and validation statistics113
Table 5.3. SWAT calibration parameters         113
Table 5.4.         Pearson's correlation values between observed and modelled water quality concentrations           114
Table 5.5. Pearson's correlation values between modelled water quality concentrations and streamflow         114

Table 6.1.         Key Study III catchments details         1	37
Table 6.2.         Abstraction conditions as defined by NRW guidelines for Zone 2 and Zone 3 site	tes
(NRW, 2020b) <b>1</b>	41

### **ABBREVIATIONS**

AR5	Fifth Assessment Report (by the IPCC)	NIC	National Infrastructure Committee
ASC	Adaptation Sub-Committee	NRFA	National River Flow Archive
	(of the CCC)	NRW	Natural Resources Wales
CCC	Committee on Climate Change	PPE	Perturbed parameter ensemble
СЕН	Centre for Ecology & Hydrology	PSO	Particle Swarm Optimisation
CHESS	Climate, Hydrology & Ecology research Support System	PWS RCM	Public water supply Regional climate model
CMIP5	5 <sup>th</sup> phase Coupled Model Intercomparison Project	RCP	Representative Concentration Pathway
DCWW	Dŵr Cymru Welsh Water	SDGs	Sustainable Development Goals (from the UN)
DEM DO	Digital elevation model Dissolved oxygen	SNHT	Standard Normal Homogeneity Test
EEA	European Environment Agency	SS	Suspended sediment
EU	European Union	SWAT	Soil and Water Assessment Tool (hydrological model)
FDC	Flow duration curve	SWAT-CUP	SWAT Calibration and
FFH	Future Flows Hydrology		Uncertainty Programme
с;т	(ualasel)	TN	Total nitrogen
	Clobal climate model	ТР	Total phosphorus
		UK	United Kingdom
НЕР	Hydroelectric power	UKCIP02	UK Climate Impacts Programme 2002
HoF	Hands-off-flow	UKCP09	UK Climate Projections 2009
HRU	Hydrological Response Unit	UKCP18	UK Climate Projections 2018
IPCC	Intergovernmental Panel on	UN	United Nations
KGE	Kling-Gupta efficiency	WEAP	Water Evaluation And Planning system
LULC	Land use/land cover	WMO	World Meteorological
МК	Mann-Kendall (trend analysis)		Organisation
монс	Met Office Hadley Centre	WSP	Water service provider
NAO	North Atlantic Oscillation	Z2 / Z3	Zone 2 / Zone 3 (HEP sites)

# **CHAPTER 1: INTRODUCTION**

### 1.1 Study context

Global climatic change, chiefly caused by anthropogenic inputs of greenhouse gases (GHGs) to the atmosphere (IPCC, 2013), has fundamentally altered the hydrological cycle in various ways (IPCC, 2014a). Changing precipitation patterns, altered snowmelt timing and speed, and increases in extreme weather events have all caused variation in the hydrological regime of river systems (IPCC, 2014a). Such alterations have had numerous impacts on river systems across the globe, such as changes to ecology/river health (Durance & Ormerod, 2007; Lynch et al., 2016; Stagl & Hattermann, 2016; Bussi et al., 2018), variation in flood characteristics (Hattermann et al., 2013; Kundzewicz et al., 2018; Langhammer & Bernsteinová, 2020; Tramblay et al., 2020), thermal regime changes (Isaak et al., 2012; Arora et al., 2016; Islam et al., 2019; Jin et al., 2020), and riverbed morphology alteration (Rumsby & Macklin, 1994; Ashmore & Church, 2001; Slater & Singer, 2013; Slater et al., 2019). In terms of human systems, one of the most important consequences is for change in available water resource (Vörösmarty, 2000; Arnell, 2004a; Alcamo et al., 2007; Konapala et al., 2020), which can be abstracted for uses such as public water supply (PWS), hydroelectric power (HEP), agriculture, and heavy industry. Water resource availability globally is undoubtably under pressure from various factors, including urbanisation (Astaraie-Imani et al., 2012; McGrane, 2016), agricultural intensification (Watts et al., 2015b; Holden et al., 2017), and population increase (Vörösmarty et al., 2010; Staddon & Scott, 2018). However, climate change and the impacts that increased global temperatures and changing precipitation patterns will have on water availability is arguably one of the most challenging.

Climate change not only has the ability to impact on the supply of water, but is also likely to have an impact on demand for water. Domestically, increased air temperature, humidity and sunshine hours can increase water use in the UK, especially for purposes such as bathing and garden watering (Downing *et al.*, 2003; Goodchild, 2003; Green & Weatherhead, 2014; Xenochristou *et al.*, 2020). Furthermore, likely increases in temperature and decreases in rainfall have the potential to generate much greater demand for water from the agricultural sector, for irrigation of crops and water supply for animals (Downing *et al.*, 2003; Knox *et al.*, 2010; Rio *et al.*, 2018; Hess *et al.*, 2020). This creates a two-part threat to future water security and water resource availability.

Another concern for catchments under future climate change is the impact on water quality, which could have far-reaching consequences for PWS (Wilby *et al.*, 2006; Whitehead *et al.*, 2009b; Mo *et al.*, 2016). Water quality is expected to decline because of alterations in flow regime, in particular extreme flows, as well due to changing precipitation patterns (Watts *et al.*, 2015a; Sayers *et al.*, 2016; Collet *et al.*, 2018; Rau *et al.*, 2020). Increasing discharge volumes cause both greater erosion of the riverbed and banks, as well as increasing the transport capacity of the flow (Delpla *et al.*, 2009; Mortazavi-Naeini *et al.*, 2019). Furthermore, intensification of extreme precipitation and flood events also allows for greater washing of pollutants from the land surface to streams, increasing pollutant concentrations (Delpla *et al.*, 2009; Whitehead *et al.*, 2009b; Ockenden *et al.*, 2017; Hashempour *et al.*, 2020). In addition, increases in the intensity and frequency of low flow events gives less opportunity for dilution of pollutants; this is particularly problematic downstream of point-sources of pollution (Delpla *et al.*, 2009; Watts & Anderson, 2016; Miller & Hutchins, 2017).

In Wales, it is expected that future climate change will take the form of warmer temperatures year-round, with exaggerated seasonality of precipitation when compared to current conditions, meaning drier summers and wetter winters (Welsh Government, 2013; ASC, 2016a; Lowe *et al.*, 2018). A greater number of extreme weather events are projected, with more frequent and intense precipitation events (Welsh Government, 2013; ASC, 2016a; Lowe *et al.*, 2018). These changes have the potential to substantially impact on the availability of surface water resources in the country, a resource that is highly relied upon in Wales for several sectors, in particular PWS (DCWW, 2019a). Furthermore, changes in water quality could also have significant impacts on PWS as well as create further challenges for many abstractors. Changes in quality of raw water entering drinking water treatment plants could prove costly, both financially and in terms of energy use, with water potentially requiring additional treatment (Mo *et al.*, 2016; Valdivia-Garcia *et al.*, 2019). As has been demonstrated, it is therefore important to have a thorough understanding of future hydrological regime characteristics under a changing climate for Wales, in order to best plan and adapt for changing water resource availability and quality.

Such climate change adaptation planning is in fact mandated in Wales, with the Climate Change Act (2008) requiring Welsh Ministers to regularly produce a report on their objectives, actions, and priorities in terms of the impact of climate change for Wales. At the time of its

enactment, the Act set a target of an 80% reduction in GHG emissions compared to 1990 levels by 2050, however, in 2019 a new, more ambitious net zero emissions target was set. The Act also established the Committee on Climate Change (CCC), a statuary body with the purpose of advising the UK and devolved governments on progress towards emission targets and climate change adaptation (CCC, 2020). The Adaptation Sub-Committee (ASC) of the CCC fulfils the requirement of the Climate Change Act (2008) to prepare a climate change risk assessment every five years, the next to be released in June 2021 (Welsh Government, 2019a). Additionally, the Environment Act (Wales) 2016 aims to position Wales as ready to adapt to climate change as part of a low carbon economy. Parts 1 and 2 of the Act specifically relate to sustainable natural resources management, and adaptation to climate change respectively.

The Water Industry Act (1991) identifies the main duties for water service providers in the UK and provides the framework for water services regulation, a role currently performed by Ofwat. Furthermore, the Water Resources Act (1991) developed the environmental regulation and oversight relating to water supply, currently under the jurisdiction of Natural Resources Wales. Additionally, the Water Resources Management Plan Regulations (2007) amended the Water Industry Act (1991) to make it a statuary obligation for water providers in England and Wales to produce a water resource management plan every five years, the latest being released in 2019. Any measures undertaken must ensure compliance with the European Union's (EU) Water Framework Directive in terms of surface water quality (although the future of this Directive is uncertain given the UKs exit from the EU) and the Water Supply (Water Quality) Regulations (Wales) 2018, in terms of drinking water supply standards.

Furthermore, the Wellbeing of Future Generations (Wales) Act 2015 requires the Welsh Government and public bodies in Wales to incorporate long-term thinking into decision making, particularly around seven wellbeing goals set out in the Act. This thesis directly addresses three of these goals, in particular, (1) a prosperous Wales, (2) a resilient Wales, and (3) a globally responsible Wales. In addition, of global relevance to this research are the United Nations' (UN) seventeen Sustainable Development Goals (SDGs; UN, 2015). No less than nine bear relevance to this work (SDGs 3, 6, 7, 8, 9, 11, 12, 13, 14); with Goal 6 (ensure availability and sustainable management of water and sanitation for all), and Goal 13 (take urgent action to combat climate change and its impacts) being the most relevant.

### **1.2 Study justification**

Wales is often viewed as a country with an abundant supply of water and higher than average precipitation when compared to the rest of the UK, indeed for some areas this is true, annual rainfall volumes in the north of the country being some of the highest in the UK (Figure 1.1; MET Office, 2020). However, in terms of PWS, the main water supplier Dŵr Cymru Welsh Water (DCWW) states that there are already regions at risk of seasonal water deficit in Wales (DCWW, 2019a), as well as areas that will face seasonal water scarcity problems in the future unless measures are put in place soon to ensure continued supply (DCWW, 2019b). This vulnerability comes from the heavy reliance on surface waters in the region, with little abstraction from groundwater as a back-up. Importantly, Wales is also an exporter of water to other areas of the UK, for example, up to 360 million litres of water per day are piped 117 km from the Elan Valley reservoirs in mid Wales to supply the Birmingham metropolitan area (Warren & Holman, 2012; DCWW, 2016a). In addition, Lake Vyrnwy in north Powys, along with the River Dee, supply over 230 million litres of water per day to Merseyside and Cheshire (United Utilities, 2018). Lake Vyrnwy is also used to regulated flow in the River Severn in times of drought, ensuring supplies for several water companies in the midlands and west of England (United Utilities, 2018). Recent plans also include the possibility of diverting water from the lake via the Severn system to supply water to London and the southeast of England, by the latter half of this century (Atkins et al., 2016). Although these two key systems do not fall within the catchments studied in this thesis, the results of this work are still relevant to them and the supplies of water to the reservoirs. This current and potentially even greater future reliance of millions of people on the catchments of Wales for drinking water supply alone, makes the country an important region to study. In addition to the heavy reliance of PWS on streamflow, the rivers of Wales also support various other sectors from the important tourism and leisure industry, to heavy industry, to agriculture and to both large- and smallscale HEP generation. HEP in particular, although relatively small in the electricity generation portfolio of Wales, is a sector that has been growing in recent years with over 300 new schemes implemented since 2002, from a starting point of approximately 40 (Welsh Government, 2019b). Despite this substantial increase in the number of new schemes, the installed capacity has only risen by 20 MW from a 160 MW baseline in 2002 (Welsh Government, 2019b). This disparity points to the nature of the schemes developed, the

majority being small-scale; these types of schemes are considerably more vulnerable to changing streamflow regimes. While changes to Feed-in-Tariffs (the rate paid for electricity generated through renewable technologies) and business rates in the last two years have seen a decline in in the installation of new schemes (Welsh Government, 2019b), it is still important to understand the performance of installed schemes under future streamflow conditions to enable robust planning in the energy sector to ensure a stable electrical supply.



Figure 1.1. UK annual average rainfall for the period 1981-2010 (Met Office, 2020).

As a result of the reliance on surface waters by various sectors in Wales, as well as the impact of individual catchment characteristics on the response of a watershed to precipitation, understanding future changes in the timing, quantity and quality of streamflows at a catchment scale is crucial. Five catchments in Wales have been selected for study, The Clwyd, Conwy, Dyfi, Teifi, and Tywi (Figure 1.2). These five have specifically been chosen to reflect the wide variety of catchment characteristics displayed across Wales, allowing for

comparisons of responses to different conditions to be made. The catchments differ in terms of topography, land use/land cover proportions, size, and location. All of these factors have the potential to influence how future climate change will impact of streamflows and water quality. Furthermore, these catchments provide water for a population in excess of 975,000 people, as well as agricultural, industrial, and commercial purposes, and more than 70 HEP schemes. The characteristics of each catchment are described in greater detail in the overall methodology (Chapter 3) and study chapters (Chapters 4, 5 and 6). The research presented in this thesis has therefore been undertaken in order to strengthen the knowledge base of future hydrological regime for the region, with the results being useful in future water resource allocation planning, as well as ensuring PWS security and efficient HEP generation.



Figure 1.2. Study catchments; streams greater than second order shown, as defined by Strahler method.

### 1.3 Research aim and objectives

The overarching aim of this thesis is an assessment of the impact of future climate change on the streamflow characteristics of catchments in Wales and an evaluation of the effect of such changes on the abstractable water resource. To achieve this, the thesis will address the following research objectives:

- I. Assess and analyse historical changes in hydroclimatic factors at a catchment-scale for Wales;
- II. Quantify the relationship between weather conditions and total demand for water from surface sources in Wales;
- III. Construct hydrological models for each study catchment and apply future weather data under a worst-case scenario for future climate change;
- IV. Measure and analyse future hydrological regime and water quality changes, given future climate forcing;
- V. Calculate catchment level change in water availability for public water supply and hydroelectric power generation.

### 1.4 Thesis structure

This thesis is comprised of eight chapters. The first (this chapter), provides context to the work, details the justification for the study, and provides a brief introduction to the chosen study area and catchments. The overarching aim and objectives are also stated.

Chapter 2 presents a review of literature relevant to the thesis. Firstly, context is given to the global climate change picture both historically and for the future. The latest report by the IPCC (AR5) is used as a basis for discussion of global change, with recent relevant peer-reviewed work being included to provide further scientific grounding. A review of the changes observed and projected for the UK and Wales is then undertaken. Projected changes in climate for both nations are discussed in the context or UK Climate Projections by the MET Office as well as peer-reviewed work. Literature detailing the impact of the discussed future climatic changes on hydrological regime and quality of surface waters in Wales are then introduced, before the potential implications for water resource use in Wales are conferred.

Chapter 3 details the data, analytical methods, and modelling processes used in the thesis, justifying the reason for their use and presenting the implications that arise from their use. Detailed descriptions of the five catchments studied, as well as an explanation for their use and discussion on the applicability of the results gained outside of Wales, are also provided.

Chapter 4 explores historical hydroclimatic change for the five study catchments, conducting trend analysis for 34 years of temperature, precipitation, and streamflow data. Daily actual abstraction volume data for PWS in each catchment is correlated with the hydroclimatic factors in order to explore the relationship between weather and total water demand.

Chapter 5 considers the impact of climate change on the hydrological regime and water quality of the study catchments, under a high future GHG emissions scenario (Representative Concentration Pathway 8.5), using the latest UK Climate Projections 2018 data. Future streamflow projections are generated using the Soil and Water Assessment Tool (SWAT) as well as projections for four water quality factors. Analysis of changes in average and extreme flows, as well as the average concentration of the water quality factors is presented for the period 2021-2079.

Chapter 6 uses the streamflow data generated in Chapter 5 to quantify the impact of climate change on the generation potential of HEP schemes in the Conwy and the Tywi. Furthermore, the temperature-water demand relationship established in Chapter 4 is implemented in order to project water demand under increasing future temperatures and relate this to available streamflow at a key PWS abstraction location. This chapter therefore provides an impact assessment of future climate change on two key streamflow-dependent sectors in Wales.

Chapter 7 provides an overall discussion of the findings of the thesis and places them in the context of previously published research. The chapter also provides the opportunity to discuss the implications of the results and conclusions of each study chapter upon each other, providing fresh insight into the implications of future climate change on a variety of sectors.

Chapter 8 addresses how the aims set out in Chapter 1 have been met, as well as delivers recommendations for future research to further advance knowledge in the field. Finally, an overall summary of the conclusions is drawn, highlighting the novel study areas and knowledge gaps addressed and detailing the implications of the findings.

A list of figures and a list of tables have been included after the contents page for navigational aid; a list of abbreviations follows this. Two appendices are included at the end of the thesis, presenting, for each catchment, (1) calibration charts comparing modelled and observed streamflow and, (2) ensemble projections of future seasonal and annual streamflow.

The study chapters of this thesis (Chapters 4, 5 and 6) have been submitted in full for publication with peer-reviewed journals. Chapter 4 has been published in Water (Dallison *et al.*, 2020), while Chapters 5 and 6 have been submitted to the *Hydrological Sciences Journal* and the *Journal of Hydrology: Regional Studies*, respectively. Conceptualisation of each of the studies has been the work of Richard Dallison and Sopan Patil. Implementation and investigation of each of the studies has been conducted by Richard Dallison, along with modelling, data analysis and curation, visualisation, and original draft writing and editing. Reviewing and editing of drafts, as well as supervision, has being undertaken by Sopan Patil and Prysor Williams for all studies, as well as Ian Harris for the second study (Chapter 5), who also assisted with resource procurement for this piece of research.

### 1.5 The Dŵr Uisce Project

The work contained within this thesis has been undertaken as part of the Dŵr Uisce Project, which has the overarching aim of improving the efficiency of water distribution in Ireland and Wales. In practice the project has four key themes, the first investigates new smart and low-carbon technologies, such as pumps-as-turbines, drain water heat recovery and smart network controls, to enable energy savings in future water distribution and end use. The second theme is policy support and guidance, work here encompasses benchmarking and energy auditing the water sector, as well as completing life cycle assessments on the technologies developed under the first theme, in order to improve efficiency in the sector. Third is dissemination and collaboration, through a smart specialisation cluster set up by the project, allowing the communication of the project outputs, as well as learning opportunities through four demonstration sites, using the technologies developed. Finally, climate change, this theme is over-arching to the whole project and is the area in which the research in this thesis has been undertaken. The aim of this section of the project is to assess the impacts of future climate change on (1) water supply and water users in the Dŵr Uisce region, and (2) the technologies developed in theme 1 of the project; this thesis has a focus on the former.

### **CHAPTER 2: LITERATURE REVIEW**

### 2.1 Global warming and climatic change

#### 2.1.1 Historical changes

The relationship between atmospheric greenhouse gases (GHGs) and global temperatures has been studied for well over a century, with Arrhenius (1896) first suggesting the idea of the Greenhouse Effect and the link between GHGs and temperature. The seminal work by Callendar (1938) then made the connection between rising carbon dioxide (CO<sub>2</sub>) and increasing temperatures, due to absorption and re-emittance of 'sky-radiation'. It is now widely accepted by the scientific community (Cook *et al.*, 2013, 2016; Oreskes, 2018), that climate change is unequivocally occurring and has been/is being exacerbated substantially by the anthropogenic input of GHGs to the atmosphere (IPCC, 2013). The Fifth Assessment Report (AR5) published by the Intergovernmental Panel on Climate Change (IPCC) states that by 2011, atmospheric concentrations of CO<sub>2</sub>, methane (CH<sub>4</sub>) and nitrous oxide (N<sub>2</sub>O) had increased by 40%, 150% and 20% respectively, compared to 1850 levels (Figure 2.1; IPCC, 2014a).



**Figure 2.1.** Atmospheric concentrations of CO<sub>2</sub>, CH<sub>4</sub> and N<sub>2</sub>O from both direct atmospheric measurement (lines) and indirect analysis of ice cores (dots; IPCC, 2013).

These GHG concentrations are higher than any seen in the last 800,000 years and roughly 40% of GHG emissions from 1750 to 2011 are estimated to have been released since 1970 (IPCC, 2013). Concurrently, warming of the global climate system has occurred at an unprecedented rate, with it being extremely likely to have been predominantly caused by GHG concentrations and other anthropogenic drivers (IPCC, 2014b). Global average temperatures reached approximately 1.0°C warmer, when compared to pre-industrial levels in 2017, with global warming likely to reach 1.5°C between 2030 and 2052 (Figure 2.2; IPCC, 2018).



**Figure 2.2.** Observed (grey line to 2017) and projected (2017-2100) global mean surface temperature. Solid orange line represents estimated anthropogenic warming, dashed continuation and orange horizontal error bar represent central estimate of time range when 1.5°C increase is likely to be reached. Future projections are based on net zero global CO<sub>2</sub> emissions by 2055 (grey plume) and 2040 (blue plume), both with declining non-CO<sub>2</sub> radiative forcing after 2030. Purple plume represents net zero global CO<sub>2</sub> emissions by 2055 combined with no decline in net non-CO<sub>2</sub> forcing after 2030. Adapted from IPCC (2018).

With regard to global precipitation changes, AR5 states that there is low confidence in changes in average precipitation since 1901, with a small increase detected (IPCC, 2013). However, when studying latitudinal zones (Figure 2.3), it is clear that more defined regional-scale changes have occurred over the last century. The most substantial change is seen in the mid-latitudes of the Northern Hemisphere (30°N to 60°N), with a trend towards increasing precipitation observed from 1901 to 2008. Further analysis reveals that this increasing trend is most pronounced in the period pre-1951 (IPCC, 2013). In contrast, evidence for change in the mid-latitudes of the Southern Hemisphere (60°S to 30°S) is less clear, with a slight increase seen to the 1980s and a small decline after this. Precipitation over the tropics (30°N to 30°S) is largely stable, up to the 1980s, whereby a small decline is seen, peaking in the mid-1990s, before returning to levels in line with the rest of the dataset (IPCC, 2013).



**Figure 2.3.** Historical over-land annual precipitation anomalies compared to the 1981-2000 mean, as detected by five global precipitation datasets. Timeseries for four latitudinal bands as well as globally shown (IPCC, 2013).

Trends in precipitation reported by the IPCC are consistent with those found in the literature prior to its publication; for example Dore (2005) reports that for the latitudinal zone 30°N to 85°N, an increase of between 7% and 12% has occurred over the 20<sup>th</sup> century. The change in the south (0°S to 55°S) is much less pronounced at around 2%, reflecting the difference in the distribution of landmasses for the two zones (Dore, 2005). Similarly, Trenberth (2011) identifies historical trends of dry areas (the sub-tropics) becoming drier, and wet areas (mid-to high-latitudes in the north) becoming wetter, with little change in the global average. Additionally, using fourteen climate models compared to observed changes which correspond with those suggested by the IPCC, Zhang *et al.* (2007) show that anthropogenic forcing has had a detectable impact on changes in precipitation between 1925-1999 and 1950-1999. Since the publication of AR5, further evidence has been published on historical global

precipitation change, this also largely supports the reported trends. Gu & Adler (2015) for example, showed how global precipitation change occurred concurrently with surface warming, especially the increases observed between 1901-2010 over high latitudes of the northern hemisphere. Further regional variations are also described, with increases in annual precipitation seen in Australia and north America, while declines are observed for tropical Africa and the Indian-Tibetan region (Gu & Adler, 2015). Furthermore Adler *et al.* (2017) study precipitation trends during the 'satellite era' (1979-2014) using the Global Precipitation Climatology Project (GPCP) monthly dataset (Adler *et al.*, 2003; Huffman *et al.*, 2009). The results of this shorter-term study suggest a near-zero trend in global precipitation change, but do highlight the same regional trends seen in the longer-term studies (Adler *et al.*, 2017).

Examining change in extreme weather and climate events, AR5 reveals that at a high confidence level, globally there has been an increase in the value of both maximum and minimum temperatures (IPCC, 2013). The rate of change of increasing minimum temperatures is also shown to be greater than the rate of change in maximum temperatures. The Asia region is most effected by increasing minimum temperatures, while Africa is more greatly affected by increases in maximum temperature (IPCC, 2013). Similar regionality is displayed for changes in the frequency of heatwaves, with Europe, Asia and Australia all being subject to a great number of heatwaves through the 20<sup>th</sup> century. As with trends in precipitation, changes in extreme temperatures presented by the IPCC agree with studies published more recently. Horton et al. (2015) demonstrate for the northern hemisphere that during the satellite era, minimum temperature have increased for all regions except western North America, especially in summer and autumn; maximum temperature increases are more observed across Eurasia, in all seasons except winter. Work by Papalexiou et al. (2018) also speaks to an increase of global maximum temperatures over the period 1966-2015; Mishra et al. (2015) also demonstrate that the increase in extreme temperatures are pronounced for urban areas globally. Global increases in the number of heatwaves are also demonstrated in work by Liu et al. (2017), Chapman et al. (2019) and Perkins-Kirkpatrick & Lewis (2020), with general consensus also being for an increase in intensity. While changes in average precipitation volume were small across the 20<sup>th</sup> century, trends in extreme precipitation events have been more extensive, with heavy precipitation events increasing disproportionality compared to the mean (IPCC, 2013). This trend is seen globally, but regional

and seasonal variation are detected, with the most consistent increases in heavy rain event severity observed in Europe and North America (IPCC, 2013). Once again, recent research by the likes of Asadieh & Krakauer (2015), Alexander (2016), and Papalexiou & Montanari (2019) supports these findings.

Regardless of location and regional differences in climate change impacts, all areas will see consequences due to future climate change. Indeed, all global regions have already experienced impacts, both on natural and human systems, such as agriculture and food production, poverty and livelihoods, ocean systems, and freshwater resources (IPCC, 2014a, 2014c). Freshwater systems are some the most effected however, due particularly to changing precipitation patterns as well as alterations in snowmelt characteristics which have had a major impact on the global hydrological cycle (IPCC, 2014a). Changes in hydrological regime have already caused important consequences in various regions for issues such as water resource allocation, fluvial ecosystems, and flood and drought management, as outlined in Chapter 1. The sustainability of water supply and increasing water scarcity are of particular concern, this is true not only for arid countries, but also more generally, especially in areas of dense population. England, for example, is not a country classed as arid, it does however have areas of water stress, in the south of the country in particular. This area already experiences water shortages (Cook, 2016), a situation which is likely to occur more frequently under future climate change. Dobson et al. (2020), for example, projects an increase in the probability of water use restriction implementation doubling by 2050 and quadrupling by 2100.

In recent years, changes in Earth's climate have been particularly stark, with 2019 being the second warmest year on record, after 2016 (WMO, 2020) and July 2019 being the hottest month on record (Blunden & Arndt, 2020). Averaged across Europe, 2019 was the warmest year on record, with central Europe experiencing drought (European Commission, 2020). Globally, the hottest 5-year period on record was also set for 2015-2019, as was the hottest decade (2010-2019), with each decade since the 1980s being warmer than the last (WMO, 2020). Records were also set in 2019 for highest global mean sea level and ocean heat content, while May, June and July saw record low sea-ice extents (WMO, 2020). Additionally, the American Meteorological Society reported that global atmospheric CO<sub>2</sub> concentration reached a record high in 2019 of 409.8 ppm, while the CO<sub>2</sub> uptake of global oceans also

increased to 33% higher than the 1997-2017 average (Blunden & Arndt, 2020). All of these changes point to an apparent worsening of the impacts of climate change in recent years.

#### 2.1.2 Future projections

Projections for global climate in the future generally show a continuation of the trends already observed over the 20<sup>th</sup> century and first two decades of the 21<sup>st</sup>. However, the extent to which this change will continue is unknown, due to its dependence on future atmospheric GHG concentrations and the degree of climate forcing and global warming that this causes. Since 2010, the context of future climate change and its impacts has been based on four Representative Concentration Pathways (RCP2.6, RCP4.5, RCP6.0, and RCP8.5), developed by the scientific community for the IPCC in advance of AR5 (Moss *et al.*, 2010; van Vuuren *et al.*, 2011). The RCPs represent a trajectory of atmospheric GHG concentrations under which a specific degree of radiative forcing would be reached. The designation of each RCP therefore relates to this degree of radiative forcing reached by 2100 (Table 2.1), with RCP2.6 for example, relating to 2.6 W m<sup>-2</sup> by 2100 (this scenario peaking at 3 W m<sup>-2</sup> before 2100 and then declining).

RCP	Radiative forcing	$CO_2$ equivalent concentration (ppm)	Pathway
RCP2.6	Peak at 3 W m <sup>-2</sup> , decline to 2.6 W m <sup>-2</sup> by 2100	Peak at 490 before 2100, then declines	Peak and decline
RCP4.5	Rising to reach 4.5 W m <sup>-2</sup> peak by 2100	650 at stabilisation in 2100	Stabilisation
RCP6.0	Rising to reach 6 W m <sup>-2</sup> by 2100, peak soon after	850 at stabilisation after 2100	Stabilisation
RCP8.5	Rising to reach 8.5 W m <sup>-2</sup> by 2100, peak in 2200s	>1,370 in 2100	Rising

**Table 2.1.** Overview of the characteristics of the four Representative Concentration Pathways (RCPs), adaptedfrom Moss *et al.* (2010).

While RCP2.6 denotes a peaking and declining pathway associated with early GHG mitigation, RCP4.5 and RCP6.0 describe stabilising pathways due to GHG emissions reduction after 2050 (Figure 2.4), with RCP4.5 peaking in terms of radiative forcing at 2100 and RCP6.0 before 2150 (van Vuuren *et al.*, 2011). RCP8.5 represents a worst-case scenario of future global emissions (Figure 2.4), the 90<sup>th</sup> percentile of the no-policy baseline scenarios, with increasing high levels of atmospheric GHG concentrations, leading to a scenario of radiative forcing still being in significant increase by 2100 (van Vuuren *et al.*, 2011).



**Figure 2.4.** CO<sub>2</sub> emissions under each Representative Concentration Pathway (RCP; coloured lines) and historically (black line). Coloured plumes show range of scenarios relating to atmospheric CO<sub>2</sub> concentrations as described in the panel on the figure (ppm; 5% to 95% range shown for each category; IPCC, 2013).

The IPCC used the fifth phase of the Coupled Model Intercomparison Project (CMIP5) in AR5 for the analysis of future climate change under the forcing of each RCP (IPCC, 2013). The CMIP5 ensemble comprises 42 global climate models (GCMs) at different spatial resolutions (IPCC, 2013; Ahmed *et al.*, 2019) and is used to give both a multi-model mean, and range of model outcomes. Projections by the IPCC under all RCPs suggest a continuation of global warming, at least in the short-term, with temperatures stabilising under RCP2.6 before 2050 at 1°C (±0.4°C [1 standard deviation]) higher than the 1986-2005 mean (IPCC, 2013). Under RCP8.5, mean temperatures for 2081-2100 are projected to rise by 3.7°C (±0.7°C), when compared to the same mean, with projections for RCP4.5 and RCP6.0 being 1.8°C (±0.5°C) and 2.2°C (±0.5°C) respectively (Figure 2.5; IPCC, 2013). These results relate to an assessment of high confidence in likely exceedance of 1.5°C above 1850-1900 temperatures by 2081-2100 for RCP4.5, RCP6.0 and RCP8.5; this is also true for exceedance of 2°C by RCP6.0 and RCP8.5 (IPCC, 2013). It is also noted by the report that it is as likely as not (medium confidence) that global temperatures will exceed 4°C under RCP8.5, occurring in more than half of the CMIP5 ensemble models.



**Figure 2.5.** Global average surface temperature (relative to 1986-2005 mean) for 2006-2100 under Representative Concentration Pathways. Multi-model mean shown for RCP2.6 (blue line) and RCP8.5 (red line), 5% to 95% range of CMIP5 model outcomes shown by bands. Multi-model mean for mean 2081-2100 temperature for all RCPs also shown on right hand side, with model ranges (IPCC, 2013).

Global precipitation levels also display an increase to 2100, although much more gradual than global temperature change; a 2% to 5% increase over the period is observed under RCP2.6 and RCP8.5 respectively (IPCC, 2013). Interestingly, little relationship is seen between changes in total precipitation and extreme precipitation (IPCC, 2013), with the intensity of extreme precipitation events projected to increase. These increases are expected both for short-term events (individual storms) and longer periods, with a 5% (RCP4.5) to 20% (RCP8.5) increase in 5-day maximum precipitation accumulation (IPCC, 2013). Return periods of large precipitation events are also shown to be impacted, extreme precipitation events being very likely to become more severe and frequent, a 10% to 20% reduction in return period is observed per 1°C warming; for tropical regions this value is even higher (IPCC, 2013). With regards to temperature extremes, it is stated that it is virtually certain that fewer cold extremes will be experienced, but a greater number of hot extremes, in line with global warming. Warmest daily maximum temperature is shown to increase between 1.5°C (RCP2.6) and 5.5°C (RCP8.5) by 2100, with coldest daily minimum increasing by an even larger margin, between 2°C (RCP2.6) and 7°C (RCP8.5). Heatwaves are also projected to very likely occur more frequently and for longer; in the same manner as for precipitation, the magnitude of events with a 20year return period is set to increase substantially for minimum and maximum temperatures (IPCC, 2013).

### 2.2 Climatic trends in the UK and Wales

### 2.2.1 Historical changes

The climate of Wales has altered in line with general trends seen globally and for the wider UK, that being over the course of the 20<sup>th</sup> century an increase in mean temperature, with daily temperatures increasing in all seasons, while annual precipitation has seen little change, but large seasonal variation. Estimations of the degree and nature of changes vary between studies, dependent on the time frame under review. Farrar & Vaze (2000), for example, report that for the period 1901 to 1998 annual temperatures warmed slightly, with more substantial change observed in autumn and winter, but little change in the summer. Similarly, the Welsh Government (2013) detailed how over the course of 1914-2006 mean annual temperature rose 0.7°C, with the greatest increase in autumn (0.99°C) and least in winter (0.47°C). While the two reports agree about increased seasonality of precipitation, decreases in summer and autumn, increases in winter and spring, there is a contrast in the annual observations. Farrar & Vaze (2000) report a ~3% increase in annual precipitation, while Welsh Government (2013) suggest a 0.9% decrease over their longer and more recent study period. These difference seem to be driven by different magnitudes or seasonal change reported, especially in summer, with Welsh Government (2013) reporting a 24% decrease, and Farrar & Vaze (2000) a smaller 15% decrease. The majority of works align with either the latter view, a small decline in annual precipitation historically, or suggest no trend at all (Perry, 2006; Jenkins et al., 2008; ASC, 2016b; de Leeuw et al., 2016).

In terms of extreme temperature and precipitation events, a report by the MET Office Hadley Centre (MOHC) on UK climate extremes states that for the period 1961-2017 annual maximum temperatures increased across the UK, including Wales specifically (MOHC, 2018a). The number of warm spell days (days associated with spells of six or greater consecutive days where temperatures exceed the daily climatology 90<sup>th</sup> percentile) also increased substantially, more than doubling for Wales when comparing the 1961-1990 and 2008-2017 averages (MOHC, 2018a). Extreme events associated with cold weather also display warming, with fewer icing days (days where maximum temperature is less than 0°C) and lowest minimum temperatures increasing (MOHC, 2018a). For Wales specifically, average lowest minimum temperature increases from -7.8°C for 1961-1990 to -6.6°C for 2008-2017; icing

days reduce from 4.2 to 2.7 (MOHC, 2018a). This non-standard record period of 2008-2017 does have the potential to miss-represent long-term trends, as the time period is more likely to be influenced by decadal time-scale variation, nonetheless it does provide a useful comparison of the current state of change in the climate. Precipitation trends reported by MOHC (2018a) display an increase for highest 5-day precipitation total from 1960 to 2000, rising 4%, with this being more pronounced in the west of the UK. Additionally, 1-day extreme rainfall (greater than 99<sup>th</sup> percentile) totals also increases by 17% nationally, with the change being more pronounced for Wales than the national average (MOHC, 2018a). Little change in the length of wet spells (consecutive rainfall days) is observed.

Further literature (Perry, 2006; Jenkins et al., 2008; Burt & Ferranti, 2012; Jones et al., 2013) studying the nature of change in extreme temperature and precipitation events, agrees with the broad findings of the MOHC report. Perry (2006), for example, corroborates the findings relating to annual maximum temperature, with observations of a 0.59°C increase for north Wales (p < 0.01) and a 0.55°C increase for south Wales (p < 0.05), for the period of 1914-2004. An increase in summer and annual minimum temperature is also reported at a significance level of p < 0.01 for both north Wales (+0.88°C and +0.84°C respectively) and south Wales (+0.56°C and +0.55°C); once again in agreement with MOHC (2018a). Furthermore, Jones et al. (2013) brings agreement to the precipitation trends shown, reporting an increase between 1961-2009 in 1-, 5- and 10-day precipitation totals for north and south Wales, albeit only statistically significant for 5- and 10-day events. Within these trends, statistically significant increases in spring 2-, 5- and 10-day totals are observed, while the same is true for summer declines, this also being significant for 1-year events. Winter totals for all four event lengths rise significantly (Jones et al., 2013). The literature also supports observations of increasing seasonality in extreme precipitation events, with a greater proportion of winter precipitation falling during intense events (Osborn & Hulme, 2002; Jenkins et al., 2008; Burt & Ferranti, 2012; Jones et al., 2013).

Studying the most recent complete year, 2019, in isolation, Kendon *et al.* (2020) report it to be the 12<sup>th</sup> warmest on record for the UK overall, with national records set for all time highest recorded temperature (38.7°C) and winter high (21.2°C). For Wales, annual average temperature was 0.5°C higher than the 1981-2010 average, mainly due to the winter average being 1.6°C higher, one of the top 10 on record (Kendon *et al.*, 2020). A below average number

of frost days also occurred for the sixth straight year, while rainfall was above both the 1981-2010 and 1961-1990 averages, at 107% and 112% respectively (Kendon *et al.*, 2020). The work conducted in Chapter 4 further examines the changes seen in temperature and precipitation for Wales for the period 1982-2015, investigating changes in annual and seasonal averages and frequency of extreme events, as well as breakpoints of marked change during the period.

#### 2.2.2 Future projections

For the UK the key point of reference on future climate is the MOHC's UK Climate Projections (UKCP). This climate analysis tool provides the most up-to-date assessments of future climate outlook for the nation, aiding with planning future risk management and adaptation plans. Recent projections were published in late 2018 (hereafter UKCP18), succeeding the 2009 iteration (UKCP09). The UKCP09 projections have been widely used in literature for impact assessments on various sectors, indeed 107 impact assessment studies using UKCP09 data are indexed by Web of Science at time of writing (6<sup>th</sup> October 2020). To date however, little peer-reviewed literature has yet been published using the UKCP18 projections, with only two impact study results appearing in Web of Science searches (6<sup>th</sup> October 2020) for these projections (Kay et al., 2020; Mohammadi et al., 2020). A further three studies have been published assessing the methods and performance of the projections (Murphy et al., 2019; Fosser et al., 2020; Rostron et al., 2020). The research by Kay et al. (2020) in fact both assess the impact of climate change on river flows in ten UK catchments, and compares the UKCP09 and UKCP18 outputs, demonstrating that for the catchments studied, the results of both projections are similar and comparable. Therefore, the results of previous work using UKCP09 projections remain valid and relevant, allowing for future works to assess new research areas.

The UKCP18 data contain three 'strands', probabilistic (strand 1), global (strand 2) and regional (strand 3). Strand 1 provides probabilistic projections at a 25 km grid resolution derived from 350 climate model simulations, expressing a wide range of potential outcomes (Murphy *et al.*, 2018). This dataset is important for exploring uncertainty in future impact assessments, giving the most complete view of outcomes under all four RCP scenarios assessed in AR5 (Murphy *et al.*, 2018). Strand 2 uses 28 climate models; a 15-member perturbed parameter ensemble (PPE) from the MOHC's HadGEM3-GC3.05 (hereafter GC3.05) model, and a 13-member ensemble from the CMIP5 coupled ocean-atmosphere model.
These thirteen CMIP5 models were selected from the 42 used by the IPCC in AR5, based on their suitability in terms of data coverage and performance criteria (Gohar *et al.*, 2018). These globally based projections under RCP8.5 are particularly useful for exploring the impact of spatially large-scale drivers of change, for example global circulation, and are provided at 60 km grid spacing to the year 2100 (Murphy *et al.*, 2018). Finally, strand 3 projections use twelve of the fifteen GC3.05 PPE members used for strand 2, downscaled to a regional projection and driven by global simulations for the period 1980-2080 under RCP8.5 conditions (Murphy *et al.*, 2018). Projections are presented as a 12 km grid for the British Isles, therefore taking account of regional-scale features such as mountains, coastlines, and large waterbodies, allowing for local level analysis (Murphy *et al.*, 2018). It is these regional projections that have been used in this thesis for future streamflow estimation (Chapter 5) and subsequent impact assessment for hydroelectric power (HEP) and public water supply (PWS; Chapter 6).

The UKCP18 overview report describes projected future UK climate to 2100 using the strand 1 probabilistic projections detailed above. Under RCP2.6 conditions, the median annual average temperature for 2080-2099 is expected to reach 1.4°C higher than a 1981-2000 baseline, under RCP8.5 this figure is 3.9°C (Figure 2.6; Lowe *et al.*, 2018), further analysis suggest that temperature increases will be larger in summer than winter (MOHC, 2019a). For precipitation the same seasonal variation is seen, with reductions in median summer average (RCP2.6 -15%; RCP8.5 -29%) being much more pronounced than increases for winter (RCP2.6 +6%; RCP8.5 +18%; Figure 2.7; Lowe *et al.*, 2018). Additionally, regional variation is also present, with southern and central England seeing larger winter increases, and the south bigger summer reductions, than the UK average (MOHC, 2019b).



**Figure 2.6.** Change in UK average temperature compared to the 1981-2000 mean, under RCP2.6 (blue) and RCP8.5 (red). Shading shows percentile boundaries, with solid line indicating the median; black lines are observed temperatures made by the National Climate Information Centre (Lowe *et al.*, 2018).



**Figure 2.7.** Percentage change in UK average precipitation compared to the 1981-2000 mean, under RCP2.6 (blue) and RCP8.5 (red) for winter (top) and summer (bottom). Shading shows percentile boundaries, with solid line indicating the median; black lines are 5-year mean of observed precipitation made by the National Climate Information Centre (Lowe *et al.*, 2018).

With respect to trends in future climate projected for Wales specifically, the UK Climate Change Risk Assessment for Wales, by the Adaptation Sub Committee (ASC) of the Committee on Climate Change, is the most recent publication to report (ASC, 2016b). This report was published prior to the release of UKCP18 datasets, so assesses trends under the former UKCP09 projections, with an update using the new projections expected by the end of 2020 (Welsh Government, 2019a). As aforementioned, these results are still valid, being shown by Kay *et al.* (2020) to be similar and comparable to the most recent projections. With that in mind, Murphy et al. (2009) present trends for Wales, based on a high-medium future GHG emissions scenario (SRES A1B, equivalent to between RCP6.0 and RCP8.5). In Wales, climate is set to follow general UK trends of warmer wetter winters and hotter drier summers, with an increase in occurrence of extreme events (Murphy et al., 2009). Mean winter temperature for example is projected to increase by 2°C (median probability; range of 1.1°C to 3.1°C; representing 10% and 90% probability level) for Wales by 2080 (compared to 1961-1990 levels); mean summer temperatures increase by a greater amount 2.5°C (1.2°C to 4.1°C; Murphy et al., 2009). Summer maximum and minimum temperatures are both expected to increase, maximum by 3.4°C (1.3°C to 6.1°C) and minimum by 2.6°C (1.1°C to 4.6°C; Murphy et al., 2009). Annual precipitation is projected to be stable at a median probability level under SRES A1B to 2080, with the 10% to 90% probability range being -4% to +5%. This masks seasonal variation however, with winter precipitation expected to increase by 14% (+2% to +30%) and summer to decrease by 16% (-36% to +6%; Murphy et al., 2009).

Work conducted in Chapter 5 characterises alterations observed under future UKCP18 climate projections for Wales, investigating changes in annual and seasonal averages up to 2080, representing an update on the UKCP09 trends presented by the ASC (2016b) and Murphy *et al.* (2009) and discussed above for Wales.

## 2.3 Changing hydrology in the UK and Wales

#### 2.3.1 Historical hydrological trends

As previously discussed, hydrological systems have been some of the worst effected systems globally by climate change historically (IPCC, 2014a), with notable variation in impact dependent on regional characteristics. The UK is no exception, with Harrigan *et al.* (2018)

demonstrating a 6.8% (interquartile range: 0.1% to 13.8%) increase in annual mean flow between 1965-2014 when studying 146 near natural catchments across the UK. The same study also analysed seasonal trends over the same period, with winter, summer and autumn mean flows increasing by 12.7% (4.1% to 26.2%), 1.4% (-11.2% to 18.5%), and 16.7% (5.5% to 25.5%) respectively; spring flow is shown to have declined by 10.7% (-19.0% to 1.6%; Harrigan *et al.*, 2018). Interestingly, mapped results for the UK reveal that annual mean streamflow trends display regionality in Wales, with catchments in the north of the country having generally positive trends, and those in the south being negative (Harrigan *et al.*, 2018). Variations in trend characteristics (magnitude and significance) are also observed across the UK, with a northwest-southeast divide being particularly prominent. For example, seasonal analysis conducted by Hannaford & Buys (2012) found that for the period 1969 to 2008, all seasons showed predominantly upwards trends for mean streamflow in catchments in Scotland and the northwest. In the south and east however, the picture was much more varied, with downward trends dominating for spring, upwards for autumn, and mixed changes seen in winter and summer (Hannaford & Buys, 2012).

With regards to extreme hydrological events, Harrigan et al. (2018), for the same study conditions as above, demonstrate a 13.5% (7.9% to 23.8%) increase in Q<sub>5</sub> volume (flow volume exceeded 5% of the time), with agreement of trend direction in 88.4% of catchments. The same level of agreement is not achieved for low flows, with 52.4% of catchments displaying an increase in volume for this metric, and 47.6% showing a decline (Harrigan et al., 2018). Additionally, work by Hannaford & Marsh (2008), was some of the first to identify the northwest-southeast divide in large streamflow event trends. Results of the work demonstrate an increase in annual flow maxima across all of the UK, but with a marked increase in magnitude in Wales and northwest England (generally upland areas with a maritime climate) compared to the east and south. For trends in 10- and 30-day maximum flows, the frequency of large flows, and flood magnitudes, significant increases are seen again for the west and north, but little evidence is seen in the south and east (Hannaford & Marsh, 2008). The increasing trends seen for the westerly catchments are supported by the results of Dixon et al. (2006), who studied 56 gauging stages in Wales and the Midlands of England for the period 1962-2001. Even here, the vast majority of trends display an east-west gradient in terms of both statistical significance and magnitude of change, being true for increases in

annual maximum flow, high flows (Q<sub>10</sub>), Q<sub>mean</sub>, low flows (Q<sub>90</sub>), and annual minimum flows (Dixon et al., 2006). Furthermore, Hannaford and Buys (2012) identify that for low flows, the northwest-southeast divide only occurs in summer (northwest increasing, southwest decreasing), although different regional variations are still seen in other seasons. In winter for example, low flows show a decline in Scotland, Northern Ireland and northeast England, while the midlands and the south are mainly positive. The divide is still present in the high flow analysis, albeit generally less defined. In winter for example, strong positive trends are seen for Scotland, Northern Ireland and the northwest; while positive trends are also seen in the south, but are substantially weaker and generally confined to only smaller catchments (Hannaford & Buys, 2012). Furthermore, in both spring and summer a positive trend in high flows is seen in the northwest, while results are more mixed in the southeast; for autumn rises are seen across the country (Hannaford & Buys, 2012). Work by Laizé & Hannah (2010), to further characterise the spatial variation of seasonal streamflow, suggests that spatial patterns show inter-season variability, meaning the divide is not clear cut. More generally the paper suggests that it is the physical characteristics of basins which drive differences, with elevation and permeability being particularly important. In the UK, this corresponds to the characteristics of the areas in the southeast and northwest, as well as to the greater exposure to prevailing westerly winds that the northwest receives (Laizé & Hannah, 2010).

Evidence reviews completed by Hannaford (2015), Watts *et al.* (2015a) and Garner *et al.* (2017), using published literature on historical UK hydrological change, all identify similar key trends for average and extreme flows, in line with those detailed above. The overarching agreement in the literature being for little change in summer flows, with small declines seen in some areas, an increase in annual and winter flows, and an increase annually in high and low flow event frequency and magnitude.

#### **2.3.2** Future projections under climate change

With regard to future impacts for the hydrology of the UK and Wales, continued changes in the timing, quantity and quality of streamflow are likely. Several studies in recent years have sought to suggest future UK streamflow scenarios for both average (Christierson *et al.*, 2012; Prudhomme *et al.*, 2012; Sanderson *et al.*, 2012; Kay *et al.*, 2020) and extreme flows (Prudhomme *et al.*, 2013a, 2013c; Kay *et al.*, 2014a, 2014b, 2020). Several more have studied

smaller areas, Charlton & Arnell (2014) for example for England only; Remesan *et al.* (2014) for the River Derwent; Cloke *et al.* (2013) for the upper Severn; and Bell *et al.* (2012) for the Thames. General consensus among these studies is for a largely stable annual average streamflow, but with large variations seasonally, in particular increases in winter and decrease in summer. These changes are in line with a continuation of historical trends discussed above, and with future climate projections for the nation.

The seasonal average picture for future streamflows is for greater winter and spring flow, with reductions in summer and autumn; as with historical flows, this is particularly applicable to north-western impermeable upland catchments. This trend was demonstrated for the near-future (end of 2020s) by Christierson et al. (2012), who reported, using UKCP09 data, a reduction in summer flows across the country. Furthermore, a small increase in winter flows was observed for western areas, with annual mean flow decreasing for all of the UK. Longerterm research by Prudhomme et al. (2012) to 2050 explored national level changes under each of the eleven ensemble members that form the HadRM3-PPE-UK model that drives UKCP09 (Murphy et al., 2007). Comparing mean seasonal flows for 2040-2069 with those for the baseline 1961-1990 period highlights a decrease in summer mean flow that is particularly pronounced in the west; autumn flows are more variable with a larger proportion of the 11 models showing a decline, more so for England and Wales than Scotland (Prudhomme et al., 2012). Winter is characterised by increasing flows, showing spatial variability between the 11 models for areas of greatest magnitude of change; generally, however, England and Wales are more effected than Scotland. Spring flows are the most variable, with little agreement in magnitude or trend direction displayed (Prudhomme et al., 2012). Sanderson et al. (2012) project future streamflow even further, to the 2080s, in their study of the UK using the same 11-member ensemble, under the SRES A1B future emissions scenario. Once again winter streamflows display an increase across the country by the 2080s, with agreement for almost the whole country on this direction of change. Coastal regions of west Wales are projected to see a particularly large increase, between 15% and 25%, with all 11 models in agreement. For spring flows all but the north of Scotland is reported to see no change or a small decrease, however there is little agreement on this between the models (Sanderson et al., 2012). Summer and autumn flows show a greater degree of agreement for trend direction between the models, with decreases for the whole of the UK in summer and all but the northwest of

Scotland in autumn. In summer, similar to winter, coastal regions in the north, west and south of Wales are projected to see large (-15% to -25%) decreases in mean flow, with agreement in all 11 models; autumn mean flow shows a more moderate decline (-5% to -15%; Sanderson *et al.*, 2012). Finally, median annual flow for the period 2040-2069 was shown by Kay *et al.* (2020) to decrease for 10 catchments studied throughout the UK, under both RCP4.5 and RCP8.5 conditions; the magnitude of this reduction only being marginally greater under RCP8.5, but with greater uncertainty.

Kay et al. (2020) also analysed changes in high  $(Q_5)$  and low  $(Q_{95})$  flows under the same conditions as for average flows above. Trends in Q<sub>5</sub> displayed variation in line with the eastwest divide aforementioned; those catchments in the west exhibiting small increases in future high flows, and those in the east presenting little or no change. Low flows were much more consistent, all catchments'  $Q_{95}$  projected to decline under both RCP4.5 and RCP8.5 (Kay *et al.*, 2020). The majority of research on future extreme streamflows is framed within the context of changing flood and drought risk. This is true of Collet et al. (2018) who's findings concur with those of Kay et al. (2020). The study used the Future Flows Hydrology (FFH) dataset (Prudhomme et al., 2013b) to compare 2080s hydrological hazards with the 1970s. The median trend of the FFH, is for increases in frequency, magnitude and duration of both floods and droughts almost exclusively in the west of the UK only, especially concentrated in the south west of England and Wales (Collet et al., 2018). The magnitude and duration of droughts is projected to show a larger percentage change than the magnitude of floods, but changes in frequency are broadly the same (Collet et al., 2018). Interestingly, the paper also assess the change in seasonality of events, showing that floods are much more likely to occur in winter in the future, while droughts are less seasonally bound, but still more likely to occur in the autumn, and to a lesser extent, the summer (Collet *et al.*, 2018).

A considerable amount of work has been completed in the last decade for the whole of the UK and regions/catchments thereof, in characterising future high flow and flood regimes under changing climate (e.g. Cloke *et al.*, 2013; Lavers *et al.*, 2013; Kay *et al.*, 2014a, 2014b; Sayers *et al.*, 2016; Jenkins *et al.*, 2017). Kay *et al.* (2014a, 2014b) comprehensively reviewed probabilistic impacts on flood magnitude in particular, for England, Wales and Scotland. Using the UKCP09 projections, it is shown for England and Wales that under the SRES A1B emissions scenario, peak flows in 20-year return period floods will increase by between 20% to 25% in

the east and by 25% to 35% in the west (Kay *et al.*, 2014a). Under a high A1F1 SRES emissions scenario (equivalent to RCP8.5), this trend remains, with magnitudes of change becoming larger, the east rising to between 25% to 35%, and the west to between 35% to 45% (Kay *et al.*, 2014a). For Scotland a similar trend is observed in terms of spatial distribution, the west being more greatly affected than the east, but the magnitude of change is larger; the very north west for example, seeing a 45% to 60% increase under SRES A1F1 scenario by the 2080s (Kay *et al.*, 2014b). Sayers *et al.* (2015) also suggest a 13% increase by 2080 under a 2°C warming scenario, for peak streamflows in Wales, increasing to 42% in east Wales, and 36% in west Wales under 4°C scenario. These results have large implications for return period changes; for example, under the 2°C scenario, the return period for an event currently at 100 years, would reduce to ~40 years; under the 4°C scenario this would be reduced to less than 11 years in east Wales, and ~13 years in the west of the country (Sayers *et al.*, 2015).

While future hydrological drought trends have not received the same level of attention as flood events in the literature, a few studies have characterised projected changes through the 21<sup>st</sup> century. As is common with many of the studies presented in this section, the HadRM3-PPE-UK 11-model ensemble has been used by Rahiz & New (2013) to comprehensively assess the severity of future droughts in the 2020s, 2050s and 2080s; the results for the 2080s are discussed here. The intensity of modest droughts is to increase substantially for England and Wales, while the increase in intensity of the most severe droughts is set to be less, although the most pronounced area in the UK is shown to be Wales (Rahiz & New, 2013). Additionally, the frequency of drought occurrence shows the same spatial pattern for the UK as intensity, but the rate of change for severe and moderate events is broadly the same, both increasing in England and Wales (Rahiz & New, 2013). Average drought length however is shown to decrease, with little change in the number of 12- and 10-month droughts projected, but an increase in 6- and 3-month events, especially the latter and for both severe and moderate events (Rahiz & New, 2013). Overall for Wales, therefore, the study projects increased intensity and frequency events, which will very likely be shorter in length than at present. The results of Rudd et al. (2019) are broadly in agreement with Rahiz & New (2013), with similar spatial patterns for intensity and drought duration shown for the 2080s. Studying low flow characteristic changes, Kay et al. (2018) compared low flow frequency curves, to quantify the percentage change in 2- and 20-year return period events between the 1975-2004 average

and the 2070-2099 average. As with the previous worked based on drought, results show a decrease in the volume of 2- and 20-year low flows, suggesting an increase in drought severity; again this is especially pronounced across England and Wales (Kay *et al.*, 2018).

In addition to changes in precipitation and temperature projected up to 2080, Chapter 5 also analyses trends seen in annual and seasonal average streamflow, as well as the intensity and frequency of extreme high and low streamflow events.

## 2.4 Water quality implications of hydroclimatic change

Changes in precipitation patterns and hydrological regime will impact and influence the instream environment. One of the key change areas will be river water quality, with key processes such as run-off, overland flow, stream power, dilution potential and carrying capacity all likely to be altered. Globally, the effect of changing precipitation extremes and streamflow patterns on water quality factors has been well studied, often at a regional or catchment-scale (e.g. Johnson et al., 2009, 2015; Tu, 2009; Whitehead et al., 2009b; Rehana & Mujumdar, 2012; Crossman et al., 2013; Michalak, 2016; Wang et al., 2018; Guo et al., 2019). General consensus is that increasing extreme precipitation events will likely lead to larger overland flows with greater capacity for eroding and flushing of non-point-source pollutants from the land surface into river systems (Nilsson & Malm Renöfält, 2008; Whitehead et al., 2008; Guo et al., 2019). In addition, large discharge events are likely to cause greater in-stream erosion of bed and banks, causing increased turbidity (Nilsson & Malm Renöfält, 2008; Bussi et al., 2016). Paradoxically, an increased likelihood of low flow events is also expected to decrease water quality, with less water available for dilution, this is clearly a larger issue downstream of point-sources of pollution, for example industrial and wastewater treatment outfalls (Nilsson & Malm Renöfält, 2008; Whitehead et al., 2008, 2009b). Work by Gau et al. (2019) for 102 water quality monitoring stations in southeast Australia identifies that same day streamflows are the largest driver of declining water quality, also suggesting that reduced flow and lack of dilution is less important; this is of course highly location dependent however.

In light of this body of research, the previously discussed projected increases in magnitude and frequency of larger streamflow and precipitation events, as well as decreasing summer

flow and increased magnitude, frequency and duration of droughts is invariably expected to impact negatively on water quality in the UK and Wales. In terms of water quality changes, less work has been published in the UK on this topic than for hydroclimatic changes, with even fewer publications focusing on or featuring Wales specifically. Nonetheless a consensus in the literature is for a general decrease in surface water quality under future climate change in the UK (Johnson *et al.*, 2009; Whitehead *et al.*, 2009b; Ritson *et al.*, 2014; Watts *et al.*, 2015a; Hutchins *et al.*, 2016). The nature of that change is not likely to be uniform, with catchment characteristics such as basin permeability, soil types, land cover proportions, topography, and pollution sources all contributing to variation (Whitehead *et al.*, 2009a). Key fundamental process will remain largely similar across the UK however, and the implications of climate change on these key processes is discussed below.

Natural Resources Wales (NRW), in its latest assessment of the ecological status of rivers for the Water Framework Directive (WFD), identified a total 1,352 reasons why sections of river in Wales were not achieving good ecological status under the WFD classifications (NRW, 2019a). When investigating the causes behind these failures, it has been found that 47.6% of reasons for not achieving a good status is due to diffuse pollution sources, with 30.3% being due to point sources (NRW, 2019a). Looking at the west Wales river basin district, which encompasses all of the catchments studied in this thesis, 733 reasons are listed for river sections not achieving good ecological status. Both diffuse and point sources of pollution cause a greater proportion of total failure reasoning for this region than the full Wales dataset, at 48.7% and 34.1% respectively (NRW, 2019a). Changes to flow regime (e.g. water impoundment or abstraction) have a similar proportion of impact for both regions, as do natural causes, such as natural barriers to fish migration or natural streamflow conditions. Physical stream modification, for the purpose of river navigation, for example, or due to construction, has a larger impact when looking at the whole of Wales, than for the west Wales region, potentially reflecting the greater level of urbanisation represented in the all Wales dataset. Table 2.2 provides an overview for all of Wales and the west Wales river basin district on the breakdown of reason for stream sections not achieving good ecological status.

Desser	All of	Wales	West Wales		
Reason –	Number	Percent	Number	Percent	
Diffuse source	644	47.6%	357	48.7%	
Point source	409	30.3%	250	34.1%	
Physical modification	176	13.0%	57	7.8%	
Flow	52	3.8%	22	3.0%	
Natural	32	2.4%	18	2.5%	
Other	39	2.9%	29	3.9%	

**Table 2.2.** Categorisation of reasons for stream sections not achieving good ecological status in the Water Framework Directive classifications, for all of Wales, and the west Wales river basin district (NRW, 2019a).

Jin et al. (2012) found that for the River Thames, nitrate concentrations increased considerably in the winter months when modelling the period 2070-2099 compared to observations for 1961-1990, with a decrease shown in summer. This is commensurate with previous research that suggests larger winter flows will cause greater flushing of nitrate from land to river systems (Whitehead et al., 2009a). Furthermore, there is evidence to suggests that longer residency times and high water temperature will cause greater loss of nitrate through de-nitrification in rivers (Whitehead & Williams, 1982), explaining the lower summer concentrations. A similar trend was found by Conlan et al. (2007) for nitrate and ammonia in the River Wharfe catchment, with levels falling under drier summer conditions. There is the risk, however, that after prolonged periods of drought (greater than a year), the effect of the return of rainfall could result in exceptionally high peaks of nitrogen in rivers systems, as projected for the River Kennet by Wilby et al. (2006). Hutchins et al. (2016) review changes in river quality for the UK also, they suggest that the use of combined sewer overflow systems, which are common in the UK and especially in Wales, with over 3,000 (DCWW, 2015), are likely to cause an increase in particulate phosphorus, especially during times of high flow, as well as increases in sediment transport. Increases in the concentration of future suspended sediments is supported by Bussi et al. (2016), who showed that under extreme precipitation events, suspended sediment yields in the River Thames increased substantially, due to greater splash erosion, greater overland flow, and larger channel flow. Additionally, Jennings et al. (2009) showed for the River Flesk (Ireland), independent of factors such as combined sewer overflows, that phosphorous levels are to substantially increase in the winter months, linked to higher precipitation levels causing greater delivery of contaminants to the river system. These trends fit with international literature such as Crossman et al. (2013), who found for the Black River in Canada, that total phosphorous loads in winter were greatly increased

under future climate projections, caused by larger precipitation volumes enabling greater transfer into streams. Nilawar & Waiker (2019) found that for the large Purna river basin in India, under future RCP8.5 conditions, sediment concentrations were set to increase fivefold when comparing 2054-2075 with a baseline period of 1980-2005; 50% of this rise was attributed to greater stream power causing riverbank erosion.

In general, increased SS loads have implications for other water quality factors, as pollutants such as phosphorous and bacteria adsorb to the grains, meaning that projected increases in SS concentrations will likely lead to higher levels of other pollutants (Dearmont et al., 1998; Loperfido, 2014). However, when looking at faecal indicator bacteria, such as coliforms, as an indicator of pathogen organism presence in freshwater, work has shown that the situation may be more nuanced. Dakhlalla & Parajuli (2020) show for the Big Sunflower River, that while increased rainfall causes greater washing of bacteria into rivers, increases in temperature cause faster die-off rates. These results are in agreement with other established research, such as Cho et al. (2010), Hofstra (2011), and Wilkes et al. (2011), which have investigated this relationship in various regions and catchment types globally. In addition, Bussi et al. (2017) show for the Conwy (also studied in this thesis) a decline in future faecal indicator bacteria even under an intensified agricultural scenario. Their research compares observed 1960-1990 bacteria levels with modelled data for 2060-2090 using the INCA-Pathogens model, under the SRES A1B scenario. The declines observed are also attributed to combined temperature increases, causing faster bacterial die-off, and precipitation declines, especially in summer, causing a reduction in runoff and therefore contaminant routing (Bussi et al., 2017). It should be noted however, that Bussi et al. (2017) assess monthly average changes, and therefore projected increases in the magnitude and frequency of large streamflow events (Kay et al., 2014a), which could cause a greater number of spikes in bacterial concentrations, are not accounted for. None of the studies mentioned explicitly account for combined sewer overflows either, these could be especially important for the study region of this thesis, when considering the large number in operation, and the potentially contribution to contamination from bacteria in times of high flows (Astaraie-Imani et al., 2012; Abdellatif et al., 2015). Therefore, while average pathogen concentrations are likely to decline under future climate change, it is also likely that larger spike will be seen, this could prove problematic for water treatment works, as well as in-stream ecology, and sectors such as the shellfish industry.

A further consideration for future water quality is the impact of changing (mainly increasing) air temperatures on stream water temperature and associated effects on water quality factors such as dissolved oxygen (DO). The relationship between air and river water temperature is well established and has been studied globally (Crisp & Howson, 1982; Preud'homme & Stefan, 1992; Erickson & Stefan, 2000; Morrill et al., 2005; Harvey et al., 2011). Morrill et al. (2005) estimated a 0.6°C to 0.8°C rise in water temperature for every 1°C rise in air temperature when studying 43 rivers in 13 countries, they did however note that an S-shaped relationship better captures changes at air temperatures close to zero and above 25°C. This S-shaped relationship is also suggested as the best fit by Harvey et al. (2011) who studied four sample sites in Newfoundland. Erickosn & Stefan (2000) also concluded that the relationship between the two factors depart from linearity after 25°C, for two catchments in the US; the relationship is also closer when studied as a weekly or monthly average, rather than daily (Erickson & Stefan, 2000; Webb et al., 2003; Morrill et al., 2005). The effect of water temperature on DO has also been well researched, with increased temperature causing a decrease in DO concentrations (Morrill et al., 2005; Nilsson & Malm Renöfält, 2008; Cox & Whitehead, 2009; Harvey et al., 2011). Future warming of summer temperatures in particular could therefore cause DO concentration to fall to critically low levels, potentially impacting in-stream ecology, threatening various aquatic life (Morrill et al., 2005; Cox & Whitehead, 2009; Harvey et al., 2011). Eutrophication also has the potential to be altered, especially when combined with increasing nutrient supply, as is expected due to increased overland flow and associated flushing of nutrients from soils into river systems (Nilsson & Malm Renöfält, 2008; Watts & Anderson, 2016). With increased nutrient supply, and chemical reactions running more quickly at warmer temperatures, there is the potential for a large increase in the size and development speed of algal blooms in the UK (Whitehead et al., 2008, 2009a, 2009b; Cox & Whitehead, 2009; Watts & Anderson, 2016). This increase in nutrient supply, temperature, and algal volume is likely to increase the biological oxygen demand of river systems, putting yet further pressure on DO levels (Whitehead *et al.*, 2008). These problems could be further exacerbated by the potential for long periods without rain in the summer in future, followed by intense precipitation events, further increasing spikes in poor water quality (Delpla et al., 2009; Whitehead et al., 2009a; Watts & Anderson, 2016).

It is therefore likely that river systems in Wales will see marked changes in water quality under future hydroclimatic change. It is crucial to understand the nature of this change in order to plan for and mitigate its effects. Therefore, Chapter 5 assesses future seasonal changes in the concentrations of four water quality parameters: suspended sediment, total phosphorous, total nitrogen, and dissolved oxygen.

## 2.5 Future water resource use implications for Wales

Water resources in Wales are overseen and regulated by NRW on behalf of the Welsh Government. NRW manages and monitors abstraction, impoundment and discharge licences for water use in the country as well as being tasked with monitoring aspects such as water quality and pollution (NRW, 2020a). The largest exploitation of natural water resource in Wales is for PWS, which for Wales is mainly under the jurisdiction of Dŵr Cymru Welsh Water (DCWW). In the past two decades the amount of water supplied by DCWW has reduced by about a fifth, to ~800 million litres per day, down from ~1000; approximately half of this reduction is due to a lowering in the amount of leakage in the system; the rest because of falling demand from heavy industry (DCWW, 2019a). This reduction is despite an increase in average consumption per capita from 110 litres per day in the 1970s to 157 litres per day in 2016 (DCWW, 2016b). Wales is highly reliant on surface waters for PWS, in fact roughly 95% of supply sources used by DCWW are taken from surface waters (DCWW, 2016c), mostly from 65 impounding reservoirs owned and operated by the company in Wales (DCWW, 2019a). While only 3% of water falling as rain is abstracted for PWS in Wales, compared to figures nearer 50% in south and east of England, the country still faces potential water supply issues. This heavy reliance on surface waters and lack of back-up ground water supplies, makes Wales more vulnerable in times of prolonged severe drought. Indeed, there are already regions at risk of water deficit in Wales during dry summer months, a situation compounded by a relative lack of transfer capabilities between water supply zones (DCWW, 2019b).

Currently the industry regulator for England and Wales, Ofwat, and the Water Act 2003, require water companies to develop a water resources management plan, in which companies plan the supply of water into the future (HM Government, 2017). The most recent for DCWW covers the period 2020-2045, but it should be noted that this is a snapshot of thinking at the time, and that actual planning is continuous (DCWW, 2019a). The latest water

resources management plan for DCWW states an expected drop in the amount of water available for abstraction, both due to reductions in abstraction license amounts by NRW and climate change. In addition, total water demand is predicted to fall by DCWW, based on targets for reductions in leakage; individual large customer usage predictions and development forecasts for Wales (DCWW, 2019a). With the majority of water being taken from large storage reservoirs for the most of Wales, and given the trends previously mentioned of increasing winter precipitation and declining summer precipitation, combined with longer and more intense droughts, means that the water supply systems could still come under strain. Increasing precipitation in winter, for example, will do little to combat shortages in the summer and autumn due to reduced filling, if reservoirs are already at maximum capacity in the proceeding winter.

Climate change not only has the ability to impact on the supply of water, but also water demand. Despite projections for an overall decline in demand by DCWW, it is likely that, as is true for precipitation, demand patterns will become more exacerbated. Domestically, increasing summer temperatures are likely to lead to a greater amount of water usage for personal bathing and gardening, with Downing et al. (2003) showing, under 2.3°C warming to 2050, a 7% (winter) to 21% (summer) increase in domestic demand. Goodchild (2003) demonstrated a 3.3 litre per person increase (range 1.3-5.7) in summer domestic water demand, equating to 2.1% increase, when studying the Essex water supply zone under future increased temperatures. It was also noted by Goodchild (2003) that peak water demand also increases, with a 7-day domestic water demand average of 183 litres per person occurring 10% of the time under future scenarios, compared to 5% at the time of study. Xenochristou et al. (2020) found that sunshine duration and air temperature are the two most important weather based controls on domestic water demand. During week days, a one hour increase in daily sunshine duration related to a 6 litre per person increase in water use; for daily air temperature, a rise of between 2.5 and 7.5 litres per person was seen per 1°C warming after 15°C, depending on the affluence of the water user (more affluent increasing the greatest; Xenochristou et al., 2020). Furthermore, likely increases in temperature and decreases in rainfall have the potential to generate much greater demand for water from the agricultural sector in the UK, for irrigation of crops and water supply for animals. Downing et al. (2003) for example showed an average 84% increase in water demand for potato production in the

UK between by 2050 under a high future emissions scenario, or 74% under a medium scenario; with potatoes being one of the most water intensive crops to grow.

Looking more widely at general water use and abstraction by all sectors, it is clear that here too there are substantial implications in terms of water availability, especially in summer months. Although very much a worst-case scenario, Figure 2.8 highlights two potential future problems for catchments and water users in Wales when looking at total abstraction demand. The scenario presented is set during low flow conditions under RCP8.5 levels of future emissions, with an increased population (ASC, 2016b). First, future demand is set to outpace supply significantly in this scenario, with substantially less water available than is needed across the UK but especially so in Wales. Second, and more important environmentally, is the lack of adequate streamflow to satisfy even minimum environmental flows. These flow levels are often assigned to water abstractors, to ensure a sufficient flow is maintained to protect the ecological and morphological status of the river. Insufficient flow to maintain these levels could have serious implications for the health of the river and its ecology.



**Figure 2.8.** UK abstraction demand in the 2050s (left) and 2080s (right) as a percentage of available discharge during low flow conditions, under high population growth and RCP8.5 equivalent future scenario. Maroon colour denotes areas of negative available resource, i.e. those areas where the available flow is not sufficient to satisfy required environmental flow, meaning no abstraction is possible (ASC, 2016b).

Additional pressure is also potentially faced by public water supplies in the form of changing water quality, as previously discussed. Nitrate peaks as projected by Wilby *et al.* (2006) for the River Kennet, for example, would be greater than the legal limit acceptable by the UK for drinking water supply, so would require additional treatment methods to be in place. This is a trend likely to be seen across the country; with decreasing water quality there is an increased environmental and financial cost in order to maintain drinking water standards. If higher levels of water pollution were seen consistently, such changes may even require the introduction of new technologies or drinking water treatment plant upgrades, once again this would be potentially highly costly.

Planning of future PWS resources has been under consideration for some time in the UK, Arnell (2004b) for example, assessed that, given a projected 30-50% reduction in Q<sub>95</sub> by 2050 and higher pressures on the supply-demand balance due to both higher consumer demand, and reduced supply availability (lowered abstraction license levels due to Water Framework Directive), a significant difficulty could be posed for the industry. In addition, Arnell and Delaney (2006) suggested that a potential 20% increase in water demand for irrigation could have the most severe knock on effects for supply sustainability; Downing et al. (2003), however, suggests peak domestic demand has the potential to be the most important factor in future water supply planning. However, work is still ongoing and needed, with ensuring future water supply often touted as one of the key challenges facing Wales and the UK under a changing climate. The Adaptation Sub-Committee (ASC) of the Committee on Climate Change, for example states that 'more action is needed' to tackle and understand the risks to public water supplies in Wales, the highest level of urgency it gives (ASC, 2016b). Furthermore, the Welsh Government's climate change adaptation plan (Welsh Government, 2019a) states that risks to public water supplies from drought and low flows is one of four key risk areas in relation to climate change. Finally, DCWW in their 2050 plan (DCWW, 2018) state that reduction in available water and water quality change are two key future trends that require planning and adaptation, to safeguard clean drinking water for all.

Work reported in Chapter 5 of this thesis addresses potential changes in streamflow and water quality, which could be important for all water users in the future. Furthermore, Chapter 6 address future water resource availability in Wales for both PWS and HEP, analysing trends in unmet demand for PWS and generation potential for HEP.

### 2.6 Summary

As can be seen by the research presented, changes in the quantity and timing of water supply historically and for the future have been relatively well studied for the UK as a whole, but less well for Wales specifically. Wales is however an important area to study, with evidence in many of the papers cited in this section detailing how the west of the UK and Wales is more likely to see a greater impact of future changes in both average and extreme hydroclimatic conditions. Furthermore, the impacts of projected climate change on water quality parameters for Wales has seen even less research, as has future water resource availability. In the light of this, as well as global, UK and Welsh level policy, and the heavy reliance on water resources in Wales, as outlined in Chapter 1, it is important to have a good understanding of how climate in Wales will change and the impacts that such changes will have on hydrology. Furthermore, given the dependence on surface waters, greater knowledge of how industries that are heavily reliant on water will be affected is crucial. While water resource management plans published by DCWW do provide important insight into the impacts on water supply, there is scope for further research assessing the influence of changing climate on water supply and demand relationships, the bearing of extreme events, and the impact of changing water quality. Likewise, the impact on other large water users has seen limited research, especially considering the consequences that changes in the seasonality and quantity of streamflow could cause. It is these knowledge gaps that this thesis aims to address, as laid out in the aim and objectives section in Chapter 1. It is hoped that by addressing the gaps relating to (1) water demand links to prevailing weather conditions, (2) future streamflow and water quality characteristics, and (3) the impacts on water abstraction locations, that more informed future planning decisions may be made.

# CHAPTER 3: METHODOLOGICAL OVERVIEW

# 3.1 Overarching methodology

The research conducted in this thesis is based on a process of results, with the findings of the earlier studies informing the research direction and methods of latter studies. While the objectives of the research were established at the project outset (Chapter 1), the overarching and individual study methodologies have been continuously developed based on the findings. This approach has been taken in order to ensure that the most appropriate methods are used given the output data, and that the results are therefore meaningful, reliable, and useful. The same study area and five catchments (details in Section 3.2) have been maintained throughout the thesis, allowing a full picture of change within each to be established.

The first study investigates historical trends in temperature, precipitation, and streamflow for each catchment, in order to establish the current direction and magnitude of change in the variables. In addition, the relationship between the variables and total water demand, using total public water supply (PWS) abstraction volume as a proxy, is also completed. This study was undertaken in order to characterise for the first time the historical trends in both average and extreme streamflow, precipitation, and temperature for several of the catchments. This characterisation is also then used to inform the methodology of the future hydrological modelling completed in Study 2, as well as to provide context and comparison with the future observed trends. In addition, the establishment of the relationship between weather and total PWS demand has not been made for Wales previously, with little work of this kind being published for the UK more widely. This work is important for the assessment of potential future PWS demand generally, with the results, for temperature in particular, also important for the future water resource availability analysis completed in Study 3.

Study 2 analyses the impact of future climate change on the streamflow regime and water quality of the five catchments. Projected future climate input is based on a worst-case scenario of future global greenhouse gas (GHG) emissions, in the form of Representative Concentration Pathway 8.5 (RCP8.5) from the Intergovernmental Panel on Climate Change (IPCC). RCP8.5 represents one of the worst emissions outcomes of a no-policy scenario, with high population growth, continued coal use, and no downturn in global GHG emissions (Riahi *et al.*, 2011; van Vuuren *et al.*, 2011), leading to a scenario of radiative forcing still rising by 2100, and not peaking until the 2200s (Moss *et al.*, 2010; van Vuuren *et al.*, 2011). This is an

extreme future scenario, in the 90<sup>th</sup> percentile of future modelled outcomes of no climate change mitigation (van Vuuren *et al.*, 2011) and has been used in this thesis despite the higher levels of uncertainty associated with it, due to the critical nature of the study subject, in particular PWS. To ensure an uninterrupted future PWS it is vital to take account of such a scenario, to allow for robust planning. Furthermore, the use of this scenario is important for all potential water users, hydroelectric power (HEP), for example. In a future society where renewable energy is likely to make up a larger part of the overall energy mix, understanding the potential fluctuation in HEP generation caused by alteration in streamflows under climate change is highly important. The same is also true for purposes such as abstraction for agricultural irrigation, also likely to become more important under a warmer and drier future summer climate in the Wales. The results of Study 2 not only inform Study 3, but also, as aforementioned, have significance to any abstractors or users of, surface waters in Wales. Changes in streamflow and water quality will impact on all water users, as well as aquatic life, the results of this work therefore have wide reaching consequences, providing context to the decision and policy making of various organisations.

The third and final study uses the daily future streamflow output modelled in Study 2 to investigate the impact of climate change on available water resource for two of the key water abstracting sectors in Wales, HEP and PWS. These two sectors were chosen based on their year-round water use, streamflow allowing (as opposed to the more seasonal nature of agricultural demand, for example), as well as their importance to society, for water supply and energy generation. The study was undertaken in order to provide a degree of quantification of the potential societal, environmental, and economic impacts of climate change induced hydrological regime change in the future, as well as to provide context to future decision making in terms of preparation and adaptation.

## 3.2 Study catchments

Five catchments in Wales have been chosen for study, the Clwyd, Conwy, Dyfi, Teifi and Tywi (Figure 3.1); with each exhibiting varied characteristics such as topography, hydrometry, land cover, and climate. They are all also used extensively for water abstraction for a variety of sectors, such as PWS, HEP, agriculture and industry. The rationale behind the selection of Wales as the study area, as well as these five particular catchments, is provided in Section 1.2.



**Figure 3.1.** Study catchments displaying streams larger than third order and catchment elevation. The location of streamflow gauges used in all three studies are shown (green circles), as well as the location of pseudo-meteorological stations (orange octagons) and water quality monitoring locations (pink triangles) used in Study 2. The 25 hydroelectric power abstractions locations examined in Study 3 are also shown (blue stars).

The following section provides detailed descriptions of each catchment and a comparison of characteristics between catchments. Reported details and figures are based on the catchment areas as modelled for Study 2. Morphometric calculations are based on data derived from the OS Terrain 5 digital elevation model (DEM) from Ordnance Survey (Ordnance Survey, 2020); land use/land cover (LULC) proportions are calculated from 2012 CORINE Land Cover data from the European Environment Agency (EEA, 2012). Gauged streamflow data (NRFA, 2020), and precipitation data from the Climate, Hydrology and Ecology research Support System (CHESS) dataset (Robinson *et al.*, 2017), for the hydrological years 1982 to 2015, has been used in the calculation of hydrometric signatures and precipitation statistics. In relation to hydrometric signatures, mean streamflow (Q) and precipitation 3.1. Values closer to 1 denote a streamflow dominated catchment, and nearer to 0 signals evapotranspiration is foremost:

$$R_{QP} = \frac{Q}{P} \tag{Eq. 3.1}$$

Baseflow index was calculated by the digital filter method (Sawicz *et al.*, 2011) to give an indication of the long-term contribution of baseflow to total streamflow. Direct runoff at a given time-step ( $Q_{Dt}$ ) is first computed, as per Equation 3.2:

$$Q_{Dt} = cQ_{Dt-1} + \frac{1+c}{2}(Q_t - Q_{t-1})$$
 (Eq. 3.2)

where c is a constant with value 0.925,  $Q_{Dt-1}$  is direct runoff at the previous time-step (at t = 0,  $Q_D$  is assumed to be 0) and  $Q_t$  and  $Q_{t-1}$  are streamflow at this and the previous time-step respectively. Baseflow ( $Q_{Bt}$ ) for the time-step is then calculated using Equation 3.3:

$$Q_{Bt} = Q_t - Q_{Dt} \tag{Eq. 3.3}$$

Baseflow index ( $I_{BF}$ ) can then be computed by Equation 3.4, summing all time-step values, where  $I_{BF}$  values closer to 1 infer catchments with higher baseflow contribution:

$$I_{BF} = \sum \frac{Q_B}{Q}$$
 (Eq. 3.4)

Finally, streamflow elasticity ( $E_{QP}$ ) has been calculated as a measure of the sensitivity of each catchment to precipitation, as established by Schaake (1990) and Dooge (1992), and modified

by Sankarasubramanian *et al.* (2001), which compares the percentage change in precipitation to the percentage change in streamflow, normalised by the runoff ratio, as in Equation 3.5:

$$E_{QP} = median\left(\frac{dQ}{dP} \times \frac{P}{Q}\right)$$
 (Eq. 3.5)

where dQ and dP represent the difference between the previous year's and current year's mean annual streamflow and precipitation respectively, and P and Q are as above. Values greater than 1 represent elastic catchments which are sensitive to precipitation changes, with values less than 1 denoting the opposite, inelastic catchments. A summary of catchment characteristics is made in Table 3.1, with detailed descriptions for each catchment after.

**Table 3.1.** Summary of comparative study catchment details based on the catchments as delineated in Study 2. Morphometric parameters derived from the 5 m resolution OS Terrain 5 DEM from Ordnance Survey (Ordnance Survey, 2020) with hydrometric signatures based on streamflow for the hydrological years 1982-2015 (NRFA, 2020). Precipitation statistics calculated from CHESS data (Robinson et al., 2017); land use/land cover derived from 2012 CORINE Land Cover data (EEA, 2012).

		Clwyd	Conwy	Dyfi	Teifi	Tywi
orphometric   hydrometric arameters	Catchment area (km <sup>2</sup> )	750.1	541.8	507.2	995.3	1364.6
	Mean elevation (m)	210.4	309.8	271.4	198.1	212.5
	Maximum elevation (m)	557	1062	903	591	801
	Mean slope (%)	12.5	19.7	25.3	11.6	16.6
	Runoff ratio	0.52	0.93	0.89	0.76	0.80
ы д Б	Baseflow index	0.59	0.28	0.38	0.54	0.48
	Streamflow elasticity	1.28	1.21	1.06	1.09	1.17
2	Mean annual rainfall (mm)	977	1968	1803	1398	1596
tio	Winter proportion (%)	28.4	32.1	31.2	30.2	30.8
oita	Spring proportion (%)	20.1	19.2	18.8	18.8	19.1
scip	Summer proportion (%)	21.0	18.2	20.1	19.8	19.7
Pre	Moan rain free days per year	50.5 10 2	50.5 AA 1	29.9 ED 1	51.2	30.4 40 E
		46.5	44.1	52.1	55.7	49.5
	Urban	2.4	0.7	0.3	0.8	0.7
	Agriculture	80.9	30.6	30.0	83.0	64.6
	Arable	9.3	0.1	0.4	1.8	0.7
use/	Pasture	71.6	30.5	29.6	81.2	63.9
	Forest	7.3	13.7	21.5	5.5	16.0
ри <u>%</u>	Broadleaf	1.6	2.7	1.8	1.2	1.0
t la ver	Coniferous	4.5	5.5	18.9	3.2	10.1
Catchment land co	Mixed	1.2	5.5	0.8	1.1	4.9
	Scrub	9.4	42.1	45.3	9.7	17.1
	Natural grassland	2.3	16.6	30.4	7.2	8.4
	Moors and heathland	6.6	23.6	11.3	1.2	6.5
	Transitional woodland scrub	0.5	1.9	3.6	1.3	2.2
	Peat bog	0.0	8.4	2.6	0.9	1.3
	Sparsely vegetated areas	0.0	4.1	0.2	0.0	0.0

The mostly easterly of the catchments, the Clwyd, is the third largest in terms of overall catchment area, with its headwaters in the Clocaenog forest and outlet to the Irish Sea at Liverpool Bay. It is largely agriculturally dominated (80.9%), with the majority of this being pasture (as is common in all catchments) at 71.6%; the largest proportion of land classed as urban (2.4%) is also found here. The Clwyd has the lowest maximum elevation of all of the catchments at 557 m, this is almost half the height of the same measurement in the neighbouring Conwy catchment; and is also the second least sloping on average, at 12.5%. In terms of hydrometric signatures, out of all of the catchments, the Clwyd has the lowest runoff ration (0.52), and conversely the highest baseflow index (0.59) and streamflow elasticity (1.28). These results show that the Clwyd is the most baseflow dominated of the catchments and has the largest proportion of total incoming precipitation exiting by evapotranspiration. The catchment is, however, still sensitive to changes in precipitation, of which it receives the least of all the catchments, the annual average being 977 mm. This precipitation has the most even split between seasons of all catchments, with autumn seeing the most (30.5%) and spring the least (20.1%).

The Conwy also flows into the Irish Sea on the north Wales coast, from its source in Snowdonia; indeed, the Snowdonia National Park area covers some 70% of the catchment. This mountainous terrain gives the Conwy the second largest mean slope value at 19.7% and the highest maximum catchment elevation, at 1062 m. This is also one of the highest rainfall areas in Wales, with the catchment receiving 1968 mm of precipitation per year on average, with only 44.1 rain free days per year, the lowest of those studied. The Conwy has the largest divergence in seasonal rainfall distribution, with 32.1% occurring in winter, compared to 18.2% in summer. Both the Conwy and the Dyfi differ from the other three catchments in that the main LULC is not agriculture. For the Conwy, scrub is the largest category (42.1%), with this being mainly moors and heathland (23.6%), and to a lesser extent, natural grasslands (16.6%). The Conwy also features the largest area of peat bog (8.4%), and, due to the mountainous terrain, 4.1% of land is sparsely vegetated. When looking at the catchment hydrometrics, it can be seen that these correspond with the morphometric, LULC and precipitation characteristics, giving the Conwy a flashy response. Runoff ratio is the highest of all catchments (0.93) and baseflow index the lowest (0.28), with streamflow elasticity the second highest, at 1.21, meaning the catchment is sensitive to changes in precipitation.

The Dyfi is the smallest catchment to be studied (507 km<sup>2</sup>) and is situated in west Wales, flowing from the Aran Mountains in the east, to the Irish Sea. It is similar to the Conwy in many regards, especially in terms of LULC proportions, precipitation volume and regime, and morphometric parameters. In terms of LULC, the Dyfi has the lowest amount of urban (0.3%) and agricultural (30.0%) land of any of the study catchments, but the highest amount of forested land, at 21.5% (1.8% broadleaf, 18.9% coniferous, 0.8% mixed). Scrub land is the largest proportion (45.3%), but in contrast to the Conwy, a larger part of this is natural grassland (30.4%) than moors and heathland (11.3%). The Dyfi has the second highest single catchment elevation point in its headwaters, at 903 m as well as the steepest average slope (25.3%). Runoff ration is second only to the Conwy, at 0.89, with baseflow index being second lowest (0.38), the streamflow elasticity score is the lowest of all catchments (1.06), but is still greater than 1, suggesting that for a given change in precipitation, a greater change is seen in streamflow. This gives the catchment a flashy nature, with little baseflow contribution, little loss to evapotranspiration, and being small and steeply sloping. Annual mean precipitation total is 1803 mm, the second largest, with 61.1% of this falling in winter and autumn; the Dyfi also has the second largest number of rain free days per year, 52.1.

The Teifi, in south Wales, flows into Cardigan Bay from its source at Llyn Teifi, in the Cambrian Mountains. The catchment is the second largest of those studied (995 km<sup>2</sup>), but also has the lowest maximum elevation, at 591 m, as well as the least steep mean catchment slope angle, 11.6%. The Teifi has the highest proportion of land classified as agricultural (83.0%) and conversely the lowest proportion of forest cover (5.5%) and scrub land (9.7%). These characteristics are similar to those seen in the Clwyd; it therefore follows that the hydrometric signatures of the Teifi are also similar to those seen in the Clwyd. Runoff ratio is 0.76 and streamflow elasticity is 1.09, both the second lowest of all five catchments; baseflow index is 0.54, the second highest in the study. As with the Clwyd, the characteristics of the catchment lead it to being less flashy in nature, in particular than the Conwy and Dyfi, but with streamflow still sensitive to changes in precipitation. The Teifi has the most rain free days of all catchments annually on average (55.7), with largest proportion of rainfall falling in autumn (31.2%) and least in spring (18.8%).

The Tywi, with its source roughly 15 km south of the Teifi's, is the largest catchment to be studied (1365 km<sup>2</sup>); flowing from its source in the Cambrian Mountains to Carmarthen Bay,

roughly 80 km of its length is designated as a special area of conservation. The river enters Llyn Brianne reservoir 10 km from its source; constructed in the 1970s the reservoir regulates the river in times of low flow, to ensure sufficient water for abstraction for PWS downstream. The majority of the Tywi is used as pasture (63.9%), with scrub and forested land being the next largest proportion of LULC (17.1% and 16.0% respectively). The catchment receives 1596 mm of rainfall annually on average, with 30.8% of this being in winter and 19.1% in spring; a mean 49.5 rain free days are experienced per year. The highest elevation in the catchment is 801 m, while the average slope is 16.6%, making it both the third highest and steepest catchment studied. Runoff ratio (0.80), baseflow index (0.48), and streamflow elasticity (1.17) values are all also mid-range when comparing the five catchments.

The bedrock geology of the catchments has been analysed using the British Geological Survey (BGS) Geology 625k (DiGMapGB-625) data set (BGS, 2019), with the majority in all catchments found to be mudstone, siltstone and sandstone in nature, ranging from 60% and 64% in the Clwyd and Conwy, to 87% in the Dyfi and Teifi, and 73% in the Tywi. The remaining bedrock in the Teifi and Tywi is made up of interbedded sandstone and conglomerate (13% and 27% respectively); this also makes up the second largest proportion of remaining bedrock in the other three catchments (Clwyd, 18%; Conwy, 17%; Dyfi, 12%). The majority of that remaining in the Dyfi and Conwy catchments is Felsic Tuff, at 1% and 12% respectively, while that remaining in the Clwyd is limestone with subordinate sandstone and argillaceous rock (16%), and siltstone and sandstone with subordinate mudstone (6%). While the bedrock of the Conwy, Dyfi, Teifi and Tywi has a low permeability, the limestone and sandstone deposits found under the Clwyd catchment are much more permeable (BGS, 2019).

As can be seen, the five catchments studied offer a variety of different catchment characteristics (morphometric parameters, hydrometric signature, LULC) which influence their hydrological regime, water quality, and response to precipitation. These characteristics and individual catchment response profiles could be important when considering future changes in climate and weather events, and the potential knock impacts this may have for catchments and water abstractors. As previously mentioned, each catchment is abstracted from for a variety of uses, these abstractions, except PWS, are shown in Figure 3.2 (PWS not shown due to data licence conditions prohibiting the identification of individual abstractions); with a breakdown in numbers by sector provided in Table 3.2.



**Figure 3.2.** Abstraction locations for the purposes of agriculture, amenity (e.g. land and pond throughflow), environment (e.g. wetland support), industrial/commercial, and energy production. Public water supply abstractions not shown due to dataset licence restrictions.

**Table 3.2.** Number of abstraction locations per sector for each catchment; 'water supply' includes both public and private water undertakings; 'energy production' accounts for both hydroelectric power and water used in other energy generation processes.

Catchment	Agriculture	Amenity	Environment	Industry/ commercial	Energy production	Water supply	Total
Clwyd	7	1		4	7	5	24
Conwy	2	2	1	5	20	5	35
Dyfi	4			4	23	11	42
Teifi	4	9		2	14	3	32
Tywi	2	3		4	10	3	22

## 3.3 Trend analysis

Long-term trend analysis of annual and seasonal datasets is a common thread throughout this thesis. In each of the three study chapters Mann-Kendall (MK) trend analysis (Mann, 1945; Kendall, 1975) has been used to detect the direction and statistical significance of trends in the meteorological, hydrological, water quality and water abstraction capability datasets studied. MK analysis is nonparametric and tests for a positive or negative monotonic trend in a time series based on the ranks of individual observations (Helsel & Hirsch, 2002). For a time series  $X = [x_1, x_2, ..., x_n]$ , MK is calculated following Equations 3.6 to 3.10. Firstly, the test statistic, *S*, is calculated by Equation 3.6:

$$S = \sum_{i < j} a_{ij} \tag{Eq. 3.6}$$

where  $a_{ij}$  is calculated as per Equation 3.7:

$$a_{ij} = sgn(x_j - x_i) = sgn(R_j - R_i) = \begin{cases} 1 & \text{if } x_i < x_j \\ 0 & \text{if } x_i = x_j \\ -1 & \text{if } x_i > x_j \end{cases}$$
(Eq. 3.7)

where  $R_j$  and  $R_i$  are the ranks of observations  $x_j$  and  $x_i$ . Under the assumption that the dataset is independent and randomly ordered, Mann (1945) showed that S tends towards normality when  $n \ge 8$ , with the mean and variance of S being calculated as per Equations 3.8 and 3.9 respectively (Kendall, 1975):

$$E(S) = 0$$
 (Eq. 3.8)

$$Var(S) = \frac{n(n-1)(2n+5)}{18}$$
 (Eq. 3.9)

where n is the length of the time series. By comparing the standardised variable u with the standard normal variate at significance  $\alpha$ , it is possible to test the significance of the trends, as in Equation 3.10:

$$u = \begin{cases} (S-1)/\sqrt{Var(S)} & for S > 0\\ 0 & for S = 0\\ (S+1)/\sqrt{Var(S)} & for S < 0 \end{cases}$$
 (Eq. 3.10)

However, the original MK test is vulnerable to false trend detection when dealing with autocorrelated data; that being data which displays a correlation with a lagged version of itself over a given time period (Hamed & Rao, 1998). This is common in hydrological and meteorological data due to the seasonality of the datasets. A modified MK test has therefore been performed, as per the Hamed and Rao autocorrelation correction method, the calculations for which are show in Equations 3.11 to 3.14 (Hamed & Rao, 1998). The variance of *S* is first recalculated as  $V^*(S)$ , as shown in Equation 3.11:

$$V^*(S) = Var(S) \times \frac{n}{n^*} = \frac{n(n-1)(2n+5)}{18} \times \frac{n}{n^*}$$
 (Eq. 3.11)

where  $n/n^*$  is a correction to Var(S) (Equation 3.9) accounting for autocorrelations in the data and is calculated using Equation 3.12:

$$\frac{n}{n^*} = 1 + \frac{2}{n(n-1)(n-2)} \times \sum_{i=1}^{n-1} (n-i)(n-i-1)(n-i-2)\rho_s(i)$$
 (Eq. 3.12)

where  $\rho_s(i)$  is the autocorrelation coefficient of the ranked data, expressed in Equation 3.13:

$$\rho_{s}(i) = \frac{\sum_{i=1}^{N-j} \left[ (x_{i} - \overline{X}) (x_{i+j} - \overline{X}) \right]}{\sum_{i=1}^{N} (x_{i} - \overline{X})^{2}}$$
(Eq. 3.13)

where N is the number of ranks of observations and  $\overline{X}$  is the dataset mean. The result of Equation 3.11 allows for a modification of Equation 3.10, enabling significance testing which reflects the autocorrelation correction (Equation 3.14):

$$u^* = \begin{cases} (S-1)/\sqrt{V^*(S)} & \text{for } S > 0\\ 0 & \text{for } S = 0\\ (S+1)/\sqrt{V^*(S)} & \text{for } S < 0 \end{cases}$$
(Eq. 3.14)

In addition, in order to better quantify the magnitude of the detected trends, Sen's nonparametric slope estimator (Sen, 1968) has also been applied. This procedure estimates the slope of a linear trend in a sample of n pairs of data. Firstly, the slopes of the entire time series of data points,  $Q_i$ , are calculated using Equation 3.15:

$$Q_i = \frac{X_j - X_k}{j - k} \text{ for } i = 1, 2, ..., n$$
 (Eq. 3.15)

where  $X_j$  and  $X_k$  are data values at time-steps j and k. All  $Q_i$  values are then ranked smallest to largest, with median slope,  $Q_{med}$ , then being calculated as Equation 3.16:

$$Q_{med} = \begin{cases} Q_{[(n+1)/2]} & \text{if } n \text{ is odd} \\ (Q_{[n/2]} + Q_{[(n+2)/2]})/2 & \text{if } n \text{ is even} \end{cases}$$
(Eq. 3.16)

A negative  $Q_{med}$  result indicates a negative trend, with the opposite true for a positive trend; the value of the result indicates the steepness of the trend, the median absolute value change per time-step. To calculate if the median slope is statistically significantly different from zero, Equation 3.17 is used:

$$C_{\alpha} = Z_{1-\alpha/2} \sqrt{V^*(S)}$$
 (Eq. 3.17)

where  $Z_{1-\alpha/2}$  is obtained from a standard normal distribution table and  $V^*(S)$  is as defined in Equation 3.11.

These methods for trend analysis and slope detection were chosen because of their nonparametric nature, allowing for the detection of trends in datasets which are non-normally distributed and that display seasonality, unlike trend analysis methods such as linear regression (Jaiswal *et al.*, 2015). This is important when dealing with hydrological and meteorological data which often displays the characteristics aforementioned (Kundzewics & Robson, 2004). This combination of modified MK trend analysis alongside Sen's slope estimator, is a method that has been used extensively for similar studies as well as more generally for the analysis of trends in hydrological, meteorological and water resources data globally (Gocic & Trajkovic, 2013; Murphy *et al.*, 2013; da Silva *et al.*, 2015; Atta & Dawood, 2017); it was therefore deemed suitable for use in this work.

For all three studies, trend analysis has been conducted on mean values based on hydrological years (1<sup>st</sup> October to 30<sup>th</sup> September) and each individual season. Results for any given year, 2021 for example, therefore are based on data from the following time periods; annual: October 2020 to September 2021; winter: December 2020 to February 2021; spring: March to May 2021; summer: June to August 2021; and autumn: September to November 2021. As can be observed, due to the nature of hydrological years ending in the middle of autumn, there is a mismatch between seasons and annual data, this has no impact however on the measurement and detection of trends seasonally or annually.

# 3.4 Study 1: Historical analysis

### 3.4.1 Hydroclimatic data

For the first study (Chapter 4), a 34-year period (1982-2015) of daily air temperature, total precipitation, and streamflow volume data has been analysed for each of the five study catchments. This time frame corresponds with the most recent period of complete and reliable data available for all three datasets across all catchments, at the time of analysis. Thirty years is classically considered as the minimum required dataset length in order to carry out climatic analysis (IPCC, 2007), not allowing smaller timeframe variations and cycles to impact on observed trends. This is also the period suggested for use by the World Meteorological Organisation (WMO) in the assessment of climate norms (WMO, 2017). Given this, the parity of study period timing for each catchment was prioritised over additional study period length for any given catchment or dataset. This decision was made to ensure cross-comparability of results for all catchments, enabling interpretation of the impact of catchment location and characteristics on the observed trends.

Daily climatic data studied (air temperature and total precipitation) was obtained from the CHESS dataset (Robinson *et al.*, 2017) from the UK Centre for Ecology and Hydrology (CEH). This gridded dataset was chosen due to its complete coverage across Wales, especially when compared to data from individual meteorological stations, of which there are few with long, complete, and consistent records in the study region, leaving some catchments underrepresented. CHESS precipitation data is based on the CEH 1 km Gridded Estimates of Areal Rainfall (GEAR) dataset, which is an interpolated dataset based on meteorological station rainfall observations adjusted for topography, to provide full UK coverage (Keller *et al.*, 2015). Temperature data for CHESS is downscaled to a 1 km grid, also taking account of topographical data, from the 0.5 degree gridded Climate Research Unit Time Series, version 3.21 dataset (CRU TS3.21), which is also based on meteorological station observations (Harris *et al.*, 2014). In order to provide a single dataset of each climatic variable for each catchment, the mean value of all grid cells contained with a catchment was calculated for each day of the study period; it is this value, after inspection for anomalies, that has been used in the analysis described in the following section.

Daily streamflow data has been obtained from the UK National River Flow Archive (NRFA) for a single velocity-area gauging station in each catchment, as shown in Figure 3.1 (NRFA, 2020). The stations used were those with as close to a natural flow regime as possible, as well as a long record length, to allow for a more robust attribution of streamflow changes to external factors, as opposed to regulation or abstraction. For each station, a review of the station information, hydrometric description, and flow record and regime descriptions was undertaken, to ensure the record was appropriate and check for any required correction to the data. This investigation showed that no correction or action was needed for any of the flow records used, details of each gauging station are provided in Table 3.3.

	Station ID	Upstream area (km²)	Station elevation (m)	Station opened	NRFA brief flow regime description
Clwyd	66001	404	15.3	1959	Low flows augmented by groundwater
Conwy	66011	345	11.6	1964	Largely natural
Dyfi	64001	471	5.9	1962	Natural to within 10% at $Q_{95}$
Teifi	62001	894	5.2	1959	Largely natural
Tywi	60006	130	7.5	1968	Natural to within 10% at $Q_{95}$

Table 3.3. Details of gauging stations used throughout the thesis (NRFA, 2020).

#### 3.4.2 Hydroclimatic analysis

In order to best characterise any changes in the climatic and streamflow data, analysis was performed annually and seasonally (as described in Section 3.3) for both average and extreme temperatures, precipitation volumes and flows. Analysed time series are based on the WMO standard methods for annual, seasonal and multi-month calculations (WMO, 2017). For each season and year, mean flow, temperature and precipitation has been calculated for the analysis of averages. For the extreme analysis, 1-day maximum and minimum flow and temperature, as well as maximum precipitation total, for each season and year has been extracted; this characteristic was chosen to give an indication of the change in magnitude of extremes during the study period. Additionally, the number of days above and below the dataset 95<sup>th</sup> and 5<sup>th</sup> percentiles respectively has been calculated for streamflow and temperature, with analysis of precipitation only taking account of the former. This analytical method was chosen to give an understanding of the change in frequency of extremes.

The above extracted time series were first analysed using the aforementioned MK analysis methods. Subsequently, a breakpoint analysis has been performed on each of the historical

average and extreme hydroclimatic times series, once again both seasonally and annually, to detect any abrupt changes during the study period. This analysis was performed in order to assist in the attribution of changes to known events that may have occurred in one or more catchments (e.g. damn construction or major LULC change or climate driver alteration). Two nonparametric methods of breakpoint analysis have been employed, Pettitt's test (Pettitt, 1979) and the Standard Normal Homogeneity Test (SNHT) method (Alexandersson, 1986; Alexandersson & Moberg, 1997). Pettitt's test has been shown to perform better at detecting changes in the middle of a time series, whereas SNHT performs more favourable at the start and end of a time series (Hawkins, 1977; Jaiswal *et al.*, 2015; Hänsel *et al.*, 2016). It was therefore decided to perform both tests to ensure the best coverage of potential breakpoints. Both methods detect changes in the value of the dataset mean and are location specific, therefore identifying the point of change within the time series, (Hänsel *et al.*, 2016).

Pettitt's test, for a time series  $x_1, x_2, ..., x_n$ , calculates the distribution function for time series data both before  $(F_1(x))$  and after  $(F_2(x))$  a breakpoint, t. The nonparametric test statistic,  $U_t$ , is calculated as per Equation 3.18:

$$U_t = \sum_{i=1}^t \sum_{j=t+1}^n sgn(x_t - x_j)$$
 (Eq. 3.18)

Where *n* is the time series length and  $sgn(x_t - x_i)$  is calculated by Equation 3.19:

$$sgn(x_t - x_j) = \begin{cases} 1 & if \ x_i < x_j \\ 0 & if \ x_i = x_j \\ -1 & if \ x_i > x_j \end{cases}$$
(Eq. 3.19)

Equation 3.20 calculates the test statistic K, which is subsequently used in the calculation of the confidence level ( $\rho$ ) for the given n, as in Equation 3.21, itself a term in the calculation of the approximate significance probability (p) of a breakpoint, in Equation 3.22:

$$K = Max|U_t| \tag{Eq. 3.20}$$

$$\rho = exp\left(\frac{-K}{n^2 + n^3}\right) \tag{Eq. 3.21}$$

$$p = 1 - \rho$$
 (Eq. 3.22)

The value of K is compared to a critical value based on the time series length, to determine whether the change is statistically significant (Jaiswal *et al.*, 2015).

SNHT compares the mean of the first n observations with the mean of the remaining observations (n - k), in the calculation of the test statistic  $T_k$ , as shown in Equation 3.23:

$$T_k = kZ_1^2 + (n-k)Z_2^2$$
 (Eq. 3.23)

where  $Z_1$  and  $Z_2$  are calculated by Equations 3.24 and 3.25 respectively:

$$Z_{1} = \frac{1}{k} \sum_{i=1}^{k} \frac{(x_{i} - \overline{x})}{\sigma x}$$
 (Eq. 3.24)

$$Z_{2} = \frac{1}{n-k} \sum_{i=k+1}^{n} \frac{(x_{i} - \overline{x})}{\sigma x}$$
 (Eq. 3.25)

where  $\overline{x}$  is the mean and  $\sigma x$  is the standard deviation, of the series.  $T_k$  is the point at which there is maximum difference in value of the means of the proceeding and succeeding time series points. The value of the test statistic is compared to a critical value based on the time series length, to determine whether the change is statistically significant (Jaiswal *et al.*, 2015).

#### 3.4.3 Water abstraction relationships

In addition to the analysis performed on the three hydroclimatic factors, an investigation of the relationship between these factors and total water demand has also been performed. The water demand data to enable this analysis has been obtained from Dŵr Cymru Welsh Water (DCWW) in the form of total daily abstraction for 22 abstraction points within the five study catchments. This daily total has been used as a proxy for overall PWS demand across all sectors and also takes account of leakage. The data was provided for just under five calendar years, from 1<sup>st</sup> January 2012 to 30<sup>th</sup> September 2016, however, due to the length of record available for the hydroclimatic data, analysis was only completed up to the 30<sup>th</sup> September 2015. The relationship between the three hydroclimatic factors and total abstraction has been characterised by both Pearson's & Spearman's rank correlation coefficients. This analysis has been conducted for both daily data, as well as monthly mean. Pearson's coefficient value (*r*) was computed using Equation 3.26:

$$r = \frac{\sum_{i=1}^{n} (hc_i - \overline{hc})(abs_i - \overline{abs})}{\sqrt{\left(\sum_{i=1}^{n} (hc_i - \overline{hc})^2\right) \left(\sum_{i=1}^{n} (abs_i - \overline{abs})^2\right)}}$$
(Eq. 3.26)

where  $hc_i$  refers to the daily/monthly average hydroclimatic variables studied,  $abs_i$  refers to daily/monthly total abstraction volume, and hc & abs represent the mean of the entire respective datasets. The nonparametric Spearman's correlation coefficient ( $\rho$ ) was calculated using Equation 3.27, a modified version of Pearson's correlation coefficient which calculates correlation between ranks, as opposed to raw data:

$$\rho = \frac{\sum_{i=1}^{n} \left( R(hc_i) - \overline{R(hc)} \right) \left( R(abs_i) - \overline{R(abs)} \right)}{\sqrt{\left( \sum_{i=1}^{n} \left( R(hc_i) - \overline{R(hc)} \right)^2 \right) \left( \sum_{i=1}^{n} \left( R(abs_i) - \overline{R(abs)} \right)^2 \right)}}$$
(Eq. 3.27)

where  $R(hc_i)$  refers to the rank of the daily/monthly average hydroclimatic variables studied,  $R(abs_i)$  refers to the rank of the daily/monthly total abstraction volume, and  $\overline{R(hc)}$  &  $\overline{R(abs)}$  represent the mean rank of the entire respective datasets. These two complementary methods have been used to test for a linear trend, in the case of Pearson's, and a monotonic trend in the case of Spearman's rank. It was important to characterise the trend between abstraction and temperature in particular at this stage, in order for the relationship to be used in the research conducted in Study 3 (Chapter 6) which required the projection of future water demand based on climatic conditions.

# 3.5 Study 2: Hydrological modelling

#### 3.5.1 Model selection

For the second study, future hydrological modelling of streamflow regime and water quality is the key focus. When selecting the hydrological model used for the study, several factors were taken into consideration while comparing multiple options. First, due to the size of the catchments to be studied, it was decided that a semi-distributed model was required, balancing greater computational efficiency, when compared to fully distributed models, with maintained representation of catchment characteristics, when compared with a lumped model (Suliman *et al.*, 2015). Second, the model needed to be operable at a continuous daily time-step in order to make use of the future climate input data, as well as enable analysis of
extreme streamflows and events. It was also important for the model to be physically based, with both land and in-stream processes being modelled, and the ability to handle both water quantity and quality data. Several well-established hydrological models were considered, including, but not limited to, APEX (Williams & Izaurralde, 2010), MIKE-SHE (Refsgaard & Storm, 1995), SWAT (Arnold *et al.*, 1998), TOPMODEL (Beven, 1997), and VIC (Liang *et al.*, 1994). The decision was made to proceed with the Soil and Water Assessment Tool (SWAT) model, using the interface with QGIS for catchment delineation and model setup. SWAT was chosen after a review of several studies comparing the performance of different hydrological models (Yang *et al.*, 2000; El-Nasr *et al.*, 2005; Golmohammadi *et al.*, 2014; Suliman *et al.*, 2015; Kauffeldt *et al.*, 2016), with SWAT performing favourably in these, especially for low flows which are important to the latter stages of the research, where future water availability is assessed. Additionally, SWAT is well supported by the model development team, offering trouble-shooting support as well as training workshops. This, coupled with the user-friendly nature of the model interface which eliminates the need for high level computer coding, was important, as the research was not being undertaken by an experienced modeller.

As was required by the model selection process, SWAT is a physically based, semi-distributed, continuous time-step model, which operates at a catchment scale. The model makes use of sub-basins, which are further sub-divided into hydrological response units (HRUs), these are lumped areas of unique LULC, soil, and slope configuration. The output from each HRU is routed to the corresponding sub-basin, and eventually the reach outlet (Neitsch et al., 2011). Hydrological processes modelled by SWAT are precipitation, evapotranspiration, infiltration, plant uptake, soil moisture redistribution, surface runoff, lateral flow, percolation to and revap from the shallow aquifer, return flow and recharge of the deep aquifer, these are shown in Figure 3.3 (Neitsch et al., 2011). In terms of in-stream processes related to water quality, the model allows for point sources of pollution to be added, as well modelling diffuse sources based on the LULC data provided. In addition, dissolved and particle transport is modelled, along with bed/bank erosion, sorption of pollutants onto sediments, biogradation and transformation, and dilution, diffusion, deposition, accumulation and resuspension of pollutants, as shown in Figure 3.4 (Neitsch et al., 2011). Owing to the extensive number of modelled processes, the continued development of the model over more than two decades, and its ability to operate in ungagged basins, SWAT has been used extensively for the

assessment of climate change impacts on water quantity (Qiao *et al.*, 2014; Coffey *et al.*, 2016; Perra *et al.*, 2018; Sultana & Choi, 2018; Yuan *et al.*, 2019) and quality (Nerantzaki *et al.*, 2016; Yang *et al.*, 2017; Pesce *et al.*, 2018; Jilo *et al.*, 2019) globally.



Figure 3.3. Representation of the processes modelled in the land phase of SWAT (Neitsch et al., 2011).



Figure 3.4. Representation of the processes modelled in the in-stream phase of SWAT (Neitsch et al., 2011).

#### 3.5.2 Model setup, calibration and validation

SWAT requires three key data inputs for initial model setup, these being elevation, LULC, and soil. In terms of elevation data, the OS Terrain 5 DEM from Ordnance Survey (Ordnance Survey, 2020) has been used. This high-resolution raster dataset, with a 5-metre post spacing (Ordnance Survey, 2017), was chosen in order to provide the most accurate catchment delineation and water routing as possible, as well as maintaining computational efficiency for the size of catchments studied. In terms of LULC, data from the CORINE Land Cover 2012 dataset (Copernicus Land Monitoring Data; EEA, 2012) was chosen due to the close match of LULC categories between it and the pre-exiting SWAT LULC categories, allowing for easy input to, and interpretation by, the model. The CORINE dataset is accurate spatially to within 25 metres, with a minimum mapping unit of 25 hectares for areal features and width of 100 metres for linear features; the accuracy of identified LULC type is greater than 85% (EEA, 2017). The dataset is generated by computer assisted photointerpretation of IRS P6 LISS III and RapidEye satellite data (EEA, 2017). Soil data was obtained from the EU soil database version 2.0, specifically the Soil Geographical Database of Eurasia, a 1:1,000,000 scale vector dataset (European Commission, 2004). The calculation of SWAT soil parameters was completed using the Pedo Transfer Function developed by Saxton & Rawls (2006), based on the characteristics associated with the given World Reference Base for Soil Resources (FAO, 1998) classification of each soil type. This dataset was used based on research and the suggestion of the SWAT model developers, having accurate and full coverage for the study area, being in a format that is easily transferable to SWAT, and having been used in the model successfully previously.

Initial setup of the SWAT model starts with catchment delineation from a DEM, this process first develops the stream network, based on the standard procedure (Martz & Garbrecht, 1992) of filling sinks in the DEM, calculation of flow direction using the D8 algorithm for each pixel, and calculation of flow accumulation; a stream formation threshold of 1 km<sup>2</sup> was used at this stage. The user then adds any additional inputs of water, reservoirs, pollution point sources and watershed outlets. For the purposes of this research, reservoirs have been ignored, partly due to the gauging stations used for model calibration being located on streams that are largely unaffected by upstream regulation, and partly due to the complex nature of reservoir operation in SWAT, for which insufficient data was available to properly

model. Outlets can operate as water abstraction locations, aside from the furthest outlet downstream, which is classified as the final watershed outlet, for this reason, outlets were added at all PWS abstraction and HEP inlet locations in each catchment, to later enable analysis of future water abstraction potential at each location. Watershed boundaries are then delineated by SWAT, followed by subbasins, and then by HRUs therewithin, which are defined based on homogenous areas of LULC, soil and slope.

Historical weather data was added to the model to enable an initial SWAT run, this was then followed by calibration. The historical daily temperature and precipitation data used was the same as that for Study 1, from CHESS, here used for the period 1982 to 1998, with a 3-year model warm-up period. Weather data input for the model was based on single daily values of temperature and precipitation for each catchment, based on the mean of the gridded data, as before. Calibration was completed for each catchment individually, for streamflow only, using NRFA historical daily streamflow, also as in Study 1. The particle swarm optimisation (PSO) method was used to perform a multi-parameter calibration in the SWAT calibration and uncertainty programme 2012 (SWAT-CUP). A total of 17 parameters (Table 3.4) were calibrated simultaneously, these were selected based on a literature review of studies operating in similar catchment types, as well as on the advice of SWAT model developers.

Parameter	Description	Input file location
ESCO	Soil evaporation compensation factor	.bsn
EPCO	Plant uptake compensation factor	.bsn
SURLAG	Surface runoff lag time	.bsn
GW_Delay	Groundwater delay	.gw
Alpha_BF	Baseflow alpha factor	.gw
GWQMIN	Threshold depth of water in shallow aquifer for return flow to occur	.gw
RCHRG_DP	Deep aquifer percolation fraction	.gw
REVAPMN	Threshold depth of water in shallow aquifer for "revap" to occur	.gw
GW_REVAP	Groundwater "revap" coefficient	.gw
ALPHA_BF_D	Baseflow alpha factor for deep aquifer	.gw
CANMX	Maximum canopy storage	.hru
CN2	SCS runoff curve number for moisture condition 2	.mgt
CH_N2	Manning's "n" value for the main channel	.rte
СН_К2	Effective hydraulic conductivity in main channel alluvium	.rte
SOL_AWC	Available water capacity of the soil layer	.sol
SOL_K	Saturated hydraulic conductivity	.sol
SOL_Z	Depth from soil surface to bottom of layer	.sol

 Table 3.4. SWAT parameters calibrated through SWAT-CUP using the particle swarm optimisation method.

PSO is a complex but highly efficient method of calibration, capitalising on swarm intelligence which takes inspiration from flocking birds (Kennedy & Eberhart, 1995). In this example, a flock of birds is searching a set area for a single food source, while no bird knows the exact location of the food, they are all aware of their individual distance from it, as well as the closest themselves and the group as a whole has come to it. The optimal solution for efficient identification of the food source is to learn from the group, travelling in the direction of the location identified as closest to the food (Lee et al., 2008; Abbaspour, 2014). This theory can be applied to parameter calibration, with birds being substituted for particles searching for the optimal parameter value within a given range (search space), to satisfy a goodness-of-fit metric (Qi et al., 2015). The group is initialised with each particle in the swarm randomly assigned a position and velocity, the initial locations are evaluated for model fit, and these automatically become each particles' current best solution. The group's best solution is set at the value of the particle with the best fit of all particles. The individual particles then iteratively work towards, with a random function, a combination of their own and the groups best known position (Qi et al., 2015). For example, for a given particle, a, at a given iteration, *i*, the velocity (*v*) for the next iteration  $v_a^{i+1}$  is updated by Equation 3.28 (Lee *et al.*, 2008):

$$v_a^{i+1} = wv_a^i + c_1r_1(p_a^i - x_a^i) + c_2r_2(p_g^i - x_a^i)$$
 (Eq. 3.28)

where w is an inertia factor for control of movement;  $c_1$  and  $c_2$  are acceleration coefficients which attract the particle in the direction of the local and global best positions respectively; and  $r_1$  and  $r_2$  are randomly generated, uniformly distributed variables between 0 and 1. The current velocity and position of particle a is represented by  $v_a^i$  and  $x_a^i$ , with  $p_a^i$  and  $p_g^i$ representing the location of the current best found position of a and the group (g), respectively. The position of a is then updated by Equation 3.29 (Lee *et al.*, 2008):

$$x_a^{i+1} = x_a^i + v_a^{i+1}$$
 (Eq. 3.29)

This cycle continues until either the objective function value is met, or the maximum number of iterations is reached. The method is highly efficient due to swarm intelligence, which enables a complex search to take place, with individual particles undertaking only simple search patterns (Xia *et al.*, 2018). This approach allows for faster centring upon the optimal solution, when compared with other calibration methods (Abbaspour, 2014) this was one of the key reasons for selecting this method. A total of 100 model runs were undertaken per catchment for this research, with ten particles undergoing ten iterations; in all cases a satisfactory objective function value had been reached at this point. The objective function used in this work was the Kling-Gupta efficiency (KGE; Gupta *et al.*, 2009), with the parameter values from the PSO iteration with the highest KGE being taken forward for implementation into SWAT. Equation 3.30 details the calculation of KGE:

$$KGE = 1 - \sqrt{(r-1)^2 + (a-1)^2 + (\beta - 1)^2}$$
 (Eq. 3.30)

where r is the linear correlation between observations and simulations;  $\alpha$  refers to the ratio between the standard deviation in the simulated flow and the standard deviation in the observed flow; and  $\beta$  is the ratio between the mean simulated and mean observed flow. Once calibrated parameter values had been finalised, these were checked to ensure that they were still within a range of what would be expected and acceptable for the catchment, so as to ensure that no over-parametrisation had occurred. These values were then updated within the SWAT databases and the newly calibrated model was validated against observed daily streamflow for the period 1999 to 2014, again with a 3-year model warm-up period. Calibration and validation KGE results for each catchment are shown in Table 3.5.

Catchment	Calibration KGE	Validation KGE
Clwyd	0.810	0.788
Conwy	0.770	0.718
Dyfi	0.788	0.728
Teifi	0.851	0.723
Tywi	0.841	0.717

**Table 3.5.** Kling-Gupta efficiency (KGE) calibration and validation values for comparison of gauged and modelleddaily mean streamflow for each catchment.

In terms of water quality, four factors have been studied, suspended sediment, total nitrogen, total phosphorous and dissolved oxygen. These factors were chosen based on the available and reliable outputs achievable in SWAT, observation data availability for output validation, as well as their importance to overall stream water quality, aquatic life, and water abstractors. While the inclusion of other factors, such as coliforms, persistent bacteria, and dissolved organic carbon, was considered, these ultimately fell short of one or more of the selection criteria, so were deemed unsuitable. Due to a lack of sufficient historical data for any of the four factors, it was not possible to calibrate the model for water quality. A validation has been

carried out however, with model output being compared with data taken from the Natural Resources Wales (NRW) water quality data archive held in WISKI (NRW, 2019b). This dataset provides occasional, and often temporally randomly distributed, data points on each of the four factors, for the period 2000-2010. Sampling frequency varies from monthly to as infrequently as once per year, with sampling rate generally increasing towards the end of the time period in all catchments, Table 3.6 summaries the number of samples available for each factor and catchment. The samples were mostly collected by NRW to check compliance with EU directives, so there is a high confidence in the dataset reliability and consistency; for this reason, and following an inspection of data for obvious outliers, no correction or alteration was made to the data. A single sampling location for each catchment was chosen (shown in Figure 3.1), with the aim of maximising the number of data points and, where possible, ensuring coverage for all four factors. Unfortunately, the latter aim was only fully achievable in three of the five catchments, with no observations recorded for total nitrogen at the Conwy or Dyfi sampling locations. Validation, where possible, took the form of the calculation of Person's correlation coefficient values between simulated and observed data points, to give an indication of the agreement in the datasets.

	Number of samples				
Catchment	Suspended sediment	Total phosphorous	Total nitrogen	Dissolved oxygen	
Clwyd	85	63	63	84	
Conwy	118	83	0	106	
Dyfi	105	85	0	104	
Teifi	82	74	75	82	
Tywi	23	19	79	22	

Table 3.6. Total samples used in the validation of studied water quality factors in SWAT.

## 3.5.3 Future model runs

With calibration and validation successfully completed, future daily weather data could be implemented into the catchment models. Section 3.1 details the rationale behind the selection of RCP8.5 as the future global GHG emissions scenario under which the future streamflow and water quality assessment was conducted. Future daily temperature and precipitation data, generated under RCP8.5 conditions, was obtained from the Met Office Hadley Centre's (MOHC) UK Climate Projections 2018 (UKCP18) dataset, specifically the

'Regional Projections on a 12 km grid over the UK for 1980-2080' (MOHC, 2018b). UKCP18 data was selected for use as the latest and most comprehensive projections for the UK. The regional projections in particular were chosen as the dataset with the finest spatial scale, at the time of implementation into SWAT, which takes account of regional-scale landscape features, allowing for a local level analysis (Murphy *et al.*, 2018). This dataset is an update to the previous version (UKCP09), which incorporates improved representation of extreme events as well as greater spatial resolution (MOHC, 2018b), both important to the analysis in Studies 2 and 3. The dataset consists of an ensemble of twelve member models, each of which has been dynamically downscaled from the 60 km gridded HadGEM3-GC3.05 global coupled model perturbed parameter ensemble (Murphy *et al.*, 2018). Each of the ensemble members differs due to natural climate variability as well as uncertainty in the projections (Kendon *et al.*, 2019). Input weather data was inspected for obvious outliers or anomalous entries before use in SWAT, with no other corrections or alterations undertaken on the data.

To enable the implementation of the UKCP18 data into SWAT, it was necessary to convert the gridded data to points, with a point at the centre of each 12 km grid being assigned that grid's daily climate variable values, and termed a pseudo-meteorological station. A total of 28 grid squares for which the central point lay within one of the five catchments were identified; a subset of thirteen of these were taken forward to be used as pseudo-meteorological stations (Figure 3.5). This selection was required because of the substantial time and computational requirement of extracting daily weather data values from the supplied NetCDF format, to allow for implementation in SWAT. Station selection was completed with the aim of ensuring a good coverage both spatially, as well as in terms of elevation, which is particularly important in SWAT for temperature and precipitation variation based on elevation and topography. While narrowing down the number of stations may limit the representation of variability due to elevation, this was mitigated in four of five catchments, those which share a boundary (Clwyd with Conwy, and Teifi with Tywi), as pseudo-meteorological station data from both catchments was made available to each individual model, increasing data coverage while maintaining computer resource efficiency. As the Dyfi has no neighbouring catchments that have been modelled in this study, this catchment has the highest density of stations, to ensure the best coverage possible (Figure 3.5).



**Figure 3.5.** Pseudo-meteorological stations used in the implementation of UKCP18 future weather data into SWAT, elevation in metres shown below station name.

Future modelling at a daily time-step was completed for the hydrological years 2021-2080, with each of the five catchments being modelled for each of the twelve ensemble members. This approach has been taken to account for some of the uncertainty in the projections, allowing for an envelope of potential future streamflow and water quality changes under RCP8.5 conditions to be established, as well as an ensemble mean. Due to the size of the catchments studied, and limits set upon the size of Microsoft Access databases, which are used to store the model output, model runs for the 60-year period had to be split in to multiple periods; as many as four, 15-year periods for the largest catchments (Table 3.7). To

counteract as far as possible any discontinuity in the output at these breaks, a 7-year model warm up period was used before the start of each period. In order to further mitigate disjointedness in model outputs, as few breaks as possible were used in each catchment, therefore smaller catchments are split into fewer, longer periods (Table 3.7), in this way individual dataset continuity was prioritised above overall approach consistency.

Catchment	Total sub- basins	Total HRUs	Number of model periods	Model period length	Model period break points
Clwyd	302	697	3	20 years	2041, 2061
Conwy	202	770	2	30 years	2051
Dyfi	227	662	2	30 years	2051
Teifi	425	1333	4	15 years	2036, 2051 2066
Tywi	496	1697	4	15 years	2036, 2051 2066

 Table 3.7. Catchment SWAT model characteristics and run period information.

#### 3.5.4 Data analysis

Trend analysis has been conducted for seasonal and annual average streamflow and water quality factor output, using the mean of the outputs of all 12 model runs. The method used for analysis is the same as for Study 1, MK trend analysis, as detailed in Section 3.3. The method used to characterise low flows also remains as conducted in Study 1; the analysis of high flows, however, has been modified. The characterisation of extreme high flows for Study 2 takes account of discrete high flow events, to give a better understanding of how such events are changing in magnitude and frequency, as well as duration. In the previous analysis, change in the number of days per year/season when streamflow exceeded the dataset 95<sup>th</sup> percentile was analysed. In this study, it is the number of events per year/season with at least one day of flow above the 60-year 95<sup>th</sup> percentile, which are analysed, where a single event could have several days above the 95<sup>th</sup> percentile. This method helps to mitigate the impact that multi-day high flow events could have on the analysis, which could potentially give the impression that high flow events are becoming more frequent, as opposed to longer in duration and magnitude. By identifying individual events, the analysis of individual event duration and frequency is also therefore possible. Individual events were classified as starting and ending when streamflow was equal to baseflow; this method was selected as it enabled multi-peak events to be easily identified. Baseflow was calculated by the digital filter method as defined by Sawicz et al. (2011) and presented in Equations 3.2 and 3.3 in Section 3.2.

# 3.6 Study 3: Future water availability analysis

# 3.6.1 Study area and analysis framework

The third study uses the Conwy and Tywi only, in the analysis of the impact of modelled streamflow changes from Study 2 on PWS and HEP. These two catchments were chosen due to their contrasting physical characteristics (Section 3.2, Table 3.1), especially in terms of LULC, slope, and catchment area, enabling investigation of the impact of these differences on catchment water resources. In addition, both catchments host several non-impoundment run-of-river HEP schemes, suitable for the analysis undertaken in this study, sixteen in the Conwy and nine in the Tywi (NRW, 2019c). Furthermore, the Tywi also contains the single largest river-based PWS abstraction location found in any of the five catchments. Indeed, it supplies the largest water treatment works in all of DCWW's network, producing water for over 400,000 consumers, therefore a highly important strategic asset (DCWW, 2019a).

Daily streamflow data for the analysis has been taken from Study 2, with the mean of all 12 model runs dataset for each catchment being used. The mean dataset has been used for computational efficiency, especially considering the scenario analysis conducted for PWS (details below) and the number of HEP locations studied. While this method does remove the ability to give a range of potential future outcomes, the results are still able to provide a sufficient picture of the future situation and the challenges likely to be faced.

For both HEP and PWS, MK trend analysis of the studied characteristics (described below) was undertaken for both a medium-term and long-term scenario. The medium-term scenario spanned the years 2021-2054, a useful study period for both industries. For HEP, this is the regular life span of small-scale systems, such as those studied here, that have been installed recently or are to be installed in the near future (Hatata *et al.*, 2019; Killingtveit, 2019). For PWS, the timeframe covers the period of focus of DCWW's most recently published water resource management plan (DCWW, 2019a), as well as their vision document to 2050 (DCWW, 2018). The long-term scenario (2021-2079) gives an indication of the far-future challenges faced by both industries and the potential future adaptation needs.

#### 3.6.2 Hydroelectric power

Calculation of the ability of future streamflows to power run-of-river HEP installations in the Conwy and Tywi was undertaken manually, using new methods developed for this study. Abstraction conditions for the total 25 schemes identified from a dataset of licensed abstractions, provided by NRW (NRW, 2019c), were first set as per the general licensing guidelines for HEP schemes, set out by NRW (NRW, 2020b). These guidelines were used due to a lack details pertaining to the actual abstraction conditions placed on the individual schemes studied. They are, however, a good proxy, as these are commonly applied, aside from in situations where specific environmental, ecological or other concerns require more stringent abstraction conditions (NRW, 2020b).

The guidelines set out several conditions relevant to the analysis of this study, these being the hands-off-flow (HoF; streamflow volume below which abstraction is not permitted), maximum daily abstraction volume (Amax), and percentage take (Qtake; proportion of flow between HoF and Amax permitted for abstraction). Two parallel sets of guidelines exist, depending on the gradient of the stream section between which water is abstracted, and returned (NRW, 2020b). While HoF volume is the same for both (set at Q<sub>95</sub>; streamflow volume exceeded 95% of the time), A<sub>max</sub> and Q<sub>take</sub> differ. Schemes with a depleted reach where the slope is less than 10% are classed as 'Zone 2' (Z2), while those with a depleted reach slope greater than 10% are classed as 'Zone 3' (Z3). Z2 schemes have an Amax volume of 1.3 times mean annual streamflow, whereas for Z3 schemes, this is less, being set at the mean annual streamflow. Conversely, Q<sub>take</sub> is lower in Z2 schemes than Z3, being 50% and 70% respectively (NRW, 2020b). Due to the fact that individual characteristics of the schemes studied were not known, the depleted reach slope has been estimated for each scheme, based on the slope of the subbasin immediately downstream of the abstraction location. While not a perfect approximation, this method does give a good indication of the situation of each scheme, allowing for the best decision to be made upon zone classification.

With the scheme classification and associated abstraction conditions of each scheme in mind, the calculation of daily abstraction ( $A_{daily}$ ) for each scheme was executed by Equations 3.31 and 3.32, develop for this study. These calculations operate under the assumption that each HEP scheme abstracts the maximum amount of flow allowable:

$$Q_{surplus} = Q - HoF$$
 (Eq. 3.31)

where  $Q_{surplus}$  is the amount of water available for abstraction, Q is daily streamflow, and *HoF* represents the hands-off-flow; the result is used in the calculation of  $A_{daily}$ :

$$A_{daily} = Q_{surplus} \times Q_{take} \begin{cases} 0, & \text{if } A_{daily} < A_{start} \\ A_{max}, & \text{if } A_{daily} > A_{max} \\ A_{daily}, & \text{if } A_{start} < A_{daily} > A_{max} \end{cases}$$
(Eq. 3.32)

where  $Q_{take}$  is the proportion of flow available for abstraction as per the zone conditions,  $A_{start}$  refers to the minimum abstraction volume required to start, and for efficient operation of, the turbine, and  $A_{max}$  represents the maximum permitted abstraction volume. It has been assumed that an impulse turbine is in use at each site, as is common with small-scale HEP schemes in upper catchment reaches, such as those analysed in this study (Lilienthal *et al.*, 2004; Cobb & Sharp, 2013; Židonis *et al.*, 2015). Impulse turbines have largely high and stable efficiency after approximately 10% of designed flow is achieved (Paish, 2002; Novara & McNabola, 2018; Chitrakar *et al.*, 2020), making them ideal for settings with variable daily abstraction (Cobb & Sharp, 2013). For this reason,  $A_{start}$  was set at 10% of  $A_{max}$  for each scheme, which has been assumed as the designed flow volume.

Four factors have been studied in the analysis of the impact of future streamflow changes on HEP operations, these being, the number of days A<sub>start</sub> is achieved, the number of days A<sub>max</sub> is reached, mean daily abstraction on days A<sub>start</sub> is achieved, and total abstraction. These factors have been calculated annually and seasonally for each scheme. The mean of all combined Z2 and Z3 schemes in each catchment has then been calculated separately, with MK analysis conducted on these mean datasets, allowing for a comparison of the response both between catchments, as well as between scheme classifications. The four factors chosen for analysis were selected in order to provide as complete a picture as possible in terms of the potential change in future HEP scheme operation and power generation.

#### 3.6.3 Public water supply

Assessment of the ability of future streamflows to satisfy PWS demand at the previously referenced abstraction location in the Tywi has been undertaken using the Water Evaluation And Planning (WEAP) system (Yates *et al.*, 2005). This programme was chosen due to its ability

to easily implement different scenarios and compare the results; this was important as the assessment of three future water demand scenarios was being undertaken. The three scenarios studied are based on an increasing, static, and decreasing future PWS demand, with demand data being derived from the 5-year PWS abstraction dataset provided by DCWW and used in Study 1. This data has been taken as a proxy for total water demand and leakage.

Future daily water demand under the increasing demand scenario was based on the linear relationship established in Study 1 between daily temperature and daily total abstraction from the abstraction location studied on the Teifi. This relationship has then been extrapolated and applied to the mean daily temperature of the three pseudo-meteorological stations positioned within the Tywi for the SWAT modelling. The daily temperature relationship was chosen as the predictor of future demand because of the difficulty that would be encountered if using the daily precipitation relationship, due to the presence of days without rainfall, which is likely to skew the data. This scenario represents a worst-case in terms of future demand, with no effort made to reduce demand or combat leakage. The static scenario represents a control dataset and presumes abstraction rates stay constant at present levels. Present levels were established by calculating mean demand on each day of the year from the 5-year dataset provided by DCWW. This year of mean daily demand was then repeated yearly for the full period of study. Finally, the decreasing demand scenario is based on a projection by DCWW for a 20% decline in total water demand from the Tywi catchment by the end of the study period, due to a decline in domestic demand and reduction of network leakage (DCWW, 2019a). The decreased demand scenario starts with the same base year as is used and repeated in the static scenario, with the 20% decline being applied linearly across the full 60-year study period and represents an optimistic scenario of substantial demand reduction and therefore abstraction requirement. The three scenarios chosen represent a wide range of future demand levels, enabling a broad assessment of future climate change impacts on abstraction capability.

The daily ability of future streamflow to satisfy the volume of demand set out in the three scenarios, has been assessed based on the abstraction license conditions under which the abstraction operates. The main condition in the license is a compensatory HoF, set at 681.91 million litres per day (SWWRA, 1965); this volume was incorporated into WEAP and total daily unmet demand ( $D_{unmet}$ ) was then calculated for each scenario, using Equation 3.33:

$$D_{unmet} = Q - HoF - D \tag{Eq. 3.33}$$

where *D* is daily total water demand; and *Q* and *HoF* are defined as Equation 3.31. At times of low flow, the abstraction licence allows for releases to be made from Llyn Brianne reservoir, in the upper reaches of the catchment, which can subsequently be abstracted downstream, up to a given limit. Due to this mechanism,  $D_{unmet}$  is therefore also assumed to be equal to the total daily required reservoir release. MK trend analysis for PWS has been undertaken seasonally and annually on three characterises of  $D_{unmet}$  for each future demand scenario and for the medium- and long-term. Total volume of unmet demand per season/year, number of days per season/year when demand is unmet, and mean annual/seasonal unmet demand have all been calculated. These three characteristics have been chosen as they provide a good overview of the changes and challenges faced in the future supply of water, in the catchment. They indicate, amongst others, the potential future volumes of water required to be released from Llyn Brianne, the frequency with which release may be required, and the mean volume of release per day when compensation of flows is needed. All of these have important implications for water resource and reservoir management.

# CHAPTER 4: STUDY I

# Influence of historical climate patterns on streamflow and water demand in Wales, UK

R.J.H. Dallison, S.D. Patil and A.P. Williams School of Natural Sciences, Bangor University

This chapter has been published in full, in *Water*, 2020, **12**(6): 1684. DOI: 10.3390/w12061684.

# Author contributions

Richard Dallison: conceptualisation, methodology, formal analysis, investigation, data curation, writing – original draft, writing – review & editing, visualisation
Sopan Patil: conceptualisation, writing – review & editing, supervision
Prysor Williams: writing – review & editing, supervision

# 4.1 Abstract

Ensuring reliable drinking water supplies is anticipated to be a key future challenge facing water service providers due to fluctuations in rainfall patterns and water demand caused by climate change. This study investigates historical trends and relationships between precipitation, air temperature and streamflow in five catchments in Wales, before correlating these with actual total abstraction data provided by the water company, to give insight into the supply-demand balance. Changes in seasonal and annual averages, as well as extreme events, are assessed for a 34-year period (1982-2015) and a breakpoint analysis is performed to better understand how climate has already changed and what this might mean for the future of water supply. Results show a north-south divide in changes in extreme temperature and streamflow; a strong warming trend in autumn average temperatures across Wales (Sen's slope range: 0.38-0.41, *p* < 0.05), but little change in precipitation. Abstraction, as a proxy for overall water demand, is shown to be positively correlated to temperature (Spearman's  $\rho$ value range: 0.094-0.403, *p* < 0.01; Pearson's *r* value range 0.073-0.369, *p* < 0.01) in four of five catchments. Our study provides new insight into the relationship between abstraction volume and hydroclimatic factors and highlights the need for catchment-scale water resource planning that accounts for hydroclimatic variations over small spatial distances, as these nuances can be vital.

**Keywords:** Abstraction; Breakpoint analysis; Climate change; Hydroclimatic change; Trend analysis; Water demand

# 4.2 Introduction

Water service providers (WSPs) face a vast array of challenges and uncertainties when planning their future operations and services. Brown *et al.* (2010) compiled a list of 94 priority research questions for the UK water sector, in which the impact of climate change on water quantity was ranked as the second most important question. Their work also highlighted the need to better understand the drivers of water demand, both domestic and commercial, in order to improve future demand forecasting. Previous work on the characterisation of hydroclimatic trends in the UK suggests that precipitation and streamflows have become more seasonal, a pattern that is expected to continue (Whitehead et al., 2009b; Suggitt et al., 2015; Garner et al., 2017). A study by Christiersen et al. (2012) showed, using UK Climate Predictions 2009 (UKCP09) data, that by the late 2020s, increases in winter precipitation levels are likely to be more prominent in northern and western parts of the UK, while decreases in summer flows will be seen more generally across the whole country. Similarly, a hydrological modelling study by Prudhomme et al. (2012) showed that summer precipitation and streamflows will decrease across the UK by varying amounts for the period 2040-2069; whereas future winter precipitation and streamflows showed an upward trend, especially for Wales. Extreme precipitation events are also projected to become more seasonal in the UK, with longer duration and more intense rainfall events in winter becoming more common (Suggitt et al., 2015). Mayes (2000) suggested that these anticipated changes will not be uniform across the UK, and current rainfall gradients are likely to be accentuated, i.e., the south getting drier in summer and the north getting wetter in winter.

Regional-scale understanding of water resource provision in the UK is particularly important because water supply is under the control of individual water companies that serve separate regions of varying sizes, populations, and physical characteristics. For instance, in south-east England, which is already a water stressed area, studies on future hydroclimatic trends suggest that summer streamflow levels will continue longer into autumn, with overall summer flow levels declining also. Furthermore, winter streamflows will increase and continue longer into spring, leading to accentuated seasonality in terms of season longevity and flow volumes (Diaz-Nieto & Wilby, 2005; Wilby & Harris, 2006; Wilby *et al.*, 2006; Cloke *et al.*, 2010; Arnell, 2011). However, an increase in winter precipitation will do little to combat summer shortages if no further storage capacity is developed soon (Whitehead *et al.*, 2013).

Borgomeo *et al.* (2014) suggested that, due to the combined effect of climate change and significant predicted population growth, the London water supply zone urgently required both supply and demand-side interventions if the current standard of water provision is to continue. In Scotland, although winter precipitation is predicted to increase in the future, a lower percentage of it will fall as snow (Capell *et al.*, 2013, 2014). This will make catchments more responsive to winter precipitation and increase the pressure on water managers to deal with larger discharge events (Baggaley *et al.*, 2009). For summer precipitation and streamflows, Blenkinsop & Fowler (2007) noted that Scotland has a limited amount of groundwater storage capacity, which heightens the drought risk from any reduction in non-winter precipitation. In Wales, studies suggest that winter and summer season characteristics, e.g. wet winters and dry summers, will be exacerbated, especially in winter (Dixon *et al.*, 2006; Fowler & Wilby, 2010; Thompson *et al.*, 2017).

In this study, we use Wales as a case study region, a country often viewed as abundant in water resources, receiving some of the highest average annual rainfall totals in the UK (Met Office, 2020), but which in reality has zones of water deficit (DCWW, 2016a). Wales is also important due to its role as an exporter of water to major metropolitan areas in England. Dŵr Cymru Welsh Water (DCWW), the major WSP for Wales, has over 20 bulk water trades, the largest of which supplies 360 million litres per year to Severn Trent Water for distribution around Birmingham (Warren & Holman, 2012; DCWW, 2016a).

Past studies on water resources in Wales have predominantly been conducted either as part of UK-wide research (Prudhomme *et al.*, 2003; Whitehead *et al.*, 2009b; Fowler & Wilby, 2010; Hannaford, 2015; Suggitt *et al.*, 2015; Watts *et al.*, 2015a; Burt *et al.*, 2016), or with a focus on the combined England and Wales region (Arnell, 2011; Henriques *et al.*, 2015). When focusing specifically on the area covering Wales, Fowler & Wilby (2010) projected a much larger magnitude of increase in winter flows from the 1960-1990 average to 30-year averages centred on 2025, 2055 and 2085, than the corresponding decreases in summer precipitation. Dixon *et al.* (2006) showed a significant upward trend across 56 Welsh and West Midlands catchments between 1962-2001 for winter high flow values, but no significant changes in the mean annual values. Conflictingly, Macdonald *et al.* (2010), demonstrated that during the period 1973-2002, there was no significant change in the seasonality of rainfall across 30 catchments in Wales, which they proved to have a significant link to streamflows; however,

they did show that the frequency of occurrence of extreme precipitation events in Wales had increased during the study period. These contrasting results highlight the need for careful consideration when selecting study period length and timeframe due to the potential impact on trend analysis results and projections.

Up until recently, supply-side measures to tackle water scarcity and to manage water resources have traditionally been the main path towards a reliable sustainable future water network. However, it has increasingly been recognised over the past decade that demandside interventions should also play a role as an adaptation measure (NIC, 2018). For this to be a viable option, further work is needed to understand the relationship between prevailing and antecedent weather conditions, and demand for water in the UK. Several studies have looked at the general interplay between the two in the UK (Herrington, 1996; Downing *et al.*, 2003; Butler & Memon, 2006; Browne et al., 2013), and abroad (Gutzler & Nims, 2005; Balling & Gober, 2007; Chang et al., 2014). This interplay has been investigated since at least as far back as the 1990s, with Herrington (1996) stating at the time, that up to 40% of total consumption in summer can be due to garden watering, which is obviously highly affected by the prevailing weather conditions. Goodchild (2003) used summer daily domestic water demand data (55% of UK piped water supply at the time) from 41 domestic properties and daily meteorological data to develop a demand prediction model. The model included ten weather variables to account for current and antecedent conditions; evapotranspiration, days since rain, and temperature were all important functions. This modelling work projected a 2.1% increase in average summer 7-day household demand by the 2020s. More recently, Parker & Wilby (2013) reviewed domestic water demand in the UK and noted the lack of studies on weather and climate. It is also important to look not only at domestic demand, but also industrial, agricultural and non-revenue water use (e.g. leakage), as they all influence the long-term sustainability of water supply.

In this study, we look at the implications of past trends in hydroclimatic data on two of the problems identified by Brown *et al.* (2010): (1) impact of climate variations on water availability and (2) understanding the factors affecting water demand. The first problem has been addressed by assessing trends in seasonal and annual average climate and streamflow data as well extreme event frequency and magnitude, using Mann-Kendall trend analysis and breakpoint analysis. We have addressed the second problem by investigating the historical

links between hydroclimatic factors and total water demand, using actual abstraction data provided by DCWW as a proxy for demand. To our knowledge, this is one of the first studies conducted independently of a WSP to use actual abstraction data provided by a WSP in this manner in the United Kingdom. The reliance on Wales for water supply in other regions, combined with a potentially inaccurate assumption of national water abundance in Wales, makes the region a crucial area of study in terms of water management and water supply availability. Our research has therefore been undertaken in order to provide information for future water resource planning and policy decisions, as well as future research. It is hoped that this will be achieved by providing evidence of the long-term trends and links between prevailing weather and flow conditions, as well as total demand for water in Wales.

# 4.3 Data and methods

# 4.3.1 Study catchments

Our study focusses on the rivers Clwyd and Conwy in the north, the Dyfi in the west, and the Teifi and Tywi in the south of Wales (Figure 4.1). We selected these catchments as a result of them being among the largest systems within the region, encompassing a range of land use/land cover (LULC), and exhibiting a variety of different catchment characteristics. Furthermore, the catchments are also mostly encompassed within DCWW's water supply zone (Figure 4.1), and all have multiple surface water abstraction locations licensed to DCWW for use for public water supply. Brief catchments descriptions have been provided below and key comparative details about each are given in Table 4.1<sup>1</sup>. Figures for LULC in the catchments were calculated from 2012 CORINE Land Cover data (Copernicus Land Monitoring Data), while the catchment size, longest stream length, elevation and slope values were calculated using a 5 m digital elevation model provided by Ordnance Survey.

<sup>&</sup>lt;sup>1</sup> Stated catchment details/characteristics, such as land cover proportions and catchment area, differ for this study from those shown in Chapters 3, 5 and 6 due to variations in delineated catchment area for this study. Catchment areas defined in this study have been delineated manually in ArcMap using the Hydrology toolset of the Spatial Analyst toolbox; other chapters use the area delineated during catchment setup for SWAT modelling.



**Figure 4.1.** (**A**) Catchments; streams (formation threshold of 1 km<sup>2</sup>); stream order (derived by Strahler method); and gauging station locations. (**B**) Catchment elevation. (**C**) Catchment land use/land cover derived from CORINE Land Cover data.

River	Catchment	Main channel	Mean catchment	Catchment land use/land cover (%)			6)	
	area (km²)	length slope (km) (degrees)	Urban	Agricultural	Forest	Scrub*	Wetland	
Clwyd	803	35	7.1	3.7	80.3	7.2	8.8	0.0
Conwy	564	43	10.7	1.6	32.3	13.2	40.7	7.9
Dyfi	676	42	14.2	0.4	30.6	20.1	43.2	5.1
Teifi	1011	80	6.6	0.9	83.1	5.5	9.5	1.1
Tywi	1363	109	9.2	0.7	64.6	16.0	17.1	1.3

**Table 4.1.** Key study catchments details. Catchment area, main channel length, and slope data derived from Ordnance Survey provided 5 m digital elevation model; land use/land cover data derived from 2012 CORINE Land Cover data.

\* Scrub-designated land includes land cover such as natural grasslands, transitional woodland-shrub, moors, and heathlands.

In terms of LULC, the Clwyd, Teifi, and Tywi are dominated by agricultural land use (Figure 4.1); while the largest proportion of LULC in Conwy and Dyfi is scrubland (mainly in the form of moors and heathland in Conwy, and natural grasslands in Dyfi). Agricultural land in all study catchments is mostly pastureland, however, Clwyd does have a larger proportion of its total LULC as arable land (10.4%). Forested land in all catchments is mainly coniferous, except in Conwy where there is an equal proportion of coniferous and mixed forest (5.1% LULC each) and 3% broadleaf forest. The bedrock geology is predominantly mudstone, siltstone, and sandstone in all five catchments, ranging from about 60% in Conwy and Clwyd, 73% in Tywi, and over 85% in Dyfi and Teifi. The catchments range in area covered from 1363 km<sup>2</sup> at the Tywi to less than half of that for the Conwy (564 km<sup>2</sup>). The steepest catchment is the Dyfi, with an average slope of 14.2 degrees, being largely in the south of the Snowdonia region, while the Teifi in the south is the least steep catchment, with an average slope of 6.6 degrees.

## 4.3.2 Data

Hydrological and meteorological data for all five study catchments was obtained for the longest complete period possible for all datasets, that being the 34-year period from 1<sup>st</sup> October 1981 to 30<sup>th</sup> September 2015 (except flow data for the Dyfi, which was only available to 5<sup>th</sup> May 2014). Daily precipitation and air temperature data were obtained from the Centre for Ecology and Hydrology's (CEH) Climate, Hydrology and Ecology research Support System (CHESS) dataset. The mean daily value across all the 1 km grids contained within each catchment was calculated for both precipitation and temperature, giving a daily average value

for each variable across each catchment. Daily streamflow data for the catchments was obtained from CEH's National River Flow Archive dataset; gauging station locations are shown in Figure 4.1.

Daily water abstraction volumes were obtained from DCWW, which is the main water supply company for this region. However, these data were only available for a 5-year period from 1<sup>st</sup> January 2012 to 30<sup>th</sup> September 2016. Therefore, comparison of water abstraction data with hydroclimatic data could only be completed for overlapping dates between the datasets, 1<sup>st</sup> January 2012 to 30<sup>th</sup> September 2015. The daily water abstraction data was provided for 22 abstraction locations within the five study catchments and has been used as a proxy for overall demand across the network for all consumers, as well as leakage.

# 4.3.3 Trend analysis

We selected the nonparametric Mann-Kendall (MK) test (Mann, 1945; Kendall, 1975) to detect any consistent trends in the hydroclimatic data over time. The decision to use a nonparametric test was taken due to the nature of both climate and hydrology data generally being non-normally distributed and displaying seasonality, which goes against the assumption of constant distribution (Kundzewics & Robson, 2004). Moreover, hydrology data often displays auto-correlation, making it unsuitable for parametric testing (Hamed & Rao, 1998). The MK test maintains the sequential order of the data and calculates Kendall's tau (a measure of association between two samples) between each value and all values proceeding it, to test for a monotonic increase or decrease relationship in the data over time (Helsel & Hirsch, 2002). MK has also been successfully applied to similar hydroclimatic data in various locations globally (Murphy *et al.*, 2013; Basarin *et al.*, 2016; Mwangi *et al.*, 2016; Hajani *et al.*, 2017) so was deemed suitable for use in this study. Details on the specific equations relating to the MK test can be found in Jaiswal *et al.* (2015).

Long-term average trends in the datasets were examined by analysing seasonal and annual averages for each catchment; where winter is December to February; spring is March to May; summer is June to August; autumn is September to November; and where, for example, the 1982 hydrological year runs from 1<sup>st</sup> October 1981 to 30<sup>th</sup> September 1982. For the abstraction data, we have taken the sum of abstractions in each catchment, with trend analysis being performed on this value, in order that the results are comparable to the

hydroclimatic factors analysed. We also examined trends in the frequency and magnitude of extreme events relating to hydroclimatic factors. This was achieved firstly by analysis of trends in maximum and minimum temperatures; maximum 1-day precipitation total and cumulative rainfall totals; and maximum and minimum average 1-day streamflow volumes. Second, we undertook an analysis of "events over threshold", in order to establish whether there are generally more or less extreme weather events per year/season over the study period. This was completed by taking the 5<sup>th</sup> and 95<sup>th</sup> percentile values of a whole dataset (seasonally and annually) and analysing the number of times in each year and season that that value is surpassed (95<sup>th</sup> percentile) or not reached (5<sup>th</sup> percentile) for temperature and streamflow, and surpassing the 95<sup>th</sup> percentile only for precipitation. The MK test was applied with Sen's slope estimator (Sen, 1968) in order to estimate the size and direction of trends; Hamed and Rao's method of auto-correlation correction was also applied in order to remove any apparent trend which the data exhibits with itself over time (Hamed & Rao, 1998).

In order to detect any sudden changes in the hydroclimatic data, we undertook a breakpoint analysis on the seasonal/annual average and extreme event data. This analysis was completed using two well-established methods, the Standard Normal Homogeneity Test (SNHT) method (Alexandersson, 1986; Alexandersson & Moberg, 1997) and the Pettitt method (Pettitt, 1979). These two methods were used to ensure the widest possible detection of breaks in the data; with the SNHT being more accurate at detecting breaks at the start and end of time series, and the Pettitt method being more reliable in the middle (Hawkins, 1977; Hänsel *et al.*, 2016). If annual values of the variable being tested are identically distributed and independent, both methods accept the null hypothesis. The alternative hypothesis (that there is a change point in the series) will be accepted however, if a shift in the value of the mean has occurred (Hänsel *et al.*, 2016). Both methods are also location specific, so will identify the year at which the mean changed within the time series. For further information on the equations that drive the two methods, we suggest Hänsel *et al.* (2016) and Jaiswal *et al.* (2015).

#### 4.3.4 Correlation analysis of hydroclimatic and abstraction data

In order to investigate the relationship between the three hydroclimatic factors and water abstraction we calculated both Pearson's & Spearman's rank correlation coefficients. The analysis was performed for each catchment with daily and monthly average temperature,

precipitation and streamflow all being compared separately to daily and monthly total abstraction volume. We calculated Pearson's coefficient values (*r*) using Equation 4.1:

$$r = \frac{\sum_{i=1}^{n} (hc_i - \overline{hc})(abs_i - \overline{abs})}{\sqrt{\left(\sum_{i=1}^{n} (hc_i - \overline{hc})^2\right) \left(\sum_{i=1}^{n} (abs_i - \overline{abs})^2\right)}}$$
(Eq. 4.1)

where  $hc_i$  refers to the daily average hydroclimatic variables studied,  $abs_i$  refers to daily total abstraction volume, and hc & abs represent the mean of the entire respective datasets. The nonparametric Spearman's correlation coefficient ( $\rho$ ) was calculated using Equation 4.2, a modified version of Pearson's correlation coefficient which calculates correlation between ranks, as opposed to raw data:

$$\rho = \frac{\sum_{i=1}^{n} \left( R(hc_i) - \overline{R(hc)} \right) \left( R(abs_i) - \overline{R(abs)} \right)}{\sqrt{\left( \sum_{i=1}^{n} \left( R(hc_i) - \overline{R(hc)} \right)^2 \right) \left( \sum_{i=1}^{n} \left( R(abs_i) - \overline{R(abs)} \right)^2 \right)}}$$
(Eq. 4.2)

where  $R(hc_i)$  refers to the rank of the daily average hydroclimatic variables studied,  $R(abs_i)$  refers to the rank of the daily total abstraction volume, and  $\overline{R(hc)} \otimes \overline{R(abs)}$  represent the mean rank of the entire respective datasets.

## 4.4 Results

Our results indicate a north-south divide when looking at extreme temperature and streamflow changes. Additionally, a strong warming trend in average autumn temperatures across Wales is observed, however little change in precipitation is seen. Abstraction, as a proxy for overall water demand, is strongly positively correlated to temperature in all catchments and negatively with streamflow and precipitation in all catchments except the Dyfi. Below we provide a detailed breakdown and explanation of these results.

#### 4.4.1 Trend and breakpoint analysis

#### 4.4.1.1 Precipitation

We found no significant trends during MK analysis or breakpoint analysis of annual and seasonal precipitation averages in any of the five catchments for the study period; but Sen's slope indicator values do suggest a slight decrease in average spring precipitation across all

catchments. In addition, only two significant trends were present in terms of extreme precipitation events; the first is a decrease in total cumulative spring precipitation in the Clwyd catchment throughout the period (Tau = -0.184, p = 0.040; Sen's Slope = -1.196). The second is a decrease in the number of extreme wet autumn days in the Teifi catchment (Tau = -0.593, p = 0.003; Sen's Slope = -0.138). Although both of these trends are mirrored in the other catchments, those trends are not statistically significant, highlighting the need to study individual catchments to account for varying characteristics. Additionally, the Pettitt breakpoint analysis showed a marked increase in winter cumulative precipitation in the Conwy after 1989 (p = 0.040), the mean for the post 1989 period increasing by 13.4% compared to the pre-1989 period.

#### 4.4.1.2 Temperature

Unlike precipitation, air temperature data does display significant trends in Wales over the course of the study period. In all catchments, we observe a warming trend in average autumn temperature (Figure 4.2); the rate of increase is marginally higher in the two north Wales catchments compared to the two in the south of the country (Table 4.2).



**Figure 4.2.** Warming trend in average autumn temperatures displayed in all catchments. Linear trend for the Dyfi catchment shown with red dashed line to exemplify the linear trend in all catchments.

**Table 4.2.** Mann-Kendall trend analysis and Sen's slope indicator results for average autumn air temperatures trends (1981-2015) in each of the five study catchments. Values for Kendall's Tau underlined are significant at p <0.05, and those in bold are significant at p <0.01.

River	Kendall's tau	Sen's slope
Clwyd	0.337	0.041
Conwy	<u>0.348</u>	0.040
Dyfi	0.344	0.038
Teifi	0.337	0.038
Tywi	0.344	0.038

These findings correlate with both the Pettitt and SNHT breakpoint analysis in all catchments, which shows a step increase in autumn temperatures in 1994, with the percentage change in pre- and post-1994 mean temperatures being larger in the north (Table 4.3). We found no significant trends in the MK analysis of annual, winter, spring, or summer datasets, however when looking at the breakpoint analysis, further changes are seen. In the two most northerly catchments, winter temperatures show a break and increase under SNHT analysis in 1987 (p = 0.042 & 0.048 for the Clwyd and Conwy respectively). All catchments also show a breakpoint increase in average spring temperatures in 1987 and annual average temperatures in 1988, both under SNHT analysis.

**Table 4.3.** Pettitt and Standard Normal Homogeneity Test (SNHT) method breakpoint analysis statistics for shown stepped increase in annual average autumn temperatures since 1994 in all study catchments.

River	Pettitt <i>p</i> value	SNHT <i>p</i> value	% change in 1994-2015 mean from 1982-1993 mean
Clwyd	0.001	0.009	+12.2%
Conwy	0.002	0.008	+12.7%
Dyfi	0.003	0.011	+11.8%
Teifi	0.002	0.010	+10.3%
Tywi	0.003	0.010	+10.6%

When looking at extreme temperature events, the three most southerly catchments, Dyfi, Teifi and Tywi, display both a decrease in the number of hottest days annually (days above the whole dataset 95<sup>th</sup> percentile; Figure 4.3), as well as a decrease in the number of coldest summer days (days below the summer dataset 5<sup>th</sup> percentile; Figure 4.4); suggesting a narrowing of temperature ranges, especially in summer. The Conwy catchment in the north also displays the latter trend of fewer of the coldest summer days (Table 4.4). This narrowing of temperature ranges has however, not been abrupt enough to cause a breakpoint in the data, with no significant changes seen.



**Figure 4.3.** Number of occurrences annually where daily average temperature is greater than the 95<sup>th</sup> percentile temperature value for the full 34-year dataset. Linear trend for the Teifi catchment shown with red dashed line to exemplify the linear trend in all three catchments.



**Figure 4.4.** Number of occurrences during summer where daily average temperature is less than the 5<sup>th</sup> percentile temperature value for the full 34-year summer dataset. Linear trend for the Teifi catchment shown with red dashed line to exemplify the linear trend in all three catchments.

**Table 4.4.** Statistically significant Mann-Kendall trend analysis and Sen's slope indicator results for extreme temperature events (1982-2015) for all catchments. No significant trends were found in the Clwyd catchment. Values for Kendall's Tau underlined are significant at p < 0.05.

River	Factor	Kendall's tau	Sen's slope
Conwy	Number of summer days below temperature dataset 5 <sup>th</sup> percentile	<u>-0.271</u>	-0.125
Dufi	Number of days annually above temperature dataset 95 <sup>th</sup> percentile	-0.165	-0.250
Dyn	Number of summer days below temperature dataset 5 <sup>th</sup> percentile	<u>-0.260</u>	-0.133
Toifi	Number of days annually above temperature dataset 95 <sup>th</sup> percentile	<u>-0.173</u>	-0.250
rem	Number of summer days below temperature dataset 5 <sup>th</sup> percentile	<u>-0.269</u>	-0.143
Travi	Number of days annually above temperature dataset 95 <sup>th</sup> percentile	<u>-0.191</u>	-0.286
TYWI	Number of summer days below temperature dataset 5 <sup>th</sup> percentile	-0.266	-0.136

#### 4.4.1.3 Streamflow

We observe only one statistically significant trend in seasonal and annual average streamflow data, this being in annual flows in the Teifi; here a significant increase in average annual flow (Figure 4.5) is detected over the study period (Tau = 0.240; Sen's slope = 0.204; p = 0.044). The Teifi also shows the only observed breakpoint in the average flow data, that being a step increase in winter flow in 2013 (p = 0.019) the mean for the post 2013 period increasing by 63.0% compared to the pre-2013 period.



**Figure 4.5.** Annual average streamflows for study catchments, with linear trend lines also shown (dashed lines). The only statistically significant trend, an increase in streamflow in the Teifi catchment, is highlighted in red.

We found various trends in terms of extreme events in streamflow in Wales (Table 4.5); once again the three most southerly catchments show similar trends, in this instance, an increase in volume of the annual and summer 1-day minimum flow volume. The Conwy shows an increase in winter minimum flow volume, along with an increase in the maximum summer flow, while the Clwyd displays an increase in annual maximum flow volume.

River	Factor	Kendall's tau	Sen's slope
Clwyd	Annual 1-day maximum	<u>0.239</u>	0.406
Conver	Summer 1-day maximum	<u>0.252</u>	1.882
COllwy	Winter 1-day minimum	<u>0.193</u>	0.030
	Summer 1-day maximum	<u>0.222</u>	1.341
	Annual 1-day maximum	0.184	1.238
Dyfi	Spring 1-day minimum	0.339	0.065
	Summer 1-day minimum	0.317	0.048
	Annual 1-day minimum	0.355	0.045
	Winter 1-day maximum	0.269	2.037
Toifi	Spring 1-day minimum	0.237	0.071
Tem	Summer 1-day minimum	0.237	0.059
	Annual 1-day minimum	<u>0.254</u>	0.056
	Autumn 1-day minimum	<u>0.239</u>	0.019
Tywi	Summer 1-day minimum	0.348	0.012
	Annual 1-day minimum	0.320	0.011

**Table 4.5.** Mann-Kendall trend analysis and Sen's slope indicator results for statistically significant trends in extreme streamflow events (1982-2015). Values for Kendall's Tau underlined are significant at p < 0.05, and those in bold are significant at p < 0.01.

# 4.4.2 Correlation analysis of hydroclimatic and abstraction data

When looking at the daily actual abstraction data (1<sup>st</sup> January 2012 to 30<sup>th</sup> September 2015) under both correlation coefficient tests, all catchments display a positive relationship between air temperature and volume of water abstracted (Table 4.6), except for the Dyfi, which shows the reverse trend. All catchments, except the Dyfi, also show a statistically significant negative correlation between precipitation and volume of water abstracted, and consequently a negative relationship between streamflow and abstraction volume (Table 4.6). In order to investigate the situation for Wales as a whole, we have also included in Table 4.6 the correlation results for total daily and monthly abstraction volume in all five catchments with average daily/monthly temperature, total daily/monthly rainfall volume, and total daily/monthly streamflow volume, for all five catchments combined. The correlations seen in this dataset are consistent with those seen in most individual catchments. In addition, all catchments also display a positive relationship between streamflow and air temperature.

**Table 4.6.** Pearson's and Spearman's rank correlation coefficient results for daily and monthly abstraction data with hydroclimatic factors (2012-2015). Statistics relating to a combined analysis of total daily/monthly abstraction in all catchments with average daily/monthly temperature, total streamflow volume and total precipitation volume across all catchments are also provided. Values for *r* and *p* underlined are significant at *p* <0.05, and those in bold are significant at *p* <0.01.

Divor	Hydroclimatic		Daily		onthly
River	variable	Pearson's r	Spearman's <i>p</i>	Pearson's r	Spearman's <i>p</i>
	Temperature	+0.369	+0.403	+0.619	+0.665
Clwyd	Precipitation	-0.181	-0.225	-0.509	-0.485
	Streamflow	-0.615	-0.553	-0.868	-0.808
	Temperature	+0.197	+0.197	+0.279	+0.248
Conwy	Precipitation	-0.111	-0.182	-0.415	<u>-0.381</u>
	Streamflow	-0.169	-0.298	-0.389	-0.429
	Temperature	-0.398	-0.404	<u>-0.372</u>	-0.250
Dyfi	Precipitation	-0.054	-0.140	-0.171	-0.250
	Streamflow	+0.012	+0.018	+0.055	-0.014
	Temperature	+0.073	+0.094	0.094	+0.125
Teifi	Precipitation	-0.073	-0.082	-0.147	-0.256
	Streamflow	-0.117	-0.235	-0.148	<u>-0.374</u>
	Temperature	+0.177	+0.218	+0.485	+0.482
Tywi	Precipitation	-0.165	-0.202	-0.768	-0.724
	Streamflow	-0.509	-0.559	-0.942	-0.961
Combined	Temperature	+0.207	+0.252	+0.504	+0.489
	Precipitation	-0.190	-0.223	-0.755	-0.729
	Streamflow	-0.547	-0.581	-0.925	-0.943

Figure 4.6 shows the aforementioned combined data at a daily time-step from all catchments which has been normalised to each factor's maximum and minimum dataset value. Clear relationships between the hydroclimatic factors and actual abstraction data are observed. When looking at temperature, a positive correlation can be seen in the scatter plot, while the time series plot clearly also shows a large amount of consistency between the average temperature and total water abstraction volume. Arguably the clearest trend seen in the scatter plots is total streamflow volume and total actual abstraction volume, showing a negative correlation. This can also be seen in the time series plot that clearly shows drops in abstraction volume at times of largest flow volumes. Generally, the weakest correlations shown in Table 4.6 relate to precipitation and actual abstraction, the same is true for the graphs presented in Figure 4.6. Although peaks in precipitation dataset, which makes these relationships more difficult to pinpoint.



**Figure 4.6.** Normalised daily hydroclimatic data compared with normalised daily abstraction data for the period 1<sup>st</sup> January 2012 to 30<sup>th</sup> September 2015, for all study catchments combined. **(A1, B1, C1)** Paired comparison, with black dashed lines representing trend direction; **(A2, B2, C2)** time series comparison. Combined data is based on an average across all catchments for temperature, while precipitation, streamflow and abstraction volumes are all total volumes summed from all catchments.

The correlations and relationships seen in the daily data are made more evident when looking at the combined normalised monthly data, as shown in Figure 4.7. Clear negative trends can be seen between total abstraction volume with both total precipitation and total streamflow volume, when looking at scatter and time series line graphs. Clear peaks in abstraction can be seen at times of lowest streamflow and precipitation levels, and vice versa. Temperature and total abstraction are shown to also be broadly in-line when looking at the time series plot, with a clear positive trend shown when looking at the scatter data.



**Figure 4.7.** Normalised monthly hydroclimatic data compared with normalised monthly abstraction data for the period 1<sup>st</sup> January 2012 to 30<sup>th</sup> September 2015, for all study catchments combined. (**A1**, **B1**, **C1**) Paired comparison, with black dashed lines representing trend direction; (**A2**, **B2**, **C2**) time series comparison. Combined data is based on an average across all catchments for temperature, while precipitation, streamflow and abstraction volumes are all total volumes summed from all catchments.

# 4.5 Discussion

# 4.5.1 Hydroclimatic trends

The results of the trend, breakpoint, and correlation analyses show a selection of spatially varying changes across the five catchments over the 34-year study period. Wales has a maritime climate which is strongly influenced by the North Atlantic Oscillation (NAO), a major cause of atmospheric circulation variability in the north Atlantic region (Dixon *et al.*, 2006;

Beranová & Huth, 2007). This is important to consider when analysing changes that have occurred across all catchments. NAO displays year-to-year and longer-term variability, therefore breakpoints in temperature, precipitation, and streamflow data that occur across all five catchments and likely to be rooted in changes in NAO. However, when explaining changes that affect a subset of the catchments, it is likely that catchment characteristics in terms of topography, LULC and surrounding landscapes have had an impact by interacting with these large-scale meteorological processes (West *et al.*, 2019). The breakpoint analysis is a prime example of this interplay, a combination of NAO, climate change and other non-climate/weather related changes at a catchment and at a national level are likely to have impacted on the climatology breaks shown. No single factor accounts for all of breaks shown across the catchments. While a change in NAO is likely to have been a key driver, other factors such as LULC change and climate variations may well have had an impact on spreading the breakpoints out over the range of approximately seven years shown (1987-1994).

When looking at both temperature and streamflows, a north-south divide in results can often be seen. Taking the extreme events results for streamflows as an example, it can be seen that in the two northerly catchments, the Clwyd and Conwy, the largest annual 1-day discharge events have become larger over the study period. Meanwhile the two most southerly catchments, the Teifi and Tywi, have seen the smallest 1-day flow values becoming larger, in particular when looking at annual and summer flows. These results are consistent with the findings of Dixon et al. (2006) who showed that in the period 1977-2001, summer minimum flow volume values had increased significantly in south Wales catchments, but not the north—mainly due to the more mountainous terrain of north Wales (Figure 4.1). Furthermore, increases in the lowest summer flow values were also seen in Osborn and Hulme (2002), and were linked to increased light summer rainfall over the period. The Dyfi catchment in mid-Wales shows a combination of these two patterns with maximum and minimum 1-day flows increasing in the summer and annually, showing a shift to generally wetter summers over the period, again corresponding to the reasoning and results put forward in Dixon et al. (2006) for mid-Wales. These results are also consistent with the mean catchment slope data shown earlier (Table 4.2) with the Dyfi and Conwy in particular being the steepest catchments, and arguably the flashiest, of the five studied. This factor may well contribute to the increase in summer maximum flows observed over the study period.

It is interesting to note that the aforementioned changes in streamflows occur despite a lack of corresponding statistically significant change in annual or seasonal average precipitation, an observation also made by Macdonald et al. (2010). This mismatch of significant trends between precipitation volumes and streamflows is surprising, as the two factors are correlated significantly in all catchments, suggesting that flows in the catchments are highly sensitive to small changes in precipitation. Similarly, the only statistically significant trend seen in extreme precipitation events, a decreasing trend in cumulative spring rainfall in the Clwyd, does not relate to a corresponding change in average or extreme spring streamflows during the study period. Although not analysed in this study, it is also possible that changing LULC during the study period could have had an effect on streamflows, independent of precipitation. Changes in agricultural land use in particular could have a large impact in the study catchments, affecting processes such as water infiltration and runoff (Wheater & Evans, 2009). Between the early 1980s and the late 2000s there was an increase in the amount of agricultural land classified as permanent grassland and rough grazing, despite overall agricultural land area remaining relatively stable (Welsh Government, 2019c). From the late 2000s to present, the total amount of agricultural land has grown to its highest level since World War II, with the vast majority of this being for grazing (Welsh Government, 2019c). Furthermore, forest cover has been continually increased throughout the study period, with greater planting taking place at the start of the study period and the rate steadily declining throughout (Welsh Government, 2015a). Forested land also impacts on the connectivity of river flows and precipitation, with processes such as evapotranspiration and interception storage diverting water from, or delaying water reaching, rivers (Wheater & Evans, 2009). This could once again explain some of the mismatch between significant precipitation and streamflow trends.

Changes in seasonal and annual averages, as well as extreme events are most observed in temperature, for the factors studied. Once again, a north-south divide can be seen in parts of these results, with the three southernmost catchments displaying a decrease in both the annual number of days that are hotter than the dataset 95<sup>th</sup> percentile, and the number of summer days that are colder than the dataset 5<sup>th</sup> percentile. These two changes suggest a narrowing of the temperature ranges in these catchments, with less extreme hot days annually and less extreme cold days in the summer. These findings are in line with wider UK
research such as Dessai & Hulme (2008) who showed, via the Central England Temperature record, that annual average summer temperatures increased between 1960-2007 when compared to a 1961-1990 mean. Nesbitt et al. (2016) also presented a trend for summer days becoming warmer over the period 1954 to 2012 for south-east and south-central UK, as did Luterbacher et al. (2004) for the period 1977 to 2003. Annual NAO index has been largely negative on average throughout the study period, especially so in the latter two thirds (Figure 4.8). Negative NAO has been linked to colder maximum temperatures (Beranová & Huth, 2007), while climate change has been causing increasing summer temperatures on average. These two factors combined could be contributing to the narrowing temperature range observed, with NAO bringing down the maximum, and climate change bringing up the minimum. Furthermore, in all catchments, an increase in average autumn temperatures across the study period has been observed, with this being slightly more pronounced in the two north Wales catchments. All of the observed trends also fit with general UK observations of a greater degree of warming in the south of the country than the north, and the exacerbation of the temperature gradient between them. Furthermore, the north-south divide also fits when considering the surrounding geography of the catchments, the mountainous Snowdonia region lies just to the south of the two most northerly catchments, heavily influencing the climate here and causing local variation in the weather that is brought in from the Atlantic (Dixon et al., 2006).



**Figure 4.8.** Annual North Atlantic Oscillation (NAO) index 1825-2019 relative to the 1951-1980 average index value (0 on this plot). Our study period has been highlighted in yellow; updated and adapted from Osborn (2011).

#### 4.5.2 Weather and demand for water

When looking at the correlations between daily and monthly total abstraction data and hydroclimatic factors, it is clear that both average temperature and precipitation volume play a crucial role in influencing the total amount of water abstracted; volumes abstracted increasing in higher temperatures and decreasing on wetter days. Given that the data takes account for total abstraction volume, these relationships are not surprising, as it is likely that on hotter days there will be both more domestic demand (more showers, car washing, garden watering etc.), as well as higher demand from agriculture, in particular for water for livestock in the catchments studied. The opposite is true for wetter days, which reduce the agricultural demand from public water supplies and reduce domestic demand, especially for water use on external areas of a property. These overall relationships do fit with other work that has sought to quantify the relationship between climate variables and domestic demand, both in the UK and abroad. Slavíková et al. (2013) showed that in the Czech Republic, air temperature relative to the season average, accounted for most variability in residential water consumption. Similarly in the UK, Goodchild (2003) showed that domestic water demand in 41 houses in an Essex case study correlated with an R<sup>2</sup> value of 0.44 with days when maximum temperatures were over 25 °C. Additionally, total daily sunshine hours were shown to have a stronger correlation with demand, having an R<sup>2</sup> value of 0.53 (Goodchild, 2003). Work by Xenochristou et al. (2020) also found that sunshine hours and air temperature were the most influential weather variables on domestic demand, along with humidity.

However, using total abstraction volumes as a proxy for demand does present some challenges. For example, factors such as the usage of water internally, within drinking water treatment plants, as well as network water leakage, may mask or alter observed trends in the dataset. In addition, it is difficult to apportion the total abstraction volume to different user groups, such as domestic, industrial, and agricultural on a daily basis. Furthermore, different user groups may have different relationships with weather conditions, for example Massoud *et al.* (2018) showed for California's Central Valley, that while agricultural demand increased in drier years, precipitation volume had little impact on urban water demand. Additionally, we recognise that the comparison period between the hydroclimatic factors and abstraction data is relatively short in this study, at four years. This does not give the opportunity to investigate relationships such as the effect of prolonged drought on water use, or other

longer-term patterns. Nevertheless, the overall abstraction data does give a baseline relationship to work with when considering the impact of future climate change on total water demand.

#### 4.5.3 Study implications

Our study has shown that the climate of Wales has changed since the early 1980s, and that this will have contributed to both the supply and demand of water in the region. On the basis of the results found, it is clear that these changes have been more keenly felt in terms of average air temperatures than precipitation volumes; this applies to both annual and seasonal averages as well as extreme events. When looking at the impact that these climatic changes have had on streamflows it can be seen that seasonal and annual average flows have remained largely unchanged, instead it is extreme flow events that have been more greatly affected. These changes occur with a north-south divide, with the largest annual flow events getting larger in the north, and the south becoming less dry in the summer when looking at the lowest flows. These changes could have important impacts if they are continued into the future, with implications not only for water supply, but also in terms of water resource management, to prevent flooding and other related natural disasters. Furthermore, large industrial users as well as applications such as hydroelectric developments could be impacted by changes in river regime. In particular the viability of some small-scale hydroelectric installations could be called in to question in some areas due to changing streamflow characteristics and flow duration curves, further emphasising the importance of a solid understanding of the relationship between, and emerging trends in, hydroclimatic factors.

Climate change induced alterations in the future timing, quantity, and quality of water available for supply, as well as policy relating to adaptation and management methods to cope with these predicted changes, needs to be further researched for Wales. In particular, hydrological modelling studies comparing current baseline streamflow (such as that presented in this study) to future streamflow under various climate change scenarios, could prove to be particularly useful. This work is crucial to better inform future water supplydemand dynamic assessments, water resource management, and adaptation planning. This is especially true if projected increases in the reoccurrence, duration, and intensity of extreme events under future climate change, as suggested in other research (Slingo, 2014; King &

Karoly, 2017; Betts *et al.*, 2018; Met Office, 2019; Harkness *et al.*, 2020), are correct. This suggested research would also go some way to addressing some of the aforementioned priority research questions laid out in Brown *et al.* (2010); but must however also keep in mind the cost and practicality of adaptation measures, in order to ensure both a continued unbroken water supply service, and affordable water for all. More broadly, this research has shown the need to research and understand historical trends and future projections of hydroclimatic factors at local, catchment levels. This is clear when looking at the presented differences in observed trends seen over small spatial distances, due to changing land characteristics, and it is these nuances that are vital to incorporate into future planning for any industry that relies on surface water abstraction.

## 4.6 Conclusions

This research has highlighted the potential for water scarcity problems even in a relatively water-rich region such as Wales. For example, with observed trends such as warmer average autumn temperatures providing for potentially greater water use in the season, the pressures on summer water supply could in the future extend further into the autumn. Although potentially increased demand could be countered by a trend of the largest discharge events becoming larger in the north of Wales, and summers becoming less dry in the south of Wales, any increase in flow is of little use if the capacity to store this additional water is not sufficient to make use of it.

Finally, we suggest that further research should focus on how future climate change will affect the relationship between weather factors, streamflow, and water demand, both in Wales and globally. For example, research concerning trigger temperatures for significant increases in water use, or the effect of long-term higher than averages temperatures on water demand, would aid understanding of the finer detail of the dynamic between hydroclimatic factors and total water abstracted. We also hope that this study will set a frame onto which future climate change research focusing on surface waters, and the future provision of water services can be built; being one of the first steps in securing the long-term sustainability of water supply services in the region and further afield.

Acknowledgements: This study was conducted as part of the Dŵr Uisce project, which aims at improving the long-term sustainability of water supply, treatment and end-use in Ireland and Wales. The project has been supported by the European Regional Development Fund (ERDF) through the Interreg Ireland-Wales Co-operation Programme 2014-2020. The work has been used as a case study for Dŵr Uisce Work Package 7 "Climate Change", and forms part of a study assessing the impact of climate change on water resource exploitation in Wales. The authors would like to thank Dŵr Cymru Welsh Water for the provision of actual abstraction volume data for the catchments studied. We also thank two anonymous reviewers for their comments and helpful suggestions, which have improved the paper.

# CHAPTER 5: STUDY II

# Modelling the impact of future climate change on streamflow and water quality in Wales, UK

R.J.H. Dallison, A.P. Williams, I.M. Harris and S.D. Patil School of Natural Sciences, Bangor University

This chapter has been submitted in full for publication in the *Hydrological Sciences Journal* and was undergoing peer-review at the time of submission of this thesis.

## **Author contributions**

**Richard Dallison:** conceptualisation, methodology, software, validation, formal analysis, investigation, data curation, writing – original draft, writing – review & editing, visualisation **Prysor Williams:** writing – review & editing, supervision

Ian Harris: resources, writing – review & editing

**Sopan Patil:** conceptualisation, software, writing – review & editing, supervision

# 5.1 Abstract

Climate change is likely to have a major impact on future hydrological regimes, impacting numerous sectors reliant on surface waters. We use the Soil and Water Assessment Tool (SWAT) to model future (2021-2080) streamflow and water quality variables (nitrogen, phosphorus, suspended sediment, and dissolved oxygen), in five catchments in Wales, under a worst-case scenario of future greenhouse gas concentrations (RCP8.5). Results show a decline in annual average flows (-4% to -13%) but larger changes seasonally (spring, up to 41% increase; autumn, up to 52% reduction). The magnitude and frequency of high flow events increases in spring (magnitude: Sen's slope range 0.165-0.589, p <0.01), with more low flows in autumn (Sen's slope range 0.064-0.090, p <0.01). Water quality declines, with higher nitrogen, phosphorus, and sediment concentrations and lower dissolved oxygen levels. The findings have economic and environmental implications for abstractors, as water resources will become more unreliable, seasonal, and polluted.

**Keywords:** Hydrological modelling; Particle swarm optimisation; Trend analysis; UKCP18; Water resource management; Water supply

## 5.2 Introduction

Globally, river catchments are likely to face significant changes due to a variety of factors, including climate change, land use/land cover change, increasing urbanisation, increased population, and greater demand for water (Rockström *et al.*, 2009; Gleick & Palaniappan, 2010; Heathwaite, 2010; Cosgrove & Loucks, 2015; Bijl *et al.*, 2016). The Intergovernmental Panel on Climate Change (IPCC), in their fifth assessment report (AR5), project a rise in both surface temperatures and the number and intensity of extreme precipitation events under all future greenhouse gas emissions scenarios throughout the 21<sup>st</sup> century (IPCC, 2014b). For the UK, a small increase is projected in average precipitation by 2100, compared to the 1986-2005 baseline; average temperatures are also expected to increase (IPCC, 2014b). Furthermore, the UK Climate Projections 2018 (UKCP18) Science Overview Report suggests a move towards warmer, wetter winters, and hotter, drier summers, as well as an increase in extreme weather events (Lowe *et al.*, 2018). These changes will have implications for catchments in terms of annual and seasonal average flows, high and low flows, and water quality.

When studying past trends, various studies have shown that alterations in streamflow patterns have already occurred due to climate change, as well as other factors. Hannaford (2015), conducted a review of climate-induced changes specifically on UK streamflows, using the 130 benchmark network gauging stations with near-natural flow regimes. The results show an increase in annual outflows, albeit only statistically significantly in Scotland, with greater seasonal variability. Winter and autumn average flows are shown to have increased between 1961-2010 across the UK, with spring and summer flows increasing in Scotland and Wales, and decreasing in England (Hannaford, 2015). High flows are also shown to have increased in Scotland and Wales significantly (p < 0.05), rising 22.4% and 26.5% respectively over the 50-year period. These results are consistent with studies such as Harrigan *et al.* (2018), which shows similar trends in UK catchments over the period 1965-2014; for example a UK-wide increase in annual and winter mean flows. In contrast to Hannaford (2015) however, large decreases in spring flow were seen across Britain (Harrigan *et al.*, 2018). Additionally, Dixon *et al.* (2006) found for catchments in western Britain, that there was an increase in annual, autumn and winter maximum flows; but little change in annual average

flows. Furthermore, Dallison *et al.* (2020)<sup>2</sup> analysed 34 years of historical streamflow and climate data for five catchments in Wales and also found little change in annual average flows. However, low flow volumes were shown to have increased in the summer in the three most southerly catchments, while annual maximum flow increased in the northerly catchments.

Several studies have analysed future seasonal and annual average streamflows for the UK, mostly using UK Climate Projections data (UKCIP02, UKCP09 & UKCP18). While little work has been published using the new UKCP18 data, Kay et al. (2020) compared it with the UKCP09 projections across ten catchments in the UK and found the estimates of change under both models were similar, although uncertainty was higher in the most recent version. A study using UKCP09 data by Watts et al. (2015a) suggests that although little change is seen in annual average rainfall, seasonal changes are more pronounced. Precipitation increases in winter are likely to be greatest along the western coast of the UK, while the largest summer decreases in precipitation are expected in southern England (Watts et al., 2015a). Investigating the impact of these changes on river flows in the short and long-term respectively, Christiersen et al. (2012) and Prudhomme et al. (2012) both projected annual averages flows to remain relatively stable. Both studies also project a reduction in spring flows, with small increase in the winter. However, the most consistent trend seen is a decrease in summer flows across the UK, especially in the north and west (Christierson et al., 2012; Prudhomme et al., 2012). These results are consistent with the findings of Chun et al. (2009) and Charlton & Arnell (2014) which both state that, despite variations due to different catchment characteristics, UK annual average streamflows will decline, with the magnitude of change less than that seen seasonally.

For extreme events in the UK, such as large discharge events/flooding and long-term drought, it is likely that climate change with alter their frequency, duration, and severity (IPCC, 2014b; Watts *et al.*, 2015a; Sayers *et al.*, 2016; Miller & Hutchins, 2017). Extreme streamflow changes are mainly driven by changes in precipitation regime. Fowler & Ekström (2009), when examining extreme rainfall for example, forecast increases in winter, spring and autumn to the 2080s, with this being especially pronounced in the west and Wales. Furthermore, Rau *et al.* (2020) found that extreme hourly precipitation is set to increase under a high future

<sup>&</sup>lt;sup>2</sup> As presented in Chapter 4.

emission scenario, especially in the long-term (2070-2100), up to 112% in some areas. Collet *et al.* (2018) demonstrated using the future flows hydrology database (Prudhomme *et al.*, 2013b), that for western areas of Britain especially, an increase in the frequency, duration and magnitude of future flood events is likely when comparing the period 1961-1990 to 2069-2098. A large, but less likely increase in the duration and magnitude of droughts is also shown (Collet *et al.*, 2018). Similarly, Charlton & Arnell (2014) showed an increase in high flows and decrease in low flows in six study catchments using the UKCP09 data. These changes were particularly linked to changes in winter precipitation for high flows, and summer temperature and precipitation for low flows (Charlton & Arnell, 2014).

In terms of changes to water quality, it is likely that the aforementioned projected alterations in average and extreme streamflows will have a deleterious impact in many instances. However, individual catchment characteristics will have a large impact on the magnitude of these changes. Whitehead et al. (2009a), for example, showed across six UK catchments that patterns in future water quality varied greatly based on the geology, topography, urbanisation, agricultural land use and point source inputs of each catchment. Furthermore, their research also showed that variation is also possible within catchments. For example, the River Tweed showed a trend of decreasing nitrate levels in the upland headwaters in summer, but increasing levels in the lower reaches (Whitehead et al., 2009a). A further study by Whitehead et al. (2009b) discussed the varied future drivers of water quality changes. These drivers include variations in flow influencing stream power and therefore the erosion and transport of sediment. Streamflow regime variations also impact the dilution and movement of pollutants; in addition, changing water temperature can alter in-stream chemical process. Delpla et al. (2009) also reviewed the impact of both flood and low flow events on water quality, finding that increases in the magnitude of rain events, as well as changing droughtrewetting cycles, are likely to increase the amount of organic matter reaching streams. These findings correspond with those shown by Ockenden et al. (2017), who showed a clear positive trend between total phosphorous load and annual rainfall, and therefore streamflow. Furthermore, Mortazavi-Naeini et al. (2019) demonstrated a positive relationship between suspended sediment concentrations and extreme winter floods, as well as an increase in total phosphorus levels during low flow events. Finally, Watts & Anderson (2016) suggest that generally lower summer flow will increase the likelihood of algal blooms and lower dissolved

oxygen levels, while subsequent summer flood events will increase wash of pollutants such as nutrients into river systems.

This work aims, via hydrological modelling, to investigate alterations to streamflow and water quality brought about by climate change in five catchments in Wales, UK. This is arguably an under-studied region of the UK in this context, especially when compared to drier regions, such as southeast England. It is a key area to study however, due to the importance of the river networks to the region. An obvious example of this is public water supply, 95% of which is taken from surface water sources (DCWW, 2019b). With some water supply zones already being in deficit in the region (DCWW, 2019b), future water supply sustainability is crucial to understand. Furthermore, Wales exports large quantities of water to metropolitan areas in central and northern England (Warren & Holman, 2012; DCWW, 2016a), meaning that alterations in future flow regime could have impacts beyond the country's borders. In addition, understanding future flow characteristics is important to sectors such as hydroelectricity generation, which is abundant, and growing, in Wales. Over 360 hydroelectric projects operate on rivers in Wales, ranging from small (<1 MW) to large (>70 MW) with a total estimated generation off 389 GWh, accounting for approximately 1.3% of total electricity generation in Wales in 2018 (Welsh Government, 2019b). Surface waters are also an important part of the tourism economy, with 96 million day visits to Wales in 2018 (Welsh Government, 2020), many attracted to the region for its natural beauty, and often visiting rivers, lakes and coast. The industry contributed £3,064 million to the Welsh economy in terms of gross value added in 2016 (Welsh Government, 2020), therefore understanding changes that may affect the natural landscape and resources of the region, and its continued appeal to tourists, is crucial. Finally, any changes in flow characteristics or water quality in particular, could have important implications for river ecology and aquatic life in both rivers and coastal waters. To ensure sufficient protection, it is important to understand potential future changes and challenges now, and to plan mitigation measures.

# 5.3 Data and methods

## 5.3.1 Study catchments

Our study focusses on five catchments in Wales: the Clwyd, Conwy, Dyfi, Teifi, and Tywi (Figure 5.1). These catchments have been selected for study due to the variety of different characteristics displayed, in factors such as topography (Figure 5.1), land use/land cover (LULC), and soil types (Figure 5.2). These are also among the largest catchments contained, source to sink, entirely within Wales; key details relating to the characteristics of each catchment are provided in Table 5.1. In addition, the catchments are important economically, socially and environmentally, in line with the reasons for selecting Wales as a study region. Water from these catchments, for example, directly contributes to the supplies of over 975,000 people (derived from DCWW, 2019b), as well as over 70 hydroelectricity scheme abstractions (derived from NRW, 2019a). Furthermore, these catchments are also important environmentally, with various land protections and designations covering areas of the catchments, such as National Parks, Sites of Special Scientific Interest, National Nature Reserves, and Special Areas of Conservation. Finally, the catchments also flow into waters that are spawning and/or nursery grounds for a variety of fish species such as cod, sole, plaice, spotted ray and sandeel (Ellis *et al.*, 2012).



Figure 5.1. Study catchments as delineated in SWAT. (A) Streams and stream orders (derived by Strahler method), gauging stations and water quality monitoring locations. (B) Catchment elevation and streams (third order and larger).



Figure 5.2. (A) Catchment land use/land cover categorisation derived from CORINE Land Cover data (EEA, 2012). (B) Catchment soil classification derived from the EU Soil Database (European Commission, 2004).

		Clwyd	Conwy	Dyfi	Teifi	Tywi
Catchme	nt area (km²)	750.1	541.8	507.2	995.3	1364.6
Maximu	m catchment elevation (m)	557	1062	903	591	801
Mean ca	tchment elevation (m)	210.4	309.8	271.4	198.1	212.5
	Urban	2.4	0.7	0.3	0.8	0.7
	Agriculture	80.9	30.6	30.0	83.0	64.6
۲ (%	Arable	9.3	0.1	0.4	1.8	0.7
Iave	Pasture	71.6	30.5	29.6	81.2	63.9
с р	Forest	7.3	13.7	21.5	5.5	16.0
lan	Broadleaf	1.6	2.7	1.8	1.2	1.0
se/	Coniferous	4.5	5.5	18.9	3.2	10.1
л q	Mixed	1.2	5.5	0.8	1.1	4.9
lan	Scrub	9.4	42.1	45.3	9.7	17.1
ent	Natural grassland	2.3	16.6	30.4	7.2	8.4
Ē	Moors and heathland	6.6	23.6	11.3	1.2	6.5
atcl	Transitional woodland scrub	0.5	1.9	3.6	1.3	2.2
0	Peat bog	0.0	8.4	2.6	0.9	1.3
	Sparsely vegetated areas	0.0	4.1	0.2	0.0	0.0

**Table 5.1.** Key study catchments details. Catchment area refers to area modelled in SWAT. Elevation data derived from 5 m resolution OS Terrain 5 DEM from Ordnance Survey; land use/land cover data derived from 2012 CORINE Land Cover data (EEA, 2012).

The catchments range in size from Dyfi at 507 km<sup>2</sup> to Tywi at over two and half times the size (1365 km<sup>2</sup>). In terms of LULC, agriculture and scrubland are the two main land cover types across the five catchments, with Clwyd, Teifi, and Tywi being heavily agriculture based (83% to 65%), and Conwy and Dyfi having higher propositions of scrubland (42% and 45%, respectively) such as natural grasslands, moors and heathlands. Most of the agricultural land in all catchments is pasture used for grazing sheep and cattle, with Clwyd having the highest proportion used for growing crops. Urban areas account for little land area in all our study catchments. Where urban areas are present, they tend to be located in the coastal regions. Forested land is the third largest land cover in all catchments, with this mostly being coniferous and often located in mid to high elevations. In terms of underlying soil classification, the vast majority in all catchments is young Cambisol (brown soil), while Podzols are most commonly found in higher altitude areas.

## 5.3.2 Future climate change data

UKCP18 data were used as future weather input for hydrological modelling, specifically the 'Regional Projections on a 12 km grid over the UK for 1980-2080' dataset (MOHC, 2018b). This provided an ensemble of twelve regional climate model (RCM) projections derived and,

dynamically downscaled, from the 60 km HadGEM3-GC3.05 global coupled model (GCM) perturbed parameter ensemble (Murphy *et al.*, 2018). The 12 RCM ensemble members are modelled on the basis of no downturn in greenhouse gas emissions, using the Representative Concentration Pathway 8.5 (RCP8.5), therefore taking a worst-case scenario approach in a similar vein to studies such as Ficklin & Barnhart (2014) and Lee *et al.* (2018). Uncertainty in the GCM, as well as the natural climate variability cause the 12 ensemble members to differ (Kendon *et al.*, 2019). The 12 km gridded UKCP18 data were extracted to a single point at the centre of each grid square, and these points were treated as pseudo-meteorological stations. Out of the total 28 grid squares that have the centre point lying within one of the five catchments, 13 were selected as pseudo-meteorological stations (Figure 5.3).



**Figure 5.3.** Pseudo-meteorological stations used in SWAT future weather input, those selected for use in the study are highlighted in green and labelled.

These selections were made to ensure good coverage across all catchments, considering the potential changes in precipitation and temperature based on elevation, and to manage the computational resource limitations in the process of extracting grid values to the point locations. Where catchments share a boundary (Clwyd and Conwy; Teifi and Tywi), all pseudo-meteorological stations in both catchments were made available as inputs to each individual catchment model to provide additional weather data coverage.

## 5.3.3 Hydrological modelling

In this study, we have used the Soil and Water Assessment Tool (SWAT), a physically based, semi-distributed hydrological model developed by the US Department of Agriculture's Agricultural Research Service (Arnold et al., 1998). SWAT has been used internationally and for a variety of purposes due to the wide range of catchment processes that the model considers (Gassman et al., 2014). In recent years, SWAT has been used for purposes such as ecological impact assessments (Chambers et al., 2017; Kakouei et al., 2018), crop/agricultural planning studies (Fereidoon & Koch, 2018; Yang et al., 2018), hydropower assessment (Park & Kim, 2014; Haguma et al., 2017; Abera et al., 2018), and future flood flow characterisation (Singh & Goyal, 2017; Xu et al., 2019). The model has also been widely used for investigating changes in the quantity and timing of streamflows under climate change in a variety of catchment types; for example, Coffey et al. (2016) for two catchments in Ireland, Perra et al. (2018) for a Mediterranean catchment, Sultana & Choi (2018) for a snow-dominated catchment in northern California, and Yuan et al. (2019) for the Yangtze River in China. Water quality impacts under future climate change have also been studied using SWAT, with papers such as Nerantzaki et al. (2016) and Jilo et al. (2019) studying the sediment yield in catchments in Greece and Ethiopia, respectively; Pesce et al. (2018) inspecting nitrogen, phosphorous and dissolved oxygen in a catchment in Italy, and Yang et al. (2017) investigating total nitrogen changes in a river in northeast China.

SWAT operates at a daily time-step and is designed for river basin-scale use. Being semidistributed, SWAT makes use of sub-basins, which are further divided into Hydrological Response Units (HRUs) of areas with similar soil type, LULC and topography. The model is computationally efficient due to calculations being performed for each HRU and later summed together at sub-basin outlets (Bailey *et al.*, 2016). Full and detailed information on

the equations, inputs and processes that govern the behaviour of the model, as well as the outputs, are provided in the SWAT theoretical documentation (Neitsch *et al.*, 2011) and the input/output documentation (Arnold *et al.*, 2012).

Here, SWAT has been used to simulate daily streamflow and four water quality parameters, total nitrogen (TN), total phosphorus (TP), suspended sediment (SS), and dissolved oxygen (DO). Each study catchment was set-up, calibrated, and validated separately. The key inputs to the SWAT model are elevation, LULC, soil characteristics, and weather data. The 5 m resolution OS Terrain 5 DEM from Ordnance Survey was used alongside soil data from the EU soil database (European Commission, 2004) and land cover data from the CORINE Land Cover dataset (Copernicus Land Monitoring Data; EEA, 2012). Historical daily observed air temperature and precipitation data were obtained from the Centre for Ecology and Hydrology's (CEH) Climate, Hydrology and Ecology research Support System (CHESS) dataset (Robinson *et al.*, 2017). Historical streamflow data was obtained from CEH's National River Flow Archive dataset (NRFA, 2020).

Calibration and validation was completed with the SWAT Calibration and Uncertainty Programme 2012 (SWAT-CUP), specifically using the Particle Swarm Optimisation (PSO) method (Kennedy & Eberhart, 1995). The PSO method iteratively improves the model simulation proficiency, taking into consideration the previous best-known calibration point, and allows for a quicker and more efficient calibration process (Clerc & Kennedy, 2002; Coello *et al.*, 2004). SWAT was calibrated at each study catchment against observed daily average streamflow only. Calibration was undertaken for the period 1985 to 1998, with a 3-year model warm-up period occurring beforehand and ten simulations with ten iterations per simulation were completed for each catchment calibration. Kling-Gupta efficiency (KGE; Equation 5.1) was used as the goodness-of-fit metric for model calibration (Gupta *et al.*, 2009):

$$KGE = 1 - \sqrt{(r-1)^2 + (a-1)^2 + (\beta-1)^2}$$
 (Eq. 5.1)

where r is the linear correlation between observations and simulations;  $\alpha$  refers to the ratio between the standard deviation in the simulated flow and the standard deviation in the observed flow; and  $\beta$  is the ratio between the mean simulated and mean observed flow. The calibrated SWAT model was then used for a validation run for the period 1999 to 2014, once again with a 3-year warm-up period beforehand. Calibration and validation KGE values for each catchment are shown in Table 5.2<sup>3</sup>. A total of 17 parameters were used in the calibration, the names and details of these are provided in Table 5.3. These parameters were selected based on a literature review of similar catchments, and after consultation with experienced SWAT modellers.

**Table 5.2.** Number of hydrological response units (HRUs) and sub-basins per catchment model. Number of subbasin containing gauging station also show, along with Kling-Gupta efficiency (KGE) values for the calibration and validation periods.

Catchment	Total HRUs	Total sub- basins	Gauge sub- basin (outfall)	Calibration KGE	Validation KGE
Clwyd	697	302	297	0.810	0.788
Conwy	770	202	194	0.770	0.718
Dyfi	662	227	223	0.788	0.728
Teifi	1333	425	420	0.851	0.723
Tywi	1697	496	459	0.841	0.717

Table 5.3.	SWAT	parameters	calibrated	through	SWAT-CUP	programme	using	particle	swarm	optimisation
method.										

Parameter	Description	Input file location
ESCO	Soil evaporation compensation factor	.bsn
EPCO	Plant uptake compensation factor	.bsn
SURLAG	Surface runoff lag time	.bsn
GW_Delay	Groundwater delay	.gw
Alpha_BF	Baseflow alpha factor	.gw
GWQMIN	Threshold depth of water in shallow aquifer for return flow to occur	.gw
RCHRG_DP	Deep aquifer percolation fraction	.gw
REVAPMN	Threshold depth of water in shallow aquifer for "revap" to occur	.gw
GW_REVAP	Groundwater "revap" coefficient	.gw
ALPHA_BF_D	Baseflow alpha factor for deep aquifer	.gw
CANMX	Maximum canopy storage	.hru
CN2	SCS runoff curve number for moisture condition 2	.mgt
CH_N2	Manning's "n" value for the main channel	.rte
СН_К2	Effective hydraulic conductivity in main channel alluvium	.rte
SOL_AWC	Available water capacity of the soil layer	.sol
SOL_K	Saturated hydraulic conductivity	.sol
SOL_Z	Depth from soil surface to bottom of layer	.sol

<sup>&</sup>lt;sup>3</sup> Charts comparing observed and simulated streamflow for each catchment during the calibration period can be found in Appendix 1, Figures A1.1 to A1.5.

Due to the lack of sufficient historical sampling data, water quality parameters were not calibrated. However, a validation of the water quality output was attempted, by comparing model output to the limited measured water quality data available from the Natural Resources Wales (NRW) water quality data archive (NRW, 2019b). While the water quality validation showed a good degree of correlation for DO in most catchments, the correlations for the other factors were less agreeable (Table 5.4). The modelled TP, TN and SS output do all correlate well with modelled streamflow however (Table 5.5), suggesting that this is the key driver of change in the uncalibrated model.

**Table 5.4.** Pearson's correlation values between observed (where available) and modelled water quality concentrations. Underlined values indicate a correlation significant at p < 0.05, those shown in bold are significant at p < 0.01.

Catchment	Suspended sediment	Total phosphorous	Total nitrogen	Dissolved oxygen
Clwyd	-0.033	0.174	0.109	0.646
Conwy	0.200	-0.026		0.022
Dyfi	-0.139	-0.090		0.060
Teifi	0.099	0.024	0.075	0.806
Tywi	0.788	-0.172	0.116	0.788

**Table 5.5.** Pearson's correlation values between modelled water quality concentrations and streamflow. Underlined values indicate a correlation significant at p < 0.05, those shown in bold are significant at p < 0.01.

Catchment	Suspended sediment	Total phosphorous	Total nitrogen	Dissolved oxygen
Clwyd	<u>0.033</u>	<u>0.035</u>	0.063	0.185
Conwy	0.654	0.626	0.707	-0.539
Dyfi	0.923	0.602	0.595	-0.588
Teifi	-0.022	-0.024	0.361	0.204
Tywi	0.091	-0.093	0.442	0.226

Future runs of the calibrated and validated SWAT model were completed for the hydrological years 2021 to 2080, where a hydrological year runs from 1<sup>st</sup> October to the following year's 30<sup>th</sup> September and is named after the later year. Some discontinuity in the model output is caused by the need to run the models in sections due to database size restrictions on the output files. This was mitigated as far as possible by including a 7-year warm-up period at the start of all model run sections. Model run sections were as follows, Conwy and Dyfi: 2020-2050, 2051-2080; Clwyd: 2020-2040, 2041-2060, 2061-2080; Teifi and Tywi: 2020-2035, 2036-2050, 2051-2065, 2066-2080.

#### 5.3.4 Trend analysis

To assess any trends displayed in the future climatic, streamflow, and water quality data, the Mann-Kendall (MK) test was performed (Mann, 1945; Kendall, 1975). MK tests for a monotonic positive or negative change in a time series by calculating Kendall's tau (association test) between each value and all of the proceeding values (Helsel & Hirsch, 2002). The nature of hydrological data lends itself to this nonparametric method as it is generally non-normally distributed and displays seasonality, going against a constant distribution assumption (Kundzewics & Robson, 2004). Additionally, auto-correlation is often also displayed in the data, making a parametric test further unsuitable; it is for this reason, that during the MK test the Hamed & Rao method of auto-correlation correction was also applied (Hamed & Rao, 1998). Sen's slope estimator (Sen, 1968) was also used in order to better estimate the size and direction of any detected trends.

Trend analysis for both streamflow volumes and water quality factors were conducted for annual (hydrological years) and seasonal (winter, December to February; spring, March to May; summer, June to August; autumn, September to November) averages. Streamflow extremes were also considered, testing seasonally and annually for trends in 1-day maximum flow, 1-day minimum flow, and number of low flow days (defined as days were streamflow is less than the 5<sup>th</sup> percentile value for the full 60-year dataset). High flow events have also been analysed, with the number of independent events per year with at least one streamflow peak above the 60-year 95<sup>th</sup> percentile value being calculated. Change in event length was also analysed, being defined as the period either side of any peak flow where streamflow was greater than baseflow; at the start and end of each event streamflow is therefore equal to baseflow. The digital filter method defined by Sawicz *et al.* (2011) has been used to calculate direct runoff, as per Equation 5.2, to subsequently enable the calculation of baseflow:

$$Q_{Dt} = cQ_{Dt-1} + \frac{1+c}{2}(Q_t - Q_{t-1})$$
 (Eq. 5.2)

where  $Q_{Dt}$  is direct runoff at a given time-step, c is a constant with value 0.925,  $Q_{Dt-1}$  is direct runoff at the previous time-step (at time-step t = 0,  $Q_D$  is assumed to be 0) and Q is total streamflow. Baseflow ( $Q_{Bt}$ ) for the same time-step was then calculated using Equation 5.3:

$$Q_{Bt} = Q_t - Q_{Dt} \tag{Eq. 5.3}$$

An event containing multiple peaks above the dataset 95<sup>th</sup> percentile is classified as a single event. In addition, a trend analysis was also performed on the input future temperature and precipitation data to identify any annual or seasonal changes occurring.

## 5.4 Results

Our results show a large decrease in summer and autumn average streamflows, with a corresponding increase in spring and winter flows. Overall, these seasonal changes result in a small decrease in annual average flows. Trends in extreme streamflows show much greater variability between catchments and seasons. However, all catchments display a trend for an increasing number of extreme low flow days, as well as lower 1-day minimum flow volumes. Water quality appears to be highly linked to streamflow, especially changes in extreme high and low flows. Summer, in particular, shows lower water quality standards across all catchments and variables. To provide further context to the streamflow and water quality results, we first include a brief trend analysis of seasonal and annual average mean temperatures and precipitation. As with all of the trend analysis subsequently presented, this has been completed using the MK trend test. Due to the lack of calibration of water quality parameters, only long-term seasonal and annual average trends have been presented in this study, unlike streamflows, where extreme flows have also been analysed.

## 5.4.1 Climate projections

Future air temperature shows a significant (p < 0.01) increase for all stations in all seasons, as well as annually, as can be seen in the statistics reported in Figure 5.4 relating to the mean temperature of all stations. For precipitation trends, winter volumes increase across the board (p < 0.01), while summer volumes decrease (p < 0.01). Although autumn precipitation volumes all decrease across the future study period, at only six of the thirteen stations is this highly significant (p < 0.01), a further two are significant at p < 0.05, and the remaining five are not significant; the average across all is significant at p < 0.05. For spring, there is combination between upward and downward trends, but none are significant.



**Figure 5.4.** Future projections (2021-2080) for seasonal and annual daily mean precipitation (left) and temperature (right). Thin coloured lines represent the 13 pseudo-meteorological stations in the catchments; thick black line denotes the average across all stations. Sen's slope statistics for the thirteen station mean also show, bold p values are significant at <0.05.

## 5.4.2 Streamflow

#### 5.4.2.1 Long-term averages

Data relating to the average of all twelve model outputs<sup>4</sup> for future streamflows display longterm trends both annually and seasonally, however, the direction and statistical significance of these varies between catchments. Using the 1990-2010 period average as a baseline, it can be seen that in most cases there is consistency between catchments (Figure 5.5). Where there are exceptions to this, these mainly relate to the Dyfi catchment, which generally displays a smaller magnitude of change, especially in the spring, autumn, and annual average. Another notable outlier is the Clwyd catchment in winter, which shows a clear decrease in the latter half of the study period, which is not observed in the other catchments. These relationships are further highlighted in the MK trend analysis results, which are summarised in Figure 5.6. A statistically significant increase in spring flows is observed in all catchments (Clwyd & Dyfi p <0.05; Conwy, Teifi & Tywi p <0.01), while a corresponding highly significant (p <0.01) decrease in autumn flows is identified in all catchments except the Dyfi (p = 0.530). Summer streamflows also decrease in all catchments, but only statistically significantly in the Conwy (p <0.05) and Dyfi (p <0.01). Winter streamflows are the most varied, with the Clwyd and Teifi decreasing, and the Conwy, Dyfi and Tywi increasing; these trends are only significant for the three most northerly catchments (p < 0.01). Finally, when looking at annual average streamflows, all catchments except the Dyfi display declining trends, albeit only significantly in the Clwyd (p < 0.01); the Dyfi increases significantly (p < 0.01).

<sup>&</sup>lt;sup>4</sup> Projected seasonal and annual average streamflow for each individual ensemble member and catchment can be found in Appendix 2, Figures A2.1 to A2.5.



**Figure 5.5.** Percentage change from the 1990-2010 baseline average (y-axis) for a 5-year moving average of 2021-2080 future streamflow data (x-axis). Seasonal and annual averages shown for each catchment. Please note that the y-axis scale on the annual average graph (bottom) is a factor of two smaller than the seasonal average graphs.

 $\land$  V Significant trend at p < 0.05

	Av	erag	e stre	amfl	ow	Legend
	Win	Spr	Sum	Aut	Ann	🔺 Kendall's Tau 0 to 0.3 $\overline{\nabla}$ Kendall's Tau <-0.6
Clwyd			$\nabla$			▲ $\triangle$ Kendall's Tau >0.3 $\overline{\nabla}$ Kendall's Tau <-0.3
Conwy			•	•	$\bigtriangledown$	$\land$ Kendall's Tau >0.6 $\checkmark$ Kendall's Tau -0.3 to 0
Dyfi				$\nabla$		$\Delta \nabla$ Non-significant trend $\blacktriangle \nabla$ Significant trend at $p < 0.05$
Teifi	$\nabla$		$\nabla$		$\bigtriangledown$	
Tywi	$\wedge$		$\nabla$		$\nabla$	

**Figure 5.6.** Overview of the direction, magnitude and significance of annual and seasonal average trends in projected future (2021-2080) average streamflow, as detected by Mann-Kendall trend analysis, based on the average of all 12 RCM model outputs.

#### 5.4.2.2 Extreme streamflows

Firstly, we can see that across all catchments, maximum 1-day flows are increasing significantly (p < 0.01) in spring, while a decrease is shown in autumn in all catchments except the Dyfi (Figure 5.7). Furthermore, the Conwy and Dyfi display a significant change (p < 0.01) in maximum 1-day flow volume in winter and summer, an increase and decrease respectively. Minimum 1-day flows decrease in summer, autumn and annually in all catchments, with this trend being statistically significant in autumn in all catchments (Figure 5.7).

	1	Max	imum	flow	v		Nº c	of hig	gh flo	w ev	ents	Mean high flow event duration					
	Win	Spr	Sum	Aut	Ann		Win	Spr	Sum	Aut	Ann	10	Win	Spr	Sum	Aut	Ann
Clwyd			Δ			Clwyd	▼		Δ		•	Clwyd	V	Δ	V	V	Δ
Conwy			-	•		Conwy			$\nabla$	•	$\land$	Conwy	Δ	Δ	$\nabla$	$\nabla$	$\nabla$
Dyfi			-			Dyfi		Δ		Δ		Dyfi	•	•	$\nabla$	$\nabla$	-
Teifi	$\nabla$		$\nabla$	-	$\nabla$	Teifi	Δ		$\nabla$	-	$\nabla$	Teifi	Δ	Δ	$\nabla$	•	$\nabla$
Tywi	Δ		Δ	-	$\nabla$	Tywi			Δ	-	Δ	Tywi	$\nabla$		$\nabla$		•
	Win	Spr	Sum	Aut	Ann		Win	Spr	Sum	Aut	Ann		Kenda	all's Ta	au O to	0.3	
		Mini	mum	flow	<u> </u>		N⁰	of lo	ow flo	w da	ays	Legend					
Clwyd		Δ	V			Clwyd		V	Δ			$\land$	Kenda	all's Ta	au >0.3	3	
Conwy	$\nabla$		•	-	-	Conwy	Δ	$\nabla$	Δ			$\triangle \Delta$	Kenda	all's Ta	au >0.6	5	
Dyfi		Δ				Dyfi		$\nabla$				$\nabla$	Kenda	all's Ta	au <-0.	6	
Teifi	$\nabla$	Δ	$\nabla$	-	$\nabla$	Teifi		$\nabla$	Δ				Kenda	all's Ta	au <-0.	3	
Tywi	Δ					Tywi	Δ	$\nabla$	Δ				Kenda	all's Ta	au -0.3	to 0	
												$\nabla \nabla$	Non-s	ignifi	cant tr	end	

**Figure 5.7.** Overview of the direction, magnitude and significance of annual and seasonal trends in projected future (2021-2080) extreme streamflows, as detected by Mann-Kendall trend analysis, based on the average of all 12 RCM model outputs.

In terms of number of extreme low flow days and high flow events, a decrease in autumn is seen for high flows, alongside an increase in the number of low flow days (Figure 5.7), suggesting a general reduction in streamflows in autumn corresponding with the average autumn flow data. Similarly, for spring in all catchments except the Dyfi, there is a statistically significant (p < 0.01) increase in the number of extreme high flow events, once again corresponding to an increase in average spring flows too. Annually, a statistically significant

(p < 0.01) increase in the number of extreme low flow days is observed in all catchments (Figure 5.7), suggesting generally drier conditions in the catchments on average.

When 1-day maximum flow volume and number of extreme high flow events are compared, the trends observed are almost identical in all catchments and seasons (Figure 5.7). In some cases, winter and spring in particular, this combination suggests an increase in both the magnitude and frequency of the largest flows. When looking at the mean high flow event duration, it can be seen that change is projected to be smaller, with a general trend towards shorter events in summer and autumn and longer events in the spring. This leads to a picture of fewer, shorter flooding events in summer and autumn across the country, with spring high flow events becoming more frequent and generally longer. Similarities can also be seen in the 1-day minimum flow volume and number of extreme low flow days; the vast majority of significant increases or decreases in minimum flows are mirrored by a decrease or increase respectively in the number of extreme low flow days. This suggests that in summer and autumn, as well as annually on average, catchments will see a greater number of low flow days, which are also more severe.

#### 5.4.2.3 Annual flow exceedance characteristics

It is clear that significant alterations are projected to take place in future streamflow in terms of both seasonal/annual averages, and extreme flows. These changes in flow regimes of the catchments are also clear when comparing the flow duration curves (FDC) of streamflow for the baseline period (1990-2010) with late-future period streamflow (2060-2080; Figure 5.8). As can be seen, in all catchments, the water volumes that are most regularly exceeded (>75% of the time) are projected to become lower. This decrease is particularly prominent in the lowest flow volumes, those exceeded more than 95% of the time; in keeping with the extreme flows MK trend analysis which also shows minimum flow levels decreasing and more extreme low flow days. High flows vary between catchments, once again in line with the extreme flows trend analysis. The Conwy and Dyfi, for example, both display an increase in discharge volumes exceeded 0 to 25% of the time in the 2060-2080 period compared to the 1990-2010 period. This once again corresponds with the observed trends of larger maximum flows, and a greater number of large flow events per year, both seen in the MK trend analysis. The Teifi & Tywi display little change in high flows between the two periods in the FDCs, which reflects

the lack of statistically significant trends seen in annual maximum flow volume and number of extreme high flow events seen in the MK trend analysis. The Clwyd displays the only statistically significant negative trend, in the MK analysis of annual maximum flow and number of extreme high flow events; this is also evident in the FDC, which shows future flow volumes being consistently lower than the 1990-2010 baseline.



**Figure 5.8.** Flow duration curves for each catchment; darker of each colour pair represents daily 1990-2010 data; lighter colour represents daily 2060-2080 data. (**A**) Conwy and Dyfi catchments, showing accentuation of both high and low flows. (**B**) Clwyd, Teifi and Tywi catchments, showing accentuation of lower flows.

## 5.4.3 Water quality

As can be seen in Figure 5.9, in the majority of cases trends seen in each water quality factor are consistent across all catchments and seasons. In terms of SS and TP concentrations specifically, 88% of all of the trends are increasing, with the Dyfi being the only catchment showing negative trends in long-term TP concentrations, in winter, spring and annually. Summer, in terms of SS concentrations, is the most spatially variable, with the Conwy and Dyfi both displaying a statistically significant decrease (p < 0.01). The Tywi is also set slightly apart from the other catchments in terms of SS, in that it has the fewest significant trends, and also shows a significant decrease in autumn SS concentrations (p < 0.01), the only catchment to do so. TN concentrations are also mostly consistent spatially, with the Dyfi once again being the main exception, with a decline in concentrations in line with those displayed for the catchment in TP. Additionally, TN levels are shown to be decreasing in winter in the Teifi also, albeit not statistically significantly. All catchments present a statistically significant (p < 0.01) increase in summer TN concentrations, while the Conwy displays the same in all seasons and annually. DO levels are generally decreasing over the study period with autumn and annual trends in particular seeing a statistically significant (p < 0.01) decrease in all catchment except the Teifi. In winter too, DO is decreasing in all catchment except the Teifi, while in spring the Clwyd, Conwy and Teifi all display an increasing trend. An increasing trend is also observed in the Dyfi for DO in summer, the opposite of that seen in the Clwyd, Teifi and Tywi.

	Sus	spen	ded s	edim	ent		т	otal	phos	ohor	us	Legen	d
	Win	Spr	Sum	Aut	Ann		Win	Spr	Sum	Aut	Ann		Kendall's Tau 0 to 0.3
Clwyd		Δ				Clwyd						$\land$	Kendall's Tau >0.3
Conwy			•	Δ		Conwy						$\triangle \triangle$	Kendall's Tau >0.6
Dyfi			•			Dyfi		$\nabla$		Δ	•	$\nabla$	Kendall's Tau <-0.6
Teifi						Teifi		Δ			$\triangle$	$\nabla$	Kendall's Tau <-0.3
Tywi	Δ		$\Delta$		$\Delta$	Tywi		$\triangle$	$\Delta$			$\mathbf{\nabla}$	Kendall's Tau -0.3 to 0
						-						$\nabla \nabla$	Non-significant trend
		Tota	l nitr	ogen	L.		D	issol	ved o	xyge	en		Significant trend at p <0.05
	Win	Spr	Sum	Aut	Ann		Win	Spr	Sum	Aut	Ann		
Clwyd	Δ	Δ		Δ		Clwyd	$\nabla$	Δ					
Conwy						Conwy	▼	Δ	Δ				
Dyfi		$\nabla$		Δ		Dyfi		$\nabla$		•			
Teifi	$\nabla$	Δ		Δ	Δ	Teifi			-	$\nabla$	Δ		
Tywi		Δ				Tywi		$\nabla$					

**Figure 5.9.** Overview of the direction, magnitude and significance of annual and seasonal average trends in projected future (2021-2080) water quality concentrations, as detected by Mann-Kendall trend analysis, based on the average of all 12 RCM model outputs.

## 5.5 Discussion

## 5.5.1 Streamflow trends

The results of the MK trend analysis show a selection of spatially varying changes across the five catchments, however, when studying summer and autumn average flows, there is a common pattern in all catchments of a general decrease. For the summer season, this trend

is only statistically significant in the Conwy and Dyfi, and in autumn, it is significant in all catchments except the Dyfi. The projected general decline in summer and autumn streamflow is also observed in other research studying mean flows, such as UK CEH (2012) and Fowler & Kilsby (2007). The only trend seen at a statistically significant level in all study catchments, an increase in spring average streamflow, is also observed in these pieces of research. Annual average flows only change significantly in the Clwyd (decrease) and Dyfi (increase), with all other catchments displaying a non-significant decline. These findings are consistent with those of Kay *et al.* (2020), who showed a decreasing trend in mean annual streamflow for four catchments in the west of England and east Wales, as we observed in all catchments except the Dyfi. The same can be said for projections made by the UK Centre for Ecology and Hydrology (CEH), who also show a small decline in mean annual streamflow for Wales up to the 2050s (UK CEH, 2012). This change is primarily driven in all catchments except the Dyfi, by declines in summer and autumn average flow being greater in magnitude than increases in winter and spring.

Arguably, in terms of the magnitude of large streamflows, the Conwy and the Dyfi catchments once again show more similarity with each other, while the Clwyd, Teifi and Tywi are all more closely connected. This can be seen for the annual data, when looking at Figure 5.8 in particular, with the Conwy and Dyfi displaying an accentuation of both the highest and lowest flows, with greater accentuation in low flows than high. The remaining three catchments only show an accentuation of low flows, this being particularly prominent in the Clwyd and Tywi. The Dyfi and Conwy are the most mountainous, steeply sloping, and smallest in area of the catchments, which makes them highly responsive to precipitation, and flashy. It is these specific catchment characteristics that partly foster the observed statistically significant trends of increasing winter and annual maximum streamflow, as well as number of high flow events. Such physical catchment characteristics are known to cause high responsiveness, with studies such as Dunne (1978), Weiler & McDonnell (2006) and Hornberger et al. (2014) linking smaller catchment area and greater steepness to a quicker catchment response and larger streamflow events. These responses are further exacerbated by the soil characteristics of the catchments, with the Conwy and the Dyfi in particular being dominated in their upper reaches by peaty and gleysol soils, with the majority of this also lacking underlying groundwater or aquifer storage (Boorman et al., 1995; European Commission, 2004). This once again causes

greater overland and subsurface flow, rather than infiltration and storage, leading to a flashier catchment regime (Boorman *et al.*, 1995).

It is notable however, when comparing the Conwy and the Dyfi, that accentuation of future low flows is much greater for the Conwy than the Dyfi (Figure 5.8), with this likely due to the Dyfi displaying the opposite trend for winter minimum flow and number of low flow days, than the Conwy. In fact, the Dyfi is the only catchment to show a significant reduction in the number of low flow days in winter, as well as an increase in winter minimum flow volume. Such changes can be attributed to LULC characteristics, in this case the larger proportion of coniferous and mixed forestry present in the Dyfi. A study by Robinson et al. (2003) demonstrated that for western Europe, forest areas, in particularly those that are coniferousdominated, help to maintain higher baseflow conditions. This relationship also aids in explaining the non-significant increase seen in the winter low flows for the Tywi, which has the second largest proportion by some margin of LULC designated as coniferous and mixed forestry. Furthermore, the statistically significant increase in annual average flows in the Dyfi is driven by increases seen in winter and spring streamflows, outweighing decreases in summer and autumn. This difference from the other catchments is also likely due to the large proportion of forested land in the Dyfi; Teutschbein et al. (2015), for example, showed that increased forest cover led to increased annual average flow in their study of sixteen small catchments in Sweden.

Annual low flows in all catchments show an accentuation of presently seen low flows, into the 2060-2080 period (Figure 5.8); this is likely due to the statistically significant declines seen in summer and autumn average flows into the future (Figure 5.6). These seasons traditionally have seen the lowest flows, so the fact that they are generally getting drier and warmer, would explain this decline in minimum flow volume. This trend observed in low flows across the catchments is in line with trends described in Kay *et al.* (2020), who showed for ten catchments in the UK, under RCP8.5 conditions, low flows are declining in volume. This is particularly seen in the more westerly of the catchments studied (Kay *et al.*, 2020), those closest to this study's catchments.

Our results correspond with other research conducted recently in the UK in terms of seasonal changes in high and low flows, especially when considering the decrease in low summer and

autumn flows, and the increase in high winter and spring flows. Visser-Quinn *et al.* (2019), for example, detailed an increase in the magnitude, duration and frequency of both high and low flow events for Wales in the far future (2090s) under RCP8.5. Our results also show an increase in the frequency and magnitude of such events, however, in terms of event duration we find the opposite trend, a decline in most seasons as well as annually. Our results also show an increasing number of extreme high winter flow events, and extreme low autumn flows specifically, also corresponding with the results of Collet *et al.* (2018). When studying the UK and Wales as part of wider studies, Marx *et al.* (2018) also showed a decline in mean annual low flows in Atlantic influenced western Europe, albeit the smallest change of all regions across Europe. Additionally, Thober *et al.* (2018) found an increase in high flows and flooding in the UK under future climate scenarios representing RCPs 2.6, 6.0 and 8.5.

#### 5.5.2 Water quality trends

Unlike the streamflow trends discussed, the results of the MK trend analysis of water quality factors is much more homogenous across the study region. The exceptions to this are the Dyfi catchment in terms of TP and TN in particular, and the Teifi in terms of DO; these outliers can be explained by studying the catchment characteristics. The Dyfi catchment, for example, has the lowest proportion of land designated as agricultural of all of the catchments, and the largest proportion designated as both forest and scrub (Table 5.1). These characteristics give the Dyfi catchment a significantly different LULC make-up than the Clwyd, Teifi and Tywi in particular. Agricultural land is well documented to be one of the highest contributors to TP and TN levels in river systems (Edwards et al., 2000; Burt et al., 2011; Mockler et al., 2017; Weigelhofer et al., 2018). Burt et al. (2011) stated that on average in England and Wales, 60% of nitrogen in surface waters originates from agricultural land, but this can be as high as 90% in agriculturally-dominated catchments. Similarly, for phosphorus, the House of Commons Environmental Audit Committee (2018) reported that run-off from agricultural land was one of the main sources of TP in surface waters, especially at times of highest water levels, i.e. winter and spring in this study's catchments. These relationships, and the significantly lower proportion of agricultural land in the Dyfi catchment, go some way to explain this marked departure from the trends seen in all other study catchments. This LULC with TP and TN relationship also explains the trends seen in all catchments except the Dyfi, for example increases in winter TP in the more agriculturally-dominated catchments. Furthermore, the

increases seen in summer TP and TN in all catchments are also indicative of the dominance of in-stream processes on their generation during warmer conditions (Edwards *et al.*, 2000). Moreover, given the dependence of TP and TN concentrations on streamflow, as aforementioned and demonstrated in Table 5.5, it can also be observed that the declines in winter and spring TP and TN in the Dyfi relate to the unique streamflow trends observed in this catchment. Specifically, the Dyfi displays the only statistically significant increase in low volumes in winter, as explained previously. Additionally, decreases in TP and TN observed annually relate to the only statistically significant increase flows displayed in any of the five catchments. These correlations also help to explain the departure from the generally congruous trends in streamflow and water quality displayed between the Conwy and Dyfi, with the only major differences observed in their streamflow responses being in annual average and winter low flows.

The SS concentration results show a consistent rise across almost all catchments and seasons. The general uplift is likely due to a variety of reasons depending on the season. Certainly, in the spring and winter, increasing trends observed in the number and magnitude of high flow events is likely the cause, along with an increase generally in average flows. These larger flows not only increase debris washed into river systems by overland flow, but also increase bank and bed erosion, as well as bed transport (Zeiger & Hubbart, 2016; Janes et al., 2017; Vercruysse & Grabowski, 2019), causing an increase in total SS concentrations. In summer and autumn, with temperatures increasing and precipitation decreasing, there is an increased likelihood of drought, as well as fewer rainy days. This means that when rain does arrive, there is much more dust and debris to be washed from the land surface into river systems, especially as dry land also has lower infiltration rates (Clark et al., 2017; Pulley & Collins, 2019). Combined with the fact that lower streamflows have less dilution potential, this is the likely cause of the SS increases seen in these seasons (Whitehead et al., 2009b). Negative trend shown in summer Conwy and Dyfi SS concentrations, are likely caused by statistically significant declines in maximum flow (both *p* <0.01) and average flow (Conwy *p* <0.05; Dyfi *p* <0.01) volumes during the season across the study period, not seen in the other catchments

The observed downward trends in DO seen for most seasons in the majority of catchments are likely linked to the rise in air temperatures seen in all seasons and annually (Figure 5.4). Raised air temperatures have been shown to correlate positively with average water

temperatures, albeit this relationship can be masked or affected by factors such as discharge volume, upstream dams and point source inputs (Webb *et al.*, 2003; Caissie, 2006; Hannah & Garner, 2015; Beaufort *et al.*, 2020; Daniels & Danner, 2020). The known negative relationship between stream water temperature and DO concentration due to the decreased carrying capacity of warmer water (Ravansalar *et al.*, 2016; Chaturvedi & Misra, 2020), is therefore likely to be the cause of the declines seen in the study catchments. Furthermore, the broadly increasing concentrations of TP and TN shown in all catchments will likely also have an impact, causing an increase in aquatic plant growth during warmer temperatures, reducing oxygen levels in waters.

When studying water quality results as a collective, a clear decrease in water quality across all catchments and seasons can be observed. There is mostly agreement in terms of trend direction, in spring and autumn, however, in terms of statistically significant trends, more of these are present in summer, winter and annually. As previously mentioned, the key outlier is the Dyfi catchment, mainly due to its unique LULC characteristics compared to the other catchments, however, here too there are still significant water quality challenges, especially in the autumn.

#### 5.5.3 Study implications

This research has been undertaken to provide a backdrop on which future planning, adaptation, and mitigation measures can be based for various industries. It is some of the first to make use of the new UKCP18 future climate projections and is certainly the first to investigate the impact of the projections for streamflow and water quality in Wales. It is hoped that by providing future projections of average and extreme high and low flows, as well as water quality changes, for a large area of Wales, that the results will be applicable to stakeholders in a variety of industries.

This study has shown how, under a worst-case scenario of future emissions, key river systems in Wales will be impacted by climate change in various ways in terms of both water quantity and quality. It is clear that the effects of climate change on streamflows will be most keenly felt seasonally, with annual average flows staying relatively stable in comparison. It is likely that seasonal streamflow patterns currently seen will be exacerbated in the future, with modelling results showing increases in winter and spring average flows and decline in summer

and autumn flows. Trends in extreme events will also intensify, with large streamflow events (floods) generally increasing in magnitude and frequency in winter and spring, and low flows (droughts) increasing in frequency and magnitude in summer and autumn in particular. These streamflow trends could have important implications in terms of river ecology, flooding, droughts, and for a variety of stakeholders, not least of all water service providers (WSPs). Given that ~95% of water abstracted for public water supply in Wales is taken from surface sources (DCWW, 2019b), the trends seen could impact on the ability of WSPs to continue to meet demand, especially in summer months, during peak demand (Dallison et al., 2020). Although annual average flows are relatively stable, if declining slightly in most catchments, it is the fact that there will be considerably less water in summer and autumn that could prove problematic from ecological and economic perspectives. Additionally, any increases in winter streamflow will do little to combat summer shortages if there is insufficient capacity to store it for use later in the year; this is particularly true for the Dyfi and Teifi, which are more reliant on river abstraction, compared to impoundment reservoirs used in the other catchments. Furthermore, the quality of incoming water could also cause problems at drinking water treatment plants, with SS, TP, and TN in particular potentially requiring plants to provide more treatment, negatively impacting on energy use, operation and maintenance costs, chemical use, and potentially plant capacity.

The changes seen in annual flow exceedance probabilities could have implications for all large water abstractors, with abstraction licences often stipulating conditions, for example minimum 'compensation flow' volumes. With lower flows expected to occur for a greater proportion of the time annually in the future, this could impact on the amount of water available for users such as industry, agriculture, and hydroelectric power generation. UK hydroelectric power generation schemes, both large and small, warrant the attention of future work in particular, due to the large potential impacts of changes in streamflow, and low flows especially. The timing, season length and total generation capacity of hydroelectric generation schemes could be impacted badly by changes in streamflow, potentially affecting the future electricity grid-mix in the UK.

Finally, the changes seen in water quality factors could have an array of impacts on the aquatic ecosystem supported by surface waters. Increases in TP and TN, for example could lead to increases in eutrophication in lakes and rivers. Additionally, the declining DO levels shown

could seriously impact on fish species within catchments, with lower levels being well documented to decrease fish survival rates and badly impact spawning and development.

## 5.6 Conclusions

This research has presented the key changes in future seasonal and annual stream quantity and quality parameters in Wales. The results highlight the need for adaptation now, in order to prepare infrastructure and systems to combat changes, this is especially true for water supply, despite Wales being seen as a water rich region. Planning and mitigating now will help to ensure the continued supply and clean drinking water into the future, especially in the summer and autumn when lower flows and water quality will challenge WSPs.

Finally, we suggest a variety of areas for future research focus. Firstly, adaptation and mitigation measures required for all large abstractors impacted by future changes in river water quantity and quality need to be investigated, to ensure a robust defence against climate-induced changes. Relating to this, further hydrological modelling, using more optimistic projections for future emissions, such as RCPs 2.6, 4.5 and 6.0 would be useful, to enable a broader range of potential future impacts to be compiled, and more informed decisions to be made. This is particularly relevant to WSPs, to enable a continued high-quality future water supply, and to plan the future provision of drinking water treatment plants. Furthermore, investigation of the impact of streamflow changes on the total output generation potential of hydroelectric systems is needed at a national level, to enable robust planning across the country, to ensure stability in the national electricity grid. Lastly, research on climate change links to total water demand is also required, in order to ensure future water supply-demand budgets are achievable under future conditions, and to put in place management measures where necessary.

**Acknowledgements:** This study was conducted as part of the Dŵr Uisce project, which aims at improving the long-term sustainability of water supply, treatment and end-use in Ireland and Wales. The project has been supported by the European Regional Development Fund (ERDF) through the Interreg Ireland-Wales Co-operation Programme 2014-2020. The work has been used as a case study for Dŵr Uisce Work Package 7 "climate change", and forms part of a study assessing the impact of climate change on water resource exploitation in Wales.
# **CHAPTER 6: STUDY III**

# Impacts of climate change on future water availability for hydropower and public water supply in Wales, UK

R.J.H. Dallison, S.D. Patil and A.P. Williams School of Natural Sciences, Bangor University

This chapter has been submitted in full for publication in the *Journal of Hydrology: Regional Studies* and was undergoing peer-review at the time of submission of this thesis.

## Author contributions

Richard Dallison: conceptualisation, methodology, software, formal analysis, investigation, data curation, writing – original draft, writing – review & editing, visualisation
Sopan Patil: conceptualisation, writing – review & editing, supervision
Prysor Williams: writing – review & editing, supervision

## 6.1 Abstract

Climate change is predicted to have a large impact on the hydrological regimes of Welsh rivers. However, its influence on the abstraction capability of key sectors, such as public water supply (PWS) and hydroelectric power (HEP), is not yet fully understood. We use the Soil and Water Assessment Tool (SWAT) to generate future (2021-2079) streamflows under a worstcase scenario of greenhouse gas emissions (Representative Concentration Pathway 8.5) at two catchments in Wales, the Conwy and Tywi. SWAT streamflow output is used to estimate total unmet demand for PWS and changes in generation characteristics for HEP. PWS unmet demand is assessed using the Water Evaluation And Planning (WEAP) system under increasing, static, and declining demand scenarios. Mann-Kendall analysis is performed to detect and characterise trends. Under all future demand scenarios, there is increased occurrence of insufficient streamflow to satisfy PWS demand. For HEP, a decrease in annual abstraction volume results in a loss of generation potential, despite an increasing number of days that maximum abstraction is reached. Changes in HEP generation and PWS availability are most pronounced in the medium-term (2021-2054), with rate of change slowing after 2060. We provide a novel perspective on future water resource availability in Wales, giving context to management planning to ensure future PWS sustainability and HEP generation efficiency.

**Keywords:** Hydroclimatic change; Hydroelectric power; Hydrological modelling; Trend analysis; Water resource management; Water supply-demand balance

#### 6.2 Introduction

Climate change is a key driver of hydrological regime alteration globally, with prevailing weather conditions being inextricably linked with streamflow. Given the dependence of society on river systems for everyday life, it is of great importance to understand how climate change will impact water resource availability. In the UK, it is widely accepted that climate change will lead to hotter, drier summers and warmer, wetter winters, as well as an increase in the magnitude, frequency and duration of extreme weather events (ASC, 2016a; Lowe et al., 2018). In terms of streamflow, an exacerbation of low and high flows is expected, as well as a reduction in average summer and autumn flows and an increase in winter and spring average flows; these opposing trends should leave annual averages relatively stable (Christierson et al., 2012; Prudhomme et al., 2012; Watts et al., 2015a; Kay et al., 2020). Climate change and streamflow responses in Wales specifically are projected to be in line with the trends suggested for the rest of the UK, with summer mean and maximum temperatures, as well as winter total and extreme precipitation volumes, predicted to increase (ASC, 2016b; Kay *et al.*, 2020). These changes have the potential to cause a large impact on major water consuming sectors such as agriculture, heavy industries, public water supply (PWS), and hydroelectric power (HEP) generation. The Adaptation Sub-Committee (ASC) of the Committee on Climate Change, for example, suggests that PWS and agricultural water demand for irrigation could be two of the most pressurised sectors in the UK under future climate and streamflow changes, with demand potentially outstripping available water resources (ASC, 2016a). Furthermore, Arnell & Delaney (2006) detailed how, even at that time of publication, PWS companies in England and Wales would need to adapt to ensure sufficient raw water resources due to changes to low flow regimes. In Wales, Carless & Whitehead (2013) demonstrate decreasing summer and autumn streamflows with corresponding increases in winter and spring. In the given context of a theoretical 99 kW micro-HEP installation, this causes an accentuation of seasonality in terms of energy generation, with extra supply in wetter seasons compensating for lower supply in drier seasons, leading to an annually stable situation (Carless & Whitehead, 2013). However, these changes are likely to be catchment-specific, depending on the individual catchment's topographic and land cover characteristics as well as the prevailing weather patterns.

In 2018, renewable energy in Wales accounted for 25% of the total electricity generated, with HEP making up 5.2% of this, and therefore 1.3% of overall Welsh electricity generation (Welsh Government, 2019b). A large increase in the number of small-scale HEP schemes has occurred in the last eight years due to financial incentives in the form of Feed-in-Tariffs (FiT; Welsh Government, 2019), and also given that most opportunities for large-scale schemes have been exhausted (Carless & Whitehead, 2013). However, due to the curtailment of the FiT scheme in 2019, the number of new developments has slowed, with further schemes only likely to be commissioned where conditions are optimal (Welsh Government, 2019b). These types of schemes, usually run-of-river in design, generally have little or no water impoundment, making them particularly vulnerable to changes in hydrological conditions. Run-of-river HEP schemes are usually designed and optimised on the basis of historical flow, with flow duration curves used to set abstraction conditions. While this method is a good starting point for designing the most efficient system to maximise power output, it neglects to account for future streamflows. Climate change has the potential to cause major alteration to the hydrological characteristics of river systems, modifying the timing and quantity of available water. High and low flows are particularly likely to be affected (Sayers et al., 2015; Watts et al., 2015a; ASC, 2016b; Kay et al., 2020), and these extreme flows are often important in terms of HEP scheme design. In 2017, the Welsh Government set a target for 70% of electricity consumed in Wales to be generated by renewables by 2030 (Welsh Government, 2019b). HEP plays a small but important role in reaching this target, especially in the winter and spring seasons when electricity generation from other sources, such as solar PV, is lower. Therefore, understanding the nature of change in the abstractable flow for HEP schemes under future climate change is crucial. This will allow for more robust planning of the future energy mix in Wales as well as maximisation of resource use efficiency and electricity generation.

PWS also relies heavily on surface waters in Wales, such as rivers and lakes, with over 95% of the supplied water originating from these sources in the country (DCWW, 2019b). In Wales, PWS is largely under the authority of Dŵr Cymru Welsh Water (DCWW), who provide an average of 800 million litres of water per day to over three million people (DCWW, 2019a). While the majority of surface water used is from lakes and reservoirs, a substantial proportion is taken from lowland river abstractions, especially for rivers in south Wales, such as the Wye,

Usk and Tywi, which are often supported by upstream reservoirs in times of low flow (DCWW, 2019a). While the water supply system in Wales is largely resilient to climate change, due to the amount of water stored, changes in climate still do have the ability to change the required water management and places greater pressure on the system (ASC, 2016b). The ASC report a 12% supply-demand surplus for PWS in Wales, however, this is set to decline under projected population growth and climate change, with three of DCWW's water resource zones expected to be in deficit by the 2080s (ASC, 2016b; DCWW, 2019b).

Projected increases in the occurrence of low flows will impact how often PWS and HEP operators are permitted to abstract water from rivers in the future, with stringent hands-off-flow (HoF) regulations applied to abstractions to ensure the protection of low flows downstream. In addition, a greater occurrence of large flow events will potentially be of little use to abstractors to compensate for less abstraction potential due to low flows, if system capabilities, such as turbine size for HEP, do not allow for it. It is therefore important to study these future changes now, to successfully plan for systems that are resilient to climate change and make the most of available future flows.

In this research, we study the impact of climate change on two catchments in Wales, UK, Conwy and Tywi, especially in terms of future water availability for small-scale HEP generation and PWS. This is a less-studied region of the UK in this regard, due to the perceived abundance of water resources in Wales. However, owing to both the reliance on surface waters for PWS and the large recent increase in small-scale HEP projects, this is an important area in which to understand future water resource changes.

# 6.3 Data and methods

## 6.3.1 Study catchments

Our two study catchments, Conwy and Tywi (Figure 6.1), were selected due to their contrasting physical characteristics, especially in terms of catchment land use/land cover (LULC), mean catchment slope, and catchment area (Table 6.1); all potentially giving rise to differences in hydrological regime. These catchments are also exploited for their water resources in the form of HEP and PWS, with Natural Resources Wales (NRW) providing licenses for sixteen HEP abstractions for non-impoundment run-of-river schemes in Conwy,

and nine in Tywi (NRW, 2019). Both catchments are also used for PWS, however, only the Tywi catchment has an abstraction taken directly from the river, with PWS in Conwy sourced from reservoirs only (DCWW, 2019a). The downstream river-based abstraction in Tywi is supported in times of low flow by upstream releases from the Llyn Brianne reservoir (DCWW, 2019b). This abstraction supplies the largest drinking water treatment plant in DCWW's network, serving a population of ~400,000 in Swansea, Neath, Bridgend and the Vale of Glamorgan (DCWW, 2019a), making the abstraction an important location to study in terms of future water supply-demand balance.



**Figure 6.1.** Study catchments, displaying CORINE Land Cover classification (EEA, 2012). Inset, location of catchments within Wales, and Wales within the UK.

The Conwy catchment is more mountainous in topography, with a steeper mean slope, and greater maximum elevation, peaking 261 m higher than the Tywi, at 1062 m, near the summit of Snowdon. The Tywi catchment, however, has more than double the drainage area of Conwy, at 1365 km<sup>2</sup> to Conwy's 541.8 km<sup>2</sup> (Table 6.1). The catchments are also contrasting in terms of LULC (Figure 6.1), with the Tywi being dominated by agriculture, pasture specifically (63.9%), which is well-distributed throughout all but the highest elevations in the catchment.

The Conwy catchment, on the other hand, has a larger proportion of scrubland (42.1%), in particular moors/heathland (23.6%) and natural grassland (16.6%), mostly to the west of the main channel, with the eastern side being more pasture-dominated (Figure 6.1). The Conwy also features a large peat bog in the south of the catchment, accounting for 8.4% of total catchment LULC; this is significantly more than is seen in the Tywi (Table 6.1). Forests cover a slightly larger area of the Tywi (16.0%) than the Conwy (13.7%), but the mix of forestry types is different, with notably more coniferous forest in the Tywi (10.1%), while the Conwy has equal proportions of coniferous and mixed forest (5.5%; Table 6.1).

**Table 6.1.** Key study catchments details. Catchment area, elevation, and slope data derived from 5 m resolution OS Terrain 5 DEM from Ordnance Survey; land use/land cover data derived from 2012 CORINE Land Cover data (EEA, 2012).

		Conwy	Tywi
Catchme	ent area (km²)	541.8	1364.6
Maximu	m catchment elevation (m)	1062	801
Mean ca	tchment slope (%)	19.7	16.6
	Urban	0.7	0.7
	Agriculture	30.6	64.6
r (%	Arable	0.1	0.7
ove	Pasture	30.5	63.9
d d	Forest	13.7	16.0
lan	Broadleaf	2.7	1.0
se/	Coniferous	5.5	10.1
np	Mixed	5.5	4.9
lan	Scrub	42.1	17.1
ent	Natural grassland	16.6	8.4
hm	Moors and heathland	23.6	6.5
atc	Transitional woodland scrub	1.9	2.2
0	Peat bog	8.4	1.3
	Sparsely vegetated areas	4.1	0.0

## 6.3.2 Future streamflow and climate projections

Future streamflows for both catchments were modelled using the Soil and Water Assessment Tool (SWAT) at a daily time-step (Arnold *et al.*, 1998, 2012; Neitsch *et al.*, 2011) from 2021 to 2079. SWAT has been used extensively in the context of future hydrological regime change and water resources assessment globally (Coffey *et al.*, 2016; Perra *et al.*, 2018; Sultana & Choi, 2018; Yuan *et al.*, 2019; Khan *et al.*, 2020), with publications focussing on both HEP (Park & Kim, 2014; Shrestha *et al.*, 2016; Haguma *et al.*, 2017; Abera *et al.*, 2018) and water availability (Sharma *et al.*, 2016; Veettil & Mishra, 2016; Bessa Santos *et al.*, 2019; Rivas-Tabares *et al.*, 2019).

Input data for SWAT were obtained from Ordnance Survey (5 m resolution OS Terrain 5 DEM; Ordnance Survey, 2020), the EU soil database (European Commission, 2004) and the CORINE Land Cover dataset (Copernicus Land Monitoring Data; EEA, 2012). Additionally, historical daily weather data (air temperature and precipitation) and streamflow data, both used for model calibration and validation, was sourced from the Centre for Ecology and Hydrology's (CEH) Climate, Hydrology and Ecology research Support System (CHESS) dataset (Robinson *et al.*, 2017) and the National River Flow Archive dataset (NRFA, 2020), respectively. Streamflow data was taken from gauges with as close to natural flow as possible, station 66011 for Conwy, and 60006 for Tywi. The catchments were calibrated individually with the streamflow data, for a 14-year period of 1985-1998. We used the Particle Swarm Optimisation method (Kennedy & Eberhart, 1995) for model calibration, with the Kling-Gupta efficiency (KGE) as the goodness-of-fit metric (Gupta *et al.*, 2009). The calibration KGE values were 0.770 for Conwy and 0.841 for Tywi. Following the calibration of model parameters, we validated the model for the 1999-2014 period and obtained the KGE values of 0.718 for Conwy and 0.717 for Tywi.

Future streamflows were projected based on the weather input derived from the UK Climate Projections 2018 dataset, 'Regional Projections on a 12 km grid over the UK for 1980-2080' (MOHC, 2018b). A worst-case scenario approach was taken, using an ensemble of twelve regional climate models downscaled from the 60 km HadGEM3-GC3.05 global coupled model (Murphy *et al.*, 2018), under Representative Concentration Pathway 8.5 (RCP8.5) conditions. RCP8.5 represents a scenario of high population growth, continued coal use, and no downturn in global greenhouse gas emissions. This approach allows planners the ability to take account of the most extreme adaptation/mitigation potentially need. A summary of future temperature, precipitation and streamflow is shown in Figure 6.2. Temperatures are projected to increase in all seasons, while precipitation and streamflow both increase in the winter and spring, but decrease in the summer and autumn, leading to a small overall annual reduction.



**Figure 6.2.** Projections for seasonal and annual mean temperature, precipitation and streamflow for Conwy (left) and Tywi (right). Bottom panels shows percentage change from 1990-2010 baseline average for a 5-year moving average of seasonal and annual mean streamflow, at the gauging station locations identified in Figure 6.3.

## 6.3.3 Hydroelectric power assessment

Assessment of changes in the abstraction regime for HEP locations was undertaken for nonimpoundment, run-of-river based schemes only, of which there are a total of sixteen in Conwy, and nine in Tywi. Abstraction locations were obtained from NRW. Due to lack of specific information pertaining to each HEP scheme, such as abstraction licence conditions, scheme layout, generation capacity etc., abstraction conditions were based on general licensing guidelines laid out by NRW (NRW, 2020b). Guidance from the organisation states that for schemes creating a depleted reach (i.e. run-of-river schemes) and that do not operate on rivers supporting salmon spawning or protected species, there are two types of abstraction permitted, Zone 2 (Z2) and Zone 3 (Z3; NRW, 2020). It was assumed for the purposes of this study that all schemes operate under Z2 or Z3; a summary of the conditions placed on abstraction rates of these zones is made in Table 6.2. Schemes are categorised into either zone based on the gradient of the depleted reach, with those below 10% gradient being Z2, and those above being Z3. As information on the actual depleted reach was not available, the average slope of the sub-basin immediately downstream of the abstraction location was taken, and the corresponding zone type applied (Figure 6.3). Under the assumption that each HEP site abstracts the maximum amount of flow available to it, and given daily average streamflow (Q), Equations 6.1 and 6.2 were used to calculate daily average abstraction volume ( $A_{daily}$ ) at each location in line with abstraction rates dependent on the site zone:

$$Q_{surplus} = Q - HoF \tag{Eq. 6.1}$$

where *HoF* represents the compensation hands-off-flow release required to protect low flows and  $Q_{surplus}$  is the amount of water available for abstraction, used to calculate  $A_{daily}$ :

$$A_{daily} = Q_{surplus} \times Q_{take} \begin{cases} 0, & \text{if } A_{daily} < A_{start} \\ A_{max}, & \text{if } A_{daily} > A_{max} \\ A_{daily}, & \text{if } A_{start} < A_{daily} > A_{max} \end{cases}$$
(Eq. 6.2)

where  $Q_{take}$  is the proportion of flow available for abstraction as per the zone conditions,  $A_{start}$  refers to the minimum abstraction volume required to start, and for efficient operation of, the turbine, and  $A_{max}$  represents the maximum permitted abstraction volume. A further assumption was that an impulse turbine is in use at each site, as is common with small-scale HEP schemes in upper catchment reaches, such as those analysed in this study (Lilienthal *et al.*, 2004; Cobb & Sharp, 2013; Židonis *et al.*, 2015). Impulse turbines have largely high and stable efficiency after approximately 10% of designed flow is achieved (Paish, 2002; Novara & McNabola, 2018; Chitrakar *et al.*, 2020), making them ideal for settings with variable  $A_{daily}$  (Cobb & Sharp, 2013). For this reason,  $A_{start}$  was set at 10% of  $A_{max}$  for each scheme, which is the assumed design flow volume.

Condition	Definition	Zone 2	Zone 3
Depleted reach gradient	Gradient of the stream between abstraction point and return flow	<10%	>10%
Hands-off-flow (HoF)	Streamflow rate below which abstraction is not permitted	Q <sub>95</sub> <sup>a</sup>	Q <sub>95</sub> <sup>a</sup>
Maximum abstraction volume (A <sub>max</sub> )	Maximum rate of abstraction, above which no additional flow can be taken	1.3 x Q <sub>mean</sub> <sup>b</sup>	Q <sub>mean</sub> <sup>b</sup>
Percentage take (Q <sub>take</sub> )	Proportion of flow between HoF and A <sub>max</sub> permitted for abstraction	50%	70%

Table 6.2. Abstraction conditions as defined by NRW guidelines for Zone 2 and Zone 3 sites (NRW, 2020b).

<sup>a</sup> Streamflow volume exceeded 95% of the time

<sup>b</sup> Mean annual streamflow volume



**Figure 6.3.** Studied abstraction locations for hydroelectric power (HEP) and public water supply (PWS); HEP locations categorised by abstraction regime type. The '>' and '<' markers denote the river section of the PWS abstraction, as to identify the specific location would violate data licence conditions. River network, with stream orders defined by Strahler method, and catchment elevation, based on 5 m resolution OS Terrain 5 DEM from Ordnance Survey, also shown.

#### 6.3.4 Public water supply assessment

Assessment of the impact of future hydrological regime change on PWS was undertaken in the Tywi catchment only, due to the lack of river-based PWS abstraction in the Conwy catchment. The supply-demand balance at the aforementioned single major abstraction location (Figure 6.3) was calculated using the Water Evaluation And Planning (WEAP) system (Raskin *et al.*, 1992; Yates *et al.*, 2005). WEAP is an integrated water resource management model that assimilates the demand and infrastructure management with physical hydrological processes and allows for multiple scenario analysis and comparison (Raskin *et al.*, 1992; Yates *et al.*, 2005). The WEAP model has been used extensively for scenario analysis related to water resource planning and allocation in a variety of contexts, such as the impacts of climate and land use change (Purkey *et al.*, 2008; Joyce *et al.*, 2011; Ashofteh *et al.*, 2013; Esteve *et al.*, 2015; Tena *et al.*, 2019), reservoir and dam operation planning (McCartney & Menker Girma, 2012; Demertzi *et al.*, 2014; Vonk *et al.*, 2014; Azari *et al.*, 2018), ecosystem requirements and environmental protection (Thompson *et al.*, 2012; Fatemi *et al.*, 2013; Flores-López *et al.*, 2016; Momblanch *et al.*, 2020), and population increase and urbanisation impacts (Höllermann *et al.*, 2010; Kumar *et al.*, 2017; Toure *et al.*, 2017; Alamanos *et al.*, 2020). However, WEAP is not designed for detailed optimisation studies, such as that required for HEP (Yates *et al.*, 2005). For this reason, the model has only been used in the PWS analysis for this study.

Future total daily water demand was calculated for three scenarios (Figure 6.4), representing increased, static, and declining abstraction requirement. The increased demand scenario was based on the linear relationship between historical daily temperature and total water abstraction from the location, as presented by Dallison *et al.* (2020). The water abstraction data was provided by DCWW for January 2012 to December 2016 period. The static demand scenario was based on the same 2012-2016 data, with a mean for each day being taken across the 5-year dataset; this year of mean values was then applied every year for the future period. The decreased demand scenario used the same starting base as the static scenario, with demand decreasing linearly by 20% across the period. This decrease is in line with that projected in DCWW's latest water resource management plan (DCWW, 2019a), based on an extensive leakage reduction programme and decreasing domestic water usage, despite a projected increase in the population served. A compensatory HoF was also implemented in WEAP, set at 681.91 million litres per day as laid out in the abstraction license for this location (SWWRA, 1965). Total daily unmet demand ( $D_{unmet}$ ) was then calculated under each scenario using Equation 6.3:

$$D_{unmet} = Q - HoF - D \tag{Eq. 6.3}$$

where D is daily total water demand; Q and HoF are defined as Equation 6.1. Due to the aforementioned system of reservoir low flow support by Llyn Brianne,  $D_{unmet}$  is therefore also assumed to be equal to the total daily required reservoir release.



**Figure 6.4.** Future monthly total water demand at the Tywi public water supply abstraction under the three future water demand scenarios: increased, static, and decreased. Thick dashed lines represent the linear trend of each scenario.

#### 6.3.5 Trend analysis

Trends in the output data from the HEP and PWS calculations were analysed using Mann-Kendall (MK) trend analysis (Mann, 1945; Kendall, 1975), in line with other studies that have analysed data relating to hydrological regime change (Murphy et al., 2013; Mwangi et al., 2016; Mudbhatkal & Amai, 2018; Dallison et al., 2020; Jin et al., 2020). The nonparametric test was deemed suitable due to the nature of hydrological data, which is non-normally distributed and exhibits seasonality. The Hamed & Rao method of auto-correlation correction (Hamed & Rao, 1998) was applied, along with Sen's slope estimator (Sen, 1968), to evaluate the direction and size of detected trends. Trends for all factors were analysed based on seasonal (winter, December to February; spring, March to May; summer, June to August; autumn, September to November) and annual averages (hydrological years), or totals, dependent on the factor. HEP trends for each catchment were analysed separately for the average of all Z2 and all Z3 abstractions. The factors analysed were: (1) number of days where A<sub>daily</sub> is greater than A<sub>start</sub>, i.e. number of days generation is possible, (2) number of days A<sub>max</sub> reached, (3) mean A<sub>daily</sub> on days generation possible, and (4) total seasonal/annual abstraction. For PWS, total unmet demand, number of days demand unmet, and mean unmet demand, were all analysed under each demand scenario. For both HEP and PWS, MK analysis

was applied to the full 2021-2079 (long-term) period, as well as to the end of a medium-term period, 2021-2054. This approach is useful for HEP, as the near future analysis is more in line with the life span of recently installed, and soon to be installed small-scale systems (Hatata *et al.*, 2019; Killingtveit, 2019). For PWS, the near future analysis is a similar period as is currently being planned for, 2050 being the end of the planning period for recently published water resource management plans (e.g. DCWW, 2019). For Wales specifically, DCWW also recently published a vision document to 2050 (DCWW, 2018). HEP and PWS baseline (2021-2030), near future (2045-2054) and far future (2070-2079) decadal averages were also taken seasonally and annually for the same factors as the MK analysis, to enable further visualisation of a potential medium- and long-term planning needs for both industries.

## 6.4 Results

#### 6.4.1 Hydroelectric power

With regards to change in the number of days annually when Astart is achievable, all abstraction locations display a decrease in both the medium-term (2021-2054) and long-term (2021-2079), although these trends are statistically significant for more HEP abstractions in the medium-term analysis than in the long-term (Figure 6.5). Trends in the number of days when A<sub>max</sub> is reached are more variable, especially in the long-term, with differences seen both within and between the two catchments. The Conwy in this period, for example, exhibits statistically significant increases for all abstractions in the west of the catchment, while those on the eastern side vary, with a statistically significant decrease being present for one location (Figure 6.6). In the medium-term, there is more agreement between abstraction locations, with all but one Z3 abstraction in each catchment seeing an increase in the number of days when maximum abstraction is achieved. The combination of these two broad trends (fewer days of abstraction, but larger abstraction volumes available) causes mean A<sub>daily</sub> on days when abstraction is possible, to increase for the vast majority of locations in both time periods, but especially so in the medium-term (Figure 6.7). However, when observing the change in total volume abstracted per year, a decrease is displayed in all locations, for both time periods (Figure 6.8). Notably in the medium-term trends for total abstraction, all Z3 abstractions had statistically significant (p < 0.05) decreases in volume, while all Z2 locations decline without statistical significance (Figure 6.8), the only factor which showed such a split.



**Figure 6.5.** Overview of the direction, magnitude and significance of annual trends in number of days A<sub>start</sub> achieved for the periods 2021-2079 (top) and 2021-2054 (bottom). Trends detected by Mann-Kendall trend analysis for individual hydroelectric power abstraction locations.



**Figure 6.6.** Overview of the direction, magnitude and significance of annual trends in number of days A<sub>max</sub> reached for the periods 2021-2079 (top) and 2021-2054 (bottom). Trends detected by Mann-Kendall trend analysis for individual hydroelectric power abstraction locations.



**Figure 6.7.** Overview of the direction, magnitude and significance of annual trends in mean A<sub>daily</sub> for the periods 2021-2079 (top) and 2021-2054 (bottom). Trends detected by Mann-Kendall trend analysis for individual hydroelectric power abstraction locations.



**Figure 6.8.** Overview of the direction, magnitude and significance of annual trends in total abstraction for the periods 2021-2079 (top) and 2021-2054 (bottom). Trends detected by Mann-Kendall trend analysis for individual hydroelectric power abstraction locations.

Figures 6.9 and 6.10 provide more details on the seasonal trends observed in the four factors studied, for long-term and medium-term respectively. Examining the trends taking place in the 2021-2079 period, it can be seen that the most significant change generally occurs in the spring and autumn, with small declines in annual total abstraction and number of days Astart is achieved. In the medium-term, a greater number of statistically significant trends are seen in winter and summer, with increases in annual mean A<sub>daily</sub> also being statistically significant. Total abstraction also varies considerably between the two periods, with significant increases seen across both catchments and abstraction zones in winter and spring, and decreases in summer and autumn, leading to an overall decrease annually at the Z3 abstractions in the medium-term analysis. Little difference can be seen between the corresponding trends for the two catchments, with trend direction agreeing in all but three occasions, and few occasions of varying magnitude of change. In terms of difference between the two abstraction types, once again little variation is seen. However, when studying the percentage change from the baseline (2021-2030) annual average, with the near and far future decadal annual averages (Table 6.3), large differences in the number of days Amax is reached are shown between the zones. In the far-future average, the Z3 abstractions display a 9% decline in Tywi and stay stable in Conwy. Nonetheless, the Z2 abstractions increase quite significantly in both catchments, 32% and 18% respectively. Few differences are seen between the zones, catchments or time periods for the other factors studied. With the exception of total abstraction, the magnitude of change from the baseline is greater for the near-future than far-future.

		No.	days	A <sub>start</sub>	achi	eved			No.	days	s A <sub>max</sub>	read	hed	Legend			
		Win	Spr	Sum	Aut	Ann			Win	Spr	Sum	Aut	Ann	$\land$	Kendall's Tau 0 to 0.3		
70	Conwy	$\nabla$		$\nabla$				Conwy	Δ		Δ		Δ	$\land$	Kendall's Tau >0.3		
22	Tywi	▼		$\nabla$		▼	22	Tywi	Δ					$\land$	Kendall's Tau >0.6		
70	Conwy	•		▼	▼	V	70	Conwy	Δ		$\nabla$	▼	Δ	$\checkmark$	<sup>7</sup> Kendall's Tau <-0.6		
23	Tywi			$\nabla$		$\nabla$	23	Tywi	Δ				$\nabla$	$\checkmark \bigtriangledown$	Kendall's Tau <-0.3		
														$\checkmark$	Kendall's Tau -0.3 to 0		
		Ме	an A	<sub>daily</sub> (g	en d	ays)			т	otal	abstr	actio	n	$\Delta \nabla$	Non-significant trend		
		Win	Spr	Sum	Aut	Ann			Win	Spr	Sum	Aut	Ann		Significant trend at p < 0.05		
72	Conwy	$\nabla$		$\nabla$		Δ	70	Conwy	$\nabla$		$\nabla$						
22	Tywi	Δ		Δ		Δ	22	Tywi	Δ		$\nabla$		•				
	<b>C</b>	•			_			<b>6</b>	•			_	_				
Z3	Conwy	Δ		V		Δ	Z3	Conwy	Δ		V		•				
	Tywi	$\Delta$		$\nabla$		$\Delta$		Tywi	$\Delta$		$\nabla$		•				

**Figure 6.9.** Overview of the direction, magnitude and significance of annual and seasonal trends in future hydroelectric power characteristics for the period 2021-2079, as detected by Mann-Kendall trend analysis. Based on the average of all Zone 2 (Z2) and Zone 3 (Z3) abstractions in each catchment with the exception of total abstraction, which is based on the sum of the total abstraction of all locations in each respective zone.

		No.	days	A <sub>start</sub>	achi	eved				No.	days	s A <sub>max</sub>	read	hed	Lege	nd
		Win	Spr	Sum	Aut	Ann				Win	Spr	Sum	Aut	Ann	$\land$	Kendall's Tau 0 to 0.3
70	Conwy	Δ						70	Conwy			$\nabla$	▼	Δ	$\land$	Kendall's Tau >0.3
~~~	Tywi	Δ		▼		▼		22	Tywi	Δ	Δ	$\nabla$	$\nabla$	Δ	$\land$	Kendall's Tau >0.6
72	Conwy	$\nabla$	Δ	▼	▼	▼	1	72	Conwy			▼	▼		$\checkmark$	′Kendall's Tau <-0.6
23	Tywi	▼	Δ					23	Tywi			$\nabla$			$\mathbf{\nabla}$	Kendall's Tau <-0.3
							-								$\checkmark$	Kendall's Tau -0.3 to 0
		Me	an A	<sub>daily</sub> (g	en d	ays)				Т	otal	abstr	actio	n	$ extsf{a} \nabla$	Non-significant trend
		Win	Spr	Sum	Aut	Ann	_			Win	Spr	Sum	Aut	Ann		Significant trend at p < 0.05
72	Conwy				$\nabla$			77	Conwy					$\nabla$		
~~~	Tywi							22	Tywi					$\nabla$		
	Conwy			•	•		- 1		Conwy			•	-	-		
Z3	Tywi			<b>•</b>				Z3	Tywi			•	Ť	•		

**Figure 6.10.** Overview of the direction, magnitude and significance of annual and seasonal trends in future hydroelectric power characteristics for the period 2021-2054, as detected by Mann-Kendall trend analysis. Based on the average of all Zone 2 (Z2) and Zone 3 (Z3) abstractions in each catchment with the exception of total abstraction, which is based on the sum of the total abstraction of all locations in each respective zone.

**Table 6.3.** Percentage change in annual hydroelectric power characteristics when comparing far (2070-2079)and near (2045-2054) decadal averages with a baseline decade of 2021-2030.

			A <sub>start</sub> achieved	A <sub>max</sub> reached	Mean A <sub>daily</sub>	Total abstraction
	Comuni	2070-2079 average	-13%	+18%	+1%	-11%
77	Conwy	2045-2054 average	-15%	+54%	+12%	-5%
22	Tywi	2070-2079 average	-9%	+32%	+1%	-7%
		2045-2054 average	-11%	+47%	+8%	-2%
	Conver	2070-2079 average	-11%	0%	+3%	-9%
72	COllwy	2045-2054 average	-12%	+11%	+5%	-8%
25	Travi	2070-2079 average	-7%	-9%	-1%	-8%
	I YWI	2045-2054 average	-13%	+11%	+5%	-7%

#### 6.4.2 Public water supply

Under all three demand scenarios, in both the medium- and long-term, we observe an increase in the number of days annually when streamflow is not sufficient to satisfy demand, as well as an increase in the mean and total unmet demand volume (Figure 6.11). These trends are larger in magnitude for the 2021-2054 period and are also statistically significant for this study period (p < 0.05), with the exception of mean unmet demand in the decreased demand scenario (Figure 6.11). No unmet demand is recorded at any point in the winter months under any scenarios, whereas Kendall's Tau results suggest a decrease in unmet demand in spring, especially in the medium-term analysis, although this is not statistically significant. Autumn displays the most consistency in trends across both time periods and the three demand scenarios, with large magnitudes of change seen in total unmet demand and number of days demand unmet.

Total unmet demand								No. days demand unmet									Mean unmet demand				
	Demand	Win	Spr	Sum	Aut	Ann		Demand	Win	Spr	Sum	Aut	Ann			Demand	Win	Spr	Sum	Aut	Ann
079	Increased		$\nabla$	Δ		Δ	670	Increased		$\nabla$	Δ			_	079	Increased		$\nabla$	Δ		
21-2	Static		$\nabla$	Δ		Δ	21-2	Static		$\nabla$	Δ				1-2	Static		$\nabla$	Δ		Δ
202	Decreased		$\nabla$	Δ		Δ	202	Decreased		$\nabla$	Δ		Δ		202	Decreased		$\nabla$	Δ	Δ	Δ
054	Increased		$\nabla$				054	Increased		$\nabla$					054	Increased		$\nabla$	$\Delta$		
1-2	Static		$\nabla$				1-2	Static		$\nabla$					1-2	Static		$\nabla$	$\Delta$		
202	Decreased		$\nabla$				202	Decreased		$\nabla$					202	Decreased		$\nabla$	$\Delta$		Δ
														_							

#### Legend

•	
▲ △ Kendall's Tau 0 to 0.3	$\Delta$ Non-significant trend
🔺 🛆 Kendall's Tau 0.3 to 0.6	Significant trend at p <0.05

**Figure 6.11.** Overview of the direction, magnitude and significance of annual and seasonal trends in future public water supply unmet demand characteristics for the periods 2021-2079 and 2021-2054, as detected by Mann-Kendall trend analysis. No unmet demand is recorded in winter throughout the future period; therefore, no trends are displayed.

Further analysis (Table 6.4) reveals that for all factors considered, and under all demand scenarios, the degree of change between the annual averages of the baseline and near future, is much greater than that between the baseline and far future. For example, under the increased demand scenario, total unmet demand is 167% higher than the baseline in the near future, whereas the far future period is only 84% higher. Similarly, for the number of days demand is not met in the static demand scenario, the near future average is 151% higher than the baseline at 43.4 days per year, compared to only 30.7 days per year in the far future period, a 77% increase on the baseline of 17.2 days (Table 6.4). The same is true for mean unmet demand in the decreased demand scenario, with the near future period being 49% higher than the baseline, and the far future being only 10% greater.

		Increas	e demand	Static	demand	Decreased demand		
		Value	% change from baseline	Value	% change from baseline	Value	% change from baseline	
Total unmet	2021-2030	1.55		1.53		1.51		
demand	2045-2054	4.12	+167%	3.95	+159%	3.60	+138%	
(million m <sup>3</sup> )	2070-2079	2.84	+84%	2.58	+69%	2.11	+40%	
Davis damand	2021-2030	17.2		17.3		17.2		
Days demand	2045-2054	43.9	+155%	43.4	+151%	42.8	+149%	
unnet	2070-2079	31.4	+83%	30.7	+77%	28.6	+66%	
Maan unmat	2021-2030	57,015		55 <i>,</i> 893		55,639		
domand (m <sup>3</sup> )	2045-2054	93,162	+63%	90,232	+61%	83,140	+49%	
demand (m <sup>*</sup> )	2070-2079	73,506	+29%	68,721	+23%	61,351	+10%	

**Table 6.4.** Percentage change in annual public water supply unmet demand characteristics when comparing near(2045-2054) and far (2070-2079) future decadal averages with a baseline decade of 2021-2030.

#### 6.5 Discussion

#### 6.5.1 Hydroelectric power

The results of the MK trend analysis conducted on HEP abstraction characteristics display a range of changes, with most variation seen between the seasons and time periods studied, rather than between catchments or abstraction zones. These patterns are driven by the streamflow changes shown in Figure 6.2. An example of this are the declines seen in spring and summer flows in Conwy and Tywi up to the late 2050s, followed by an increase in summer flows to the end of the study period for Tywi and a plateau for Conwy. Autumn flows plateau after this point for both catchments. These seasonal streamflow changes lead to the differences in trends observed in the MK analysis, with larger and more statistically significant results being seen in the medium-term than long for declines in mean abstraction, number of days A<sub>start</sub> achieved, and number of days A<sub>max</sub> reached. Changes in these factors in summer in particular are only significant in the medium-term analysis, due to the levelling off (Conwy) and increase (Tywi) seen in streamflow after this period. This observed shift towards less days and volume of abstraction in the summer and autumn, and an increase in the winter and spring is in line with previously published research for Wales and the UK. To our knowledge, the only previous study that has investigated climate change impacts on hydropower in Wales specifically is the aforementioned work of Carless & Whitehead (2013). Their research presents trends in streamflow and potential HEP output for streams in the upper Severn catchment, similar in nature to the abstraction locations studied in the present research. A baseline, 2020, and 2050 scenarios are considered, with a marked decline in flow and power output in summer and early autumn months displayed in 2050 (Carless & Whitehead, 2013). Winter and early spring months show a smaller magnitude increase for the same period, leading to a stable situation annually (Carless & Whitehead, 2013). These results align well with that of our study's medium-term trend analysis for total abstraction. While no further studies use Wales as a region of focus specifically, other research based on the UK, or regions thereof, seems to be consistent with our results; Scotland for example, which is comparable to Wales due to the similar nature of catchments and also due to the high uptake of HEP in the country. Sample et al. (2015) summarises the research on the potential impacts of climate change on Scotland's HEP. It is suggested that run-off, which is highly sensitive to climate change, is closely linked to generation potential. This relationship, in line with the results of

this study, is projected to cause a decline in summer generation potential; although this is likely to be offset in the winter, HEP schemes may be unable to make use of the higher flows due to design limitations (Sample *et al.*, 2015). Further work in Scotland by Thompson (2012) on the impacts of climate change on streamflow also lend weight to the findings of this research, once again suggesting an accentuation of low summer flows and high winter flows in the future. These trends are also echoed on a UK-scale by research such as UK CEH (2012) and Kay *et al.* (2020) both corroborating the seasonal and annual driving streamflow trends observed in the study catchments of this research.

Furthermore, the trends observed of mean abstraction volume increasing, while total abstraction volume is decreasing, suggests that there will be fewer days of abstraction in the future, but with a greater abstraction per day. This trend also fits with the observed decrease in the number of days A<sub>start</sub> is reached, and the increases seen in the number of days A<sub>max</sub> is achieved. Once again, this is particularly pronounced in the medium-term analysis and is supported by research on low and high flows in Wales. Collet *et al.* (2018), Marx *et al.* (2018), Thober *et al.* (2018) and Visser-Quinn *et al.* (2019) all suggest that for Wales, in the future, the magnitude, duration and frequency of extreme high and low streamflows will increase.

In terms of observed differences between Z2 and Z3 abstraction types, a smaller magnitude of decline is observed for change in total abstraction in Z2 abstractions compared to Z3 in the medium- and long-term, as well as lack of statistical significance. This suggests that Z2 abstractions are potentially more resilient to climate-induced streamflow change in terms of annual total generation potential. This is possibly caused by the greater volume of water that it is allowable to abstract at these locations, being 1.3 times Q<sub>mean</sub>, as opposed to Q<sub>mean</sub> only in Z3 abstractions. This difference allows for Z2 abstractions to make greater use of larger magnitude flows and the observed increases in number of days A<sub>max</sub> is reached annually. This is further exemplified in the fact that Z2 abstractions on average in both the medium- and long-term have a larger increase in the number of days A<sub>max</sub> is reached, than Z3 abstractions.

While differences are seen for certain aspects between the two abstraction conditions zone types, little difference in observed trends is seen when comparing the two catchments. This lack of variation is likely due to the fact that the vast majority of the currently operating schemes, and therefore those studied, are located in the uppermost reaches of the

catchments, predominantly on first order streams. This positioning leaves little time for LULC differences to cause impact to run-off generation and streamflow significantly between catchments. The same is true for topographical characteristics, with the majority of abstractions being located in topographically homogenous areas across the two catchments. While all four of the factors studied are important for HEP scheme design, perhaps the most important in terms of future profitability is total abstraction. For all time periods, catchments and abstraction zones, a decline is observed, suggesting less generation output in the future.

#### 6.5.2 Public water supply

When viewing the unmet demand trend analysis results with the streamflow and demand trends, it can be seen that streamflow is the major factor, rather than water demand. In particular, large declines observed in summer and autumn streamflow lead to continued shortages of water, even in the reduced demand scenario. Similarly to HEP, the characteristics of future streamflow change also correspond with the difference seen between the mediumand long-term trend analysis conducted for PWS unmet demand. Spring streamflows increase considerably from the mid-2040s, while winter flows display a small increase; annual streamflow volume remains relatively stable throughout. Increases in winter and spring streamflows however do not factor into annual mean and total unmet demand, as streamflow is continually sufficient during these seasons. This means there is no offset for the increased demand and decreased streamflow seen in summer and autumn, leading to an increase annually in the number of days and volume of unmet demand under all scenarios. This problem is particularly pronounced in the medium-term, with both summer and autumn streamflow declining most steeply up to the mid-2050s. These results agree with published literature, with studies such as Sanderson et al. (2012), Henriques et al. (2015), and Afzal & Ragab (2019) projecting large decrease in water availability by the 2050s and beyond for the UK, and Wales specifically. Furthermore, the findings are commensurate with reports from organisation such as the ASC (2016a, 2016b), stating more action is needed to tackle future water supply security; Welsh Government (2015b), detailing the necessary steps for resilience in Welsh water supply; and, for catchments reliant on river-based abstraction such as Tywyn Aberdyfi and Pembrokeshire, DCWW (2019a). Imbalances observed could lead to problems for water supply, especially in times of prolonged drought, as increases in winter and spring flow will do little to combat summer and autumn shortages if reservoirs are already full.

#### 6.5.3 Study implications

With regards to HEP abstractions, due to lack of scheme specific data, the results represent an indication of future change in generation potential and timing for the study catchments, and more broadly for catchments across Wales, and the 364 HEP projects there within (Welsh Government, 2019b). For example, seasonal shifts in streamflow, and therefore generation capacity, will likely still impact on specific scheme layouts in a similar way to that that has been assumed in this study. The results are therefore helpful in planning future schemes and more generally, alteration in UK energy mix. Significantly less generation is expected to be possible in the summer and autumn in the future, which will be important to consider when planning the future resilience of the energy network. Overall, the results show that in the medium-term (2021-2054), abstraction characteristics and potential at installed HEP generation sites is likely to change greatly compared to the baseline; this is significant as this period falls at the end of the general lifespan of small hydroelectric projects (~40 years) being built in the last and upcoming five years (Hatata et al., 2019; Killingtveit, 2019). For those sites under development in the next five years, it may be beneficial to review scheme design in view of future flow conditions, for example by installing a larger turbine to make more use of increases in high flows, thus maximising generation potential over the lifetime of the project. In the longer-term, the magnitude of observed changes is smaller for all factors except total abstraction, than the medium-term. Therefore, when site redevelopment and replacement opportunities arise in the future, it will be important to once again looking at future projected change to best determine the most efficient scheme upgrades and alterations.

In terms of PWS, while the specific abstraction studied may be relatively resilient to future changes in streamflow, due to the supporting flow provided by the large upstream reservoir, the results of this work could have important implications for other, non-supported, direct river abstractions. The changes in streamflow and results on unmet demand demonstrated in this study, if applied to other catchments in Wales and the UK, could have major implications for future water supply sustainability. The increases in unmet demand that have been observed, even under a decreased demand scenario, clearly suggests that the pace of change in reduction of water demand by processes such as leakage reduction and domestic water use education and awareness raising, needs to be implemented as soon as possible. These

measures should certainly be the first step in ensuring future water supply security, rather than large infrastructure projects to increase supply, in all but the most pressing of cases.

The results and implications of this work can also be paralleled to other large water abstractors in the region which will be operating under the same streamflow regime changes. Industrial and agricultural users in particular could face challenges, causing alterations in operation practices and timing, due to lack of available water, especially for those abstracting directly from rivers and streams. The abstraction location and use data used to identify runof-river HEP schemes for this study details over 217 surface water-based abstractions used for industrial and commercial purposes, and a further 444 used for agriculture across Wales. These numbers highlight the potential scale of impact of future hydrological regime changes, and the need to act now to ensure resilience to such changes.

## 6.6 Conclusions

This research has demonstrated clear trends in future availability of water for the HEP locations studied in the Conwy and Tywi catchments, as well as the major PWS abstraction in the Tywi. Our results highlight that spring and autumn are the key change seasons long-term in terms of water availability, while winter and summer are more impactful in the medium-term. The results also demonstrate the need for action now, especially in terms of the planning and design of HEP installations. Schemes and turbines must be designed with future streamflow patterns in mind, in particular increasing winter and spring flows, rather than decreasing summer and autumn flows, thus allowing for the maximisation of power generation over the course of a year. Additionally, regardless of future demand decline, increased pressure is likely to be placed on PWS due to large declines seen in future streamflows, therefore careful planning is also required here in order to ensure continued water supply.

Finally, we suggest avenues of future research. Firstly, the impact of different climate change/emissions scenarios on future HEP output and PWS availability is required at a national-scale. While a worst-case scenario approach has been taken in this study, it is paramount to consider the range of potential outcomes under different emissions pathways, to ensure that the most proportionate future planning and adaptation measure are

undertaken. Furthermore, research on specific HEP schemes, with historical generation data, would be highly beneficial as this would allow future climate change scenarios to be tested after calibration with actual data, giving a better understanding of how future hydrological regime change will impact on generation potential. Secondly, further research taking a holistic view of all catchment abstractions is needed, including industrial, commercial, and agricultural purposes, as well as PWS and HEP. This approach would give greater insight to how total catchment water demand will impact on individual water users, as well as the catchment as a whole. In addition to this, the inclusion of impacts of other climate change mitigation measures, such as upland management for flood control, would give a complete picture of future potential alterations. This holistic catchment level approach would give the best chance of ensuring fair and equitable distribution of changing future water resources.

Acknowledgements: This study was conducted as part of the Dŵr Uisce project, which aims at improving the long-term sustainability of water supply, treatment and end-use in Ireland and Wales. The project has been supported by the European Regional Development Fund (ERDF) through the Interreg Ireland-Wales Co-operation Programme 2014-2020. The work has been used as a case study for Dŵr Uisce Work Package 7 "climate change", and forms part of a study assessing the impact of climate change on water resource exploitation in Wales. The authors would like to thank Natural Resources Wales for the provision of abstraction location and type/use data, and Dŵr Cymru Welsh Water for the provision of actual abstraction volume data.

# **CHAPTER 7: DISCUSSION**

## 7.1 Hydroclimatic trends and implications

While a discussion on the findings of each study is found in each of the study chapters, here we will discuss the results in relation to each other, and their applicability at a larger scale. When the results of all three studies (Chapters 4, 5, and 6) undertaken in thesis are viewed together, or as a subset thereof, several aspects become clear. First, it is evident that observed historical changes in hydroclimatology are indicative of the beginning of what could be much larger change for Wales in the future. For example, in the analysis of the 34 years of historical data, only small change is seen for the most part in seasonal and extreme streamflow, however, much greater change is projected for the future, as increasing global greenhouse gas concentrations continue to take greater effect on the global climate system (IPCC, 2013). It is also notable that those statistically significant trends seen in the historical data continue into the future; the observed trend of increased average autumn temperatures, for example, continues for all catchments to 2080 (Section 5.4.1). This trend also follows the expected acceleration of global temperature changes over the next 60 years (IPCC, 2018), with the average rate of change across all five catchments between 1982-2015 being 0.039°C per year, compared to 0.062°C per year between 2021-2080. Additionally, trends observed in both Study 1 and Study 2 demonstrate a greater degree of change in extreme events than seasonal and annual averages; historically, this change is in agreement with work for the UK such as Hannaford & Buys (2012), Hannaford (2015) and Watts et al. (2015a). Accentuation of streamflow seasonality is most pronounced in the future average and extreme trends presented in Study 2 (Section 5.4.2), with winter and spring set to get wetter, and summer and autumn drier, in line with accepted projections for the UK (Prudhomme et al., 2012; Suggitt *et al.*, 2015; Garner *et al.*, 2017; Lowe *et al.*, 2018).

The results shown in Studies 1 and 2 are also indicative of the results found at a European scale. Stahl *et al.* (2010), for example, when studying 441 near-natural catchments across Europe for the period 1962-2004, presented similar trends to those found in Study 1 for Wales. This similarity in changes shown for Wales is especially true for trends in seasonality, with these in turn being very similar to the magnitude of change presented for other Atlantic-influenced western European countries (Stahl *et al.*, 2010). In addition, Forzieri *et al.* (2014) found similar future seasonal shifts for non-snow influenced catchments in Europe as are presented in Study 2 of this thesis. The 7-day average streamflow of a control period (1961-

1990) was compared with projected streamflow in the 2080s under the SRES A1B scenario (equivalent to between Representative Concentration Pathways (RCP) 6.0 and 8.5) from the fourth assessment report of the Intergovernmental Panel on Climate Change (IPCC). Of the twelve catchments studied, those in western and central continental Europe in particular (especially France, Italy, and the Czech Republic) show the same accentuation of winter high flows and summer low flows (Forzieri *et al.*, 2014) as projected in the results of this research (Section 5.4.2.3). In addition, the work of Schneider *et al.* (2013) shows trends for decreases in annual 1-day minimum flow, and increases in 1-day maximum flow in temperate oceanic Europe, when comparing a baseline period of 1971-2000 with modelled data for 2041-2070. Their trend observations clearly correlate with those shown in Section 5.4.2.2 of this thesis, being similar in magnitude as well as direction.

The benefit of conducting climate change evaluation studies and impact assessments at a catchment scale is highlighted in particular in Studies 1 and 2. The results of both chapters demonstrate how factors such as physical geographic location, as well as catchment characteristics can impact on the influences of nationwide climatic forcing. The streamflow results presented in Study 1 (Section 4.4.1.3), for example, show how the catchments in the north of Wales display different trends to those in the south, due to the impact of the North Atlantic Oscillation, as evidenced in earlier research by the likes of Dixon et al. (2006), Beranová & Huth (2007), and Osborn (2011). Furthermore, Study 2 describes how different and individual catchment characteristics such as underlying soil type, topography, and land cover can influence the responses of a catchment to long-term changes in climate. This is particularly well demonstrated by the Conwy and the Dyfi, which display similar changes in streamflow characteristics, aside from low flows in winter and spring, as well as annual average flow (Section 5.4.2). These differences are chiefly caused by variation in land cover characteristics, in particular a greater proportion of coniferous forest in the Dyfi which has been shown by Robinson et al. (2003) to help maintain baseflow conditions. This difference is the key driver of the distinct opposition of trends seen in terms of future total phosphorous (TP) and total nitrogen (TN) concentrations in winter and annually in particular, with the Dyfi showing a decline, and the Conwy an increase (Section 5.4.3). Both TP and TN are shown to correlate with streamflow statistically significantly in the modelling, explaining the discrepancies between the two catchments, despite the broadly similar hydrological response

overall. These findings correspond with the findings of Dearmont *et al.* (1998) and Price & Heberling (2018) who detail decreases in overall water quality, including nutrients, with increased streamflow. Furthermore, a study by Whitehead *et al.* (2009b) also reports a decline in water quality linked with climate change influenced alterations in streamflow. Dissolved oxygen (DO) and TP concentrations in particular are shown to be highly sensitive to streamflow (Whitehead *et al.*, 2009b), once again demonstrating that a catchment approach, taking account of local streamflow conditions, is key for robust projections of future water quality changes.

Moreover, results presented in Study 3 show how even within catchments, hydrological responses can vary (Section 6.4.1). The Conwy catchment in particular displays significant trends only in the most westerly hydroelectric power (HEP) abstraction sites for long-term (2021-2079) changes in the number of days maximum abstraction volume is achievable, and the mean annual abstraction of water (Section 6.4.1). This finding of intra-catchment variation corresponds with those of Coulthard et al. (2005), who linked within catchment variation in erosion and deposition phases with local discharge distinction in the River Ouse, Yorkshire. Similarly, Tetzlaff et al. (2007) showed that differences in soil type was the key driver of intracatchment variability in streamflow for a 223 km<sup>2</sup> case study catchment in Scotland. Additionally, changes in HEP generation characteristics seen in this thesis, such as a decrease in the number of days per year when generation is achievable, link well with alterations in extreme streamflow characteristics detailed in Study 2 (Section 5.4.2.2). In this case a decline in minimum flow volumes and an increase in number of low flow days is the driving cause; consistent with the findings of Carless & Whitehead (2013) for the River Severn. Furthermore, nuances in the few differences between the HEP results for each catchment also link to variation in extreme streamflows. Taking the number of days maximum abstraction is achievable in summer for Zone 3 abstractions in the long-term, as an example, it can be seen that the opposing trends (decrease in Conwy, increase in Tywi) correlate with the corresponding opposing trends for both catchments in 1-day maximum flow, and to a lesser extent, number of high flow days. This high sensitivity of HEP schemes to small changes in streamflow trends is mirrored by the findings of Sample et al. (2015), for Scotland, who found generation potential, and therefore profit margin to be highly sensitive to changes in precipitation and streamflow. Furthermore, one of the first papers to look at climate change

impact on HEP in the UK, Harrison (2006), found that decreases in generation potential in summer months are likely to outweigh the increases in winter months, due to design limitations of installed infrastructure. Relatively little work has been completed either nationally, regionally, or globally on the impact of climate change on HEP generation, the findings presented here go some way to starting to address this for small scale HEP, with the results being more widely applicable than the area studied. Small-scale HEP technology is broadly standard globally, and while clearly different countries will have different regulations in terms of abstraction licence conditions, the results presented here can still be generalised to similar catchments in other regions. Indeed, with both the historical and future streamflow patterns presented in this thesis corresponding well with the trends seen in northern and central Europe in other work, as discussed earlier, it is reasonable to assume that HEP results presented in Study 3 are also indicative of likely change in abstraction and generation potential for those areas.

Study 3 also highlights the importance of study length when conducting impact assessments, with considerable differences being observed in the trends displayed for medium-term (2021-2054) and long-term (2021-2079) analysis periods (Section 6.4). Therefore, it is crucial that impact assessment studies take account of this, to ensure that projected future trends, and consequently their impact, are realistic and useful to the industry being studied. In this instance, the medium-term scenario is particularly applicable to HEP as it falls within the initial expected generation lifespans of schemes installed recently, giving important insight into the potential changes in generation regime. For public water supply (PWS), both future periods studied are useful, allowing for medium- and long-term planning to be implemented.

## 7.2 Future water quality implications for abstractors

One key relationship that has not previously been discussed in this thesis is the impact of the water quality findings presented in Study 2 (Section 5.4.3), in relation to the results of Study 3 (Section 6.4) for both PWS and HEP. For HEP, changes in suspended sediment (SS) loads are likely to have a deleterious impact on infrastructure, in particular the turbine blades and casing. Changes in SS loads are expected to cause additional abrasion, scouring and pitting, especially when considering that some of the largest increases in SS loads are seen in the periods of heaviest streamflows, and therefore greatest throughput of water and sediment

through the HEP system. Indeed, Padhy & Saini (2008) conclude that sediment concentration, flow velocity, grain size, and grain hardness are the key determinants of abrasion potential of suspended sediments on HEP turbines. Work by Naidu (1999) supports this, describing how a 10% increase in water velocity (analogous to sediment velocity here) increases erosion capability by 27%, while Neopane (2010) describes a directly proportional relationship between erosion potential and particle concentration. Under the future projections made in Studies 2 and 3, both of these factors increase, with greater concentrations of SS shown to be likely during the main generation seasons, as well a larger proportion of time that HEP schemes are running at maximum capacity, and mean abstraction on generation days becoming larger. In addition, while we can assume that grain hardness remains constant over the period of study, larger flow velocities such as those projected in this work also have the ability to suspend larger particles, thereby further increasing the abrasion potential. The findings of Worster & Denny (1955) demonstrate that when considering erosion caused by sliding and grazing, particle size is directly proportional to amount of wear. Although, further findings by Padhy & Saini (2011) showed that fine sediments can also be highly damaging to turbines, especially in systems with high head.

The multi-faceted change in abrasion potential clearly have implications for the maintenance and life-span of HEP schemes, even despite the projected overall decrease in future total abstraction potential. These changes could impact on their viability, productivity, and overall output of schemes, further confounding pressures already faced by small-scale HEP such as business rate increases and the curtailment of feed-in-tariffs, as detailed in Section 1.2. Careful consideration should therefore be given to this area during the design, construction, and operation phases of new HEP schemes, in order to ensure that either greater protections are made (e.g. finer screens on intake weirs, or sediment bypass systems) or more regular servicing of parts is completed. For schemes that are already in operation, greater monitoring should be undertaken, to guard against component failure or system damage, and in some cases, upgrading of parts or retrofitting of systems may be needed (Hauer *et al.*, 2018).

In addition, the impact of the studied water quality parameters on PWS is likely to be large, not just due to the four factors themselves, but also due to knock-on impacts, such as higher TN and TP levels giving rise to greater eutrophication. These alterations are likely to lead to an increase in required treatment to maintain drinking water standards, this has economic

and operational efficiency impacts, and consequently an environmental toll. The link between raw water quality and treatment cost has long been established, with Dearmont *et al.* (1998), for example, showing that for twelve water treatment works in Texas, a 1% increase in SS concentrations increases chemical costs by 0.25%. This increase in cost is not just related to the removal of sediments however, as SS can be seen as an indicator of wider water quality, which also requires greater treatment (Dearmont *et al.*, 1998). Indeed, SS is an important transport medium for various pollutants which can become adsorbed to sediment grains (Dearmont *et al.*, 1998), including phosphorus (House *et al.*, 1998), heavy metals, and bacteria (Loperfido, 2014). This clearly makes SS a highly influential water quality parameter, with increases in concentrations bringing a variety of challenges for water treatment facilities, likely increasing costs and chemical requirements, and lowering treatment capacity.

A review by Price & Heberling (2018) investigated the impacted of changes in water quality parameters on the operation and maintenance (O&M) cost of water treatment plants. Along with sediment, nutrients and pesticides (TP, TN and nitrates) were found to be among the key drivers of O&M cost increases induced by lower raw water quality (Price & Heberling, 2018). Various studies support this conclusion, Forster & Murray (2007), for example, found a 0.08% increase in O&M cost per 1% increase in total pesticide load, when studying the agricultural Maumee Catchment in Ohio. These results correspond with the results of Murray (2001) and Fiquepron *et al.* (2013) who found 0.12% and 0.01% increases in O&M costs per 1% increase in total pesticide load, catchments across France respectively. Also looking at nitrogen loads individually, Fiquepron *et al.* (2013) demonstrated a 0.05% O&M cost increase, while Mosheim & Ribaudo (2017) found the value to be 0.06%, both results are relative to 1% increase in total nitrogen concentrations. For phosphorous individually, the results are similar, McDonald & Shemie (2014) reporting a 0.19% increase per 1% concentration increase, and Heberling *et al.* (2015) suggesting a value of 0.06%.

When considering the wastewater treatment side of the water supply process, Astaraie-Imani *et al.* (2012) found that future climate change will reduce water quality downstream of wastewater treatment works in the UK. This is caused by more frequent releases from storm tanks and combined sewer overflows due to increases in the frequency and magnitude of extreme precipitation events (Astaraie-Imani *et al.*, 2012), as also demonstrated in Section 5.4.2.2. These projections are supported by the work of Abdellatif *et al.* (2015) for the

northwest of England, who project a 37% increase in total combined sewer overflow spill volume and a 32% increase in total spill duration, when comparing 1961-1990 with 2070-2099 under a future emission scenario equivalent to RCP8.5. These results are congruous with those of Tait *et al.* (2008) who indicate a combined 40% increase in sewer overflow and flood tank spill volume by 2080 for an urban catchment in the UK. The most prominent water quality implications of these changes are declines in dissolved oxygen (DO) concentrations, increases in nutrient levels and the release of harmful bacteria to waterways (Astaraie-Imani *et al.*, 2012). These changes have important implications for downstream ecology, as well impact on bathing waters and shellfish harvesting. Given that combined sewer systems are common in Wales and the UK, and that coastal waters support a variety of industries, not least of all tourism and fishing, this is clearly an area of concern for the region, and nation.

In light of the findings presented above, and given the degree of projected water quality change in Study 2, it is clear that water service providers could face substantial challenges in maintaining both acceptable future water supply quality and water quality downstream of wastewater treatment facilities. In terms of drink water supply, the first steps in trying to ensure output quality could be to lengthen retention times in the different treatment processes or altering the dosing of chemicals used, followed by changing treatment processes if needed. Both of these options are likely to decrease the volume of water that is possible to treat in a given timeframe, therefore effecting plant efficiency, expenditure per litre treated, and the environmental impact in terms of additional chemical use. It is also possible that such changes would not be sufficient to maintain drinking water standards, therefore investment in both new treatment infrastructure, as well as new water sources may be necessary to ensure the safety and continuity of supplies (Jones *et al.*, 2007; Hanson *et al.*, 2016).

### 7.3 Summary

Overall, this work has shown that catchments in Wales have already undergone alterations due to climate change, and that these alterations will continue into the future, likely at an expedited rate compared to that already seen. Such changes in average and extreme streamflows, demonstrated in this thesis, will have wide-reaching consequences. Firstly, river systems themselves will alter due to changing flow regime as a consequence of altered precipitation input. This will be especially felt in terms of decreased water quality and likely

changes in aquatic life and geomorphological characteristics. Furthermore, catchments as a water resource, it has been shown, will alter, with changing timing, quantity, and quality of water available for abstraction or use. Given the reliance on surface waters in Wales, this could have significant impacts in terms of agricultural productivity, energy generation, manufacturing activity, the leisure and tourism industry, and PWS; consequently, these changes could have a large impact on the economy of the nation.

In addition, the trends observed in this research appear also to be relevant to a wider region than that focussed on in this thesis. Several studies that have examined the whole of Europe, and that have been presented in this discussion (Stahl *et al.*, 2010; Schneider *et al.*, 2013; Forzieri *et al.*, 2014; Papadimitriou *et al.*, 2016), have shown that the magnitude and direction of trends seen in Wales, generally corresponds with other northern and western European areas where the climate is also influenced by the Atlantic. The results of this research can therefore be seen as also adding value to the knowledge base of this wider region, and not just those catchments and the area specifically studied. It should also be kept in mind, that the projections shown in this work offer a worst-case scenario of the future, based on high future global greenhouse gas emissions. Therefore, while planning for the future impacts of climate change is highly important, and these results do offer crucial insight, it is likely that the magnitude of changes seen, and therefore impacts, will be reduced. Nonetheless, robust planning for future water resource allocation is critical, therefore, the new findings and conclusions drawn from this thesis and the research within are important and useful.
# **CHAPTER 8: CONCLUSIONS**

Conclusions

#### 8.1 Final conclusions

The research presented in this thesis has been undertaken to further our knowledge in the field of catchment level hydroclimatic changes in Wales, and the impacts of such changes on the exploitation of surface water resources. Through worst-case future emission scenario hydrological modelling it has been shown that Wales could potentially face substantial challenges in the future in terms of supply-demand balance, not just for public water supply (PWS), but for all sectors that make use of surface water resources. Increased future hydrological seasonality has been shown, with spring and autumn being the most susceptible to change both in terms of average flows and extreme events. Furthermore, significant potential future challenges in terms of water quality have been identified, which will impact on those tasked with managing surface waters in Wales, as well as any users or abstractors of the resource in the region. Two sectors have been explored in terms of the changes identified, PWS and small-scale hydroelectric power (HEP). Both sectors face the need for adaptation in order to cope with changes in the timing, quantity, and quality of surface water resources in the medium- and long-term.

Overall, this thesis has been successful in achieving its aim of an assessment of the impact of future climate change on the streamflow characteristics of catchments in Wales, and an evaluation of the effect of such changes on the abstractable water resource (as laid out in Section 1.3). Historical changes in daily temperature, precipitation and streamflow have all been analysed for the period 1982-2015, as well as corelated with total water demand for 2012-2015, giving new insight into water use under changing weather conditions in the region, and addressing objectives I and II of this thesis. Objectives III and IV were achieved with five induvial hydrological models being set-up, calibrated, validated, manipulated, and implemented with future worst-case emissions scenario climate data, allowing for an analysis of future hydrological regime and water quality change. Finally, objective V has been addressed in terms of a projection of future unmet demand for a key river-based PWS abstraction location under three demand scenarios, as well as future abstraction potential for 25 HEP installations in two catchments.

The future hydroclimatic picture for Wales is complex and potentially faces major change, but that is not to say it is without hope, especially with studies such as those presented in this

thesis providing valuable insight into the future planning, management and adaptation need. A significant proportion of future Welsh prosperity and development could be linked to the important surface water resource. Therefore, further study focussing on Wales specifically and the challenges it faces, in the manor outlined above, is essential. As aforementioned, the study has also highlighted the importance of local-scale study, as well as national level analysis, with each scale important for different purposes and a combination of the two approaches required to best plan for the future management and use of water resources.

#### 8.2 Recommendations for further work

While the findings of this research have addressed knowledge gaps surrounding the impacts of climate change on the hydrology of Wales, and water resource use thereafter, further research is still required, especially around impact assessment. It is clear from the results of this work for example, that extreme high and low streamflow events are increasing in terms of frequency and magnitude (Chapters 4 and 5). While Chapter 6 goes someway to address the implications of these changes on PWS and HEP, more work is required to fully understand the nature of these changes, especially the duration of drought periods, which could have potentially large consequences for all water users, but PWS especially. In terms of extreme high flow events, this work provides background for further study on areas such as fluvial flooding, with modelling at a finer temporal and spatial resolution required to develop the most informed future planning strategies. Furthermore, the relationship established in Chapter 4 between prevailing weather conditions and overall water demand is an important first step in better managing the future supply-demand balance in Wales. This work could be continued further by the implementation of a component analysis, to give a better understanding of how different user groups (agriculture, domestic, industry, etc.) respond to changes in factors such as air temperature. Analysis of this kind would enable a more rounded picture of future water demand to be achieved, enabling robust water resource planning.

The results presented in Chapter 6 represent an overview of the likely impacts of hydroclimatic change on the two industries studied. While PWS impacts have been studied and shared by Dŵr Cymru Welsh Water (DCWW) in the form of their water resource management plan (DCWW, 2019a) and drought plan documents (DCWW, 2019b), little work has been completed for other industries, especially with a focus on Wales. HEP may play a

Conclusions

relatively small role in the overall energy portfolio of Wales, but the projected changes in the seasonality and performance of the sector could have significant implications for HEP schemes, both locally and nationally. It is therefore crucial that further research be completed for the whole of Wales, and more broadly the UK, to fully understand the significance of such changes for the energy provision mix of the nation. On a more localised level, it is important that research be carried out on individual HEP schemes, to ensure optimised operation for the future. Looking more broadly than those industries studied in this research, the impact of projected hydroclimatic changes on sectors such as heavy industry and agriculture are also important to understand, especially for the latter, given the increasing pressure on global and local food systems. Clearly, further work is needed to fully understand the risks posed by future hydroclimatic changes on the overall exploitation of water resources in Wales, and the potential economic and environmental impacts of any changes.

Moreover, Chapter 5 indicates the potential changes in four water quality variables, linked to changing hydroclimatic factors, with substantial alterations likely for all four as well as water quality more generally. Once again, the implications of these changes in Wales has received little attention thus far, clearly making this a key area of future research. Various elements would benefit from further work in this regard: in-stream ecology risks; eutrophication changes; the impact on drinking water treatment plant processes, efficiency, and finances; suspended sediment load changes and implications for HEP; and future compliance with environmental regulations, to name just a few. Furthermore, it has been shown that changes in water temperature are very likely given projected changes in air temperature and the known relationship between the two (Crisp & Howson, 1982; Erickson & Stefan, 2000; Morrill *et al.*, 2005; Hannah & Garner, 2015). It is therefore suggested that an area for future research should be the impact of increased water temperatures for Wales, which has the potential to impact not only in-stream processes, and therefore overall water quality, but also processes and chemical reactions in environments such as drinking water treatment plants.

#### 8.3 Final summary

This thesis has provided a grounding knowledge of spatial and temporal changes in Welsh hydroclimatology, giving insight into the future environment that abstractors, water resource managers, and water users will be operating in. The thesis also presents the need for greater

work at a catchment level, with Chapters 4 and 5 in particular demonstrating how significant change can occur between catchments even over small distances, due to factors such the topographical characteristics of the catchment and its surrounds. Indeed, Chapter 6 highlights that variation in streamflow response can be found even within catchments. This underscores the need for awareness of small-scale studies and local management plans in the future, especially when working to address or mitigate against the impacts of future climate change.

Several knowledge gaps have been addressed, as well as novel research methods used, through the course of the thesis. First, a daily and monthly relationship between total water abstraction and three hydroclimatic factors has been established for the first time for Wales, allowing, as conducted in Study 3, for projections of future abstraction requirements for PWS to be made. In terms of Study 2, the research conducted is some of the first to make use of the latest UK climate projections (UKCP18), with little peer-reviewed literature having been published using this data at the time of writing. This is especially true when considering both the study area and research topic, with no articles having been published using the data for Wales, or in reference to future water resource use. Furthermore, the use of the Soil and Water Assessment Tool (SWAT) for modelling of streamflow and water quality is one of the first for Wales, with the future water quality analysis itself being previously sparsely published on for the region specifically. Finally, Study 3 develops for the first time a method for assessing the impacts of streamflow changes on the abstraction potential of multiple individual HEP schemes, where the abstraction licences conditions are unknown. This represents a methodological advancement, allowing for large-scale assessments of the impact of climate change on HEP generation potential, this has previously not been available.

Overall, the research conducted has been successful in addressing the objectives of the thesis and, therefore, achieving the overarching aim laid out at the start of project. Furthermore, the work presented represents an advancement of knowledge for the field, as set-out above, by addressing knowledge gaps in relation to the hydroclimatic character of catchments within Wales and implications for water resource use. It is hoped that the findings of this research will be used to inform future multi-sector water resources management planning, as well as being useful to the specific sectors studied and more, with the results being applicable to a variety of stakeholders. In this manor it is hoped that the findings of the thesis will help towards ensuring long-term sustainability of water abstraction and use in Wales, and beyond.

## REFERENCES

- Abbaspour, K.C. 2014. SWAT-CUP 2012: SWAT Calibration and Uncertainty Programs A user manual. Duebendorf, Swtizerland: Swiss Federal Institute of Aquatic Science and Technology.
- Abdellatif, M., Atherton, W., Alkhaddar, R.M. & Osman, Y.Z. 2015. Quantitative assessment of sewer overflow performance with climate change in northwest England. *Hydrological Sciences Journal*, **60**: 636–650.
- Abera, F., Asfaw, D., Engida, A. & Melesse, A. 2018. Optimal operation of hydropower reservoirs under climate change: The case of Tekeze reservoir, eastern Nile. *Water*, **10**: 273.
- Adler, R.F., Gu, G., Sapiano, M., Wang, J.J. & Huffman, G.J. 2017. Global precipitation: Means, variations and trends during the satellite era (1979–2014). *Surveys in Geophysics*, **38**: 679–699.
- Adler, R.F., Huffman, G.J., Chang, A., Ferraro, R., Xie, P.P., Janowiak, J., Rudolf, B., Schneider, U., Curtis, S., Bolvin, D., Gruber, A., Susskind, J., Arkin, P. & Nelkin, E. 2003. The Version-2 Global Precipitation Climatology Project (GPCP) monthly precipitation analysis (1979–present). *Journal of Hydrometeorology*, 4: 1147–1167.
- Afzal, M. & Ragab, R. 2019. Drought risk under climate and land use changes: Implication to water resource availability at catchment scale. *Water*, **11**: 1790.
- Ahmed, K., Sachindra, D.A., Shahid, S., Demirel, M.C. & Chung, E.S. 2019. Selection of multimodel ensemble of general circulation models for the simulation of precipitation and maximum and minimum temperature based on spatial assessment metrics. *Hydrology* and Earth System Sciences, 23: 4803–4824.
- Alamanos, A., Sfyris, S., Fafoutis, C. & Mylopoulos, N. 2020. Urban water demand assessment for sustainable water resources management, under climate change and socioeconomic changes. *Water Supply*, **20**: 679–687.
- Alcamo, J., Florke, M. & Marker, M. 2007. Future long-term changes in global water resources driven by socio-economic and climatic changes. *Hydrological Sciences Journal*, **52**: 247– 275.
- Alexander, L. V. 2016. Global observed long-term changes in temperature and precipitation extremes: A review of progress and limitations in IPCC assessments and beyond. *Weather and Climate Extremes*, **11**: 4–16.
- Alexandersson, H. 1986. A homogeneity test applied to precipitation data. *Journal of Climatology*, **6**: 661–675.
- Alexandersson, H. & Moberg, A. 1997. Homogenization of Swedish temperature data. Part I: Homogeneity test for linear trends. *International Journal of Climatology*, **17**: 25–34.
- Arnell, N.W. 2004a. Climate change and global water resources: SRES emissions and socioeconomic scenarios. *Global Environmental Change*, **14**: 31–52.
- Arnell, N.W. 2004b. Climate change impacts on river flows in Britain: The UKCIP02 scenarios. *Water and Environment Journal*, **18**: 112–117.
- Arnell, N.W. 2011. Incorporating climate change into water resources planning in England and Wales. *Journal of the American Water Resources Association*, **47**: 541–549.

- Arnell, N.W. & Delaney, E. 2006. Adapting to climate change: Public water supply in England and Wales. *Climatic Change*, **78**: 227–255.
- Arnold, J.G., Kiniry, J.R., Srinivasan, R., Williams, J.R., Haney, E.B. & Neitsch, S.L. 2012. *Input/output documentation*. Texas, USA: Texas Water Resources Institute.
- Arnold, J.G., Srinivasan, R., Muttiah, R.S. & Williams, J.R. 1998. Large area hydrologic modelling and assessment part I: Model development. *Journal of the American Water Resources Association*, 34: 73–89.
- Arora, R., Tockner, K. & Venohr, M. 2016. Changing river temperatures in northern Germany: Trends and drivers of change. *Hydrological Processes*, **30**: 3084–3096.
- Arrhenius, S. 1896. On the influence of carbonic acid in the air upon the temperature of the ground. *Philosophical Magazine and Journal of Science*, **41**: 237–276.
- Asadieh, B. & Krakauer, N.Y. 2015. Global trends in extreme precipitation: Climate models versus observations. *Hydrology and Earth System Sciences*, **19**: 877–891.
- ASC. 2016a. UK climate change risk assessment 2017 synthesis report: Priorities for the next five years. London, England: Adaptation Sub-Committee of the Committee on Climate Change.
- ASC. 2016b. UK climate change risk assessment 2017 evidence report: Summary for Wales. London, England: Adaptation Sub-Committee of the Committee on Climate Change.
- Ashmore, P. & Church, M. 2001. *The impact of climate change on rivers and river processes in Canada*. Ottawa, Canada: Natural Resources Canada.
- Ashofteh, P.S., Haddad, O.B. & Mariño, M.A. 2013. Climate change impact on reservoir performance indexes in agricultural water supply. *Journal of Irrigation and Drainage Engineering*, **139**: 85–97.
- Astaraie-Imani, M., Kapelan, Z., Fu, G. & Butler, D. 2012. Assessing the combined effects of urbanisation and climate change on the river water quality in an integrated urban wastewater system in the UK. *Journal of Environmental Management*, **112**: 1–9.
- Atkins, Mott MacDonald, Nera, HR Wallingford & Oxford University. 2016. *Water resources long term planning framework (2015-2065): Technical report*. London, England: Water UK.
- Atta, R. & Dawood, M. 2017. Spatio-statistical analysis of temperature fluctuation using Mann–Kendall and Sen's slope approach. *Climate Dynamics*, **48**: 783–797.
- Azari, A., Hamzeh, S. & Naderi, S. 2018. Multi-objective optimization of the reservoir system operation by using the hedging policy. *Water Resources Management*, **32**: 2061–2078.
- Baggaley, N.J., Langan, S.J., Futter, M.N., Potts, J.M. & Dunn, S. 2009. Long-term trends in hydro-climatology of a major Scottish mountain river. *Science of The Total Environment*, 407: 4633–4641.
- Bailey, R.T., Wible, T.C., Arabi, M., Records, R.M. & Ditty, J. 2016. Assessing regional-scale spatio-temporal patterns of groundwater-surface water interactions using a coupled SWAT-MODFLOW model. *Hydrological Processes*, **30**: 4420–4433.
- Balling, R.C. & Gober, P. 2007. Climate variability and residential water use in the city of Phoenix, Arizona. *Journal of Applied Meteorology and Climatology*, **46**: 1130–1137.

- Basarin, B., Lukić, T., Pavić, D. & Wilby, R.L. 2016. Trends and multi-annual variability of water temperatures in the river Danube, Serbia. *Hydrological Processes*, **30**: 3315–3329.
- Beaufort, A., Moatar, F., Sauquet, E., Loicq, P. & Hannah, D.M. 2020. Influence of landscape and hydrological factors on stream-air temperature relationships at regional scale. *Hydrological Processes*, **34**: 583–597.
- Bell, V.A., Kay, A.L., Cole, S.J., Jones, R.G., Moore, R.J. & Reynard, N.S. 2012. How might climate change affect river flows across the Thames Basin? An area-wide analysis using the UKCP09 Regional Climate Model ensemble. *Journal of Hydrology*, **442–443**: 89–104.
- Beranová, R. & Huth, R. 2007. Time variations of the relationships between the North Atlantic Oscillation and European winter temperature and precipitation. *Studia Geophysica et Geodaetica*, **51**: 575–590.
- Bessa Santos, R.M., Sanches Fernandes, L.F., Vitor Cortes, R.M. & Leal Pacheco, F.A. 2019. Development of a hydrologic and water allocation model to assess water availability in the Sabor River Basin (Portugal). *International Journal of Environmental Research and Public Health*, **16**: 2419.
- Betts, R.A., Alfieri, L., Bradshaw, C., Caesar, J., Feyen, L., Friedlingstein, P., Gohar, L., Koutroulis, A., Lewis, K., Morfopoulos, C., Papadimitriou, L., Richardson, K.J., Tsanis, I. & Wyser, K. 2018. Changes in climate extremes, fresh water availability and vulnerability to food insecurity projected at 1.5°C and 2°C global warming with a higher-resolution global climate model. *Philosophical Transactions of the Royal Society A: Mathematical, Physical and Engineering Sciences*, **376**: 20160452.
- Beven, K. 1997. TOPMODEL: A critique. *Hydrological Processes*, **11**: 1069–1085.
- BGS. 2019. BGS Geology 625k (DiGMapGB-625). [Online]. Available at: https://www.bgs.ac.uk/products/digitalmaps/DiGMapGB\_625.html. Accessed: 5 March 2021.
- Bijl, D.L., Bogaart, P.W., Kram, T., de Vries, B.J.M. & van Vuuren, D.P. 2016. Long-term water demand for electricity, industry and households. *Environmental Science & Policy*, 55: 75– 86.
- Blenkinsop, S. & Fowler, H.J. 2007. Changes in drought frequency, severity and duration for the British Isles projected by the PRUDENCE regional climate models. *Journal of Hydrology*, **342**: 50–71.
- Blunden, J. & Arndt, D.S. 2020. State of the climate in 2019. *Bulletin of the American Meteorological Society*, **101**: S1–S429.
- Boorman, D.B., Hollis, J.M. & Lilly, A. 1995. *Hydrology of soil types: A hydrologically-based classification of the soils of the United Kingdom*. Wallingford, England: Institute of Hydrology.
- Borgomeo, E., Hall, J., Fung, F., Watts, G., Colquhoun, K. & Lambert, C. 2014. Risk-based water resources planning: Incorporating probabilistic nonstationary climate uncertainties. *Water Resources Research*, **50**: 6850–6873.

- Brown, L.E., Mitchell, G., Holden, J., Folkard, A., Wright, N., Beharry-Borg, N., Berry, G., Brierley, B., Chapman, P., Clarke, S.J., Cotton, L., Dobson, M., Dollar, E., Fletcher, M., Foster, J., Hanlon, A., Hildon, S., Hiley, P., Hillis, P., Hoseason, J., Johnston, K., Kay, P., McDonald, A., Parrott, A., Powell, A., Slack, R.J., Sleigh, A., Spray, C., Tapley, K., Underhill, R. & Woulds, C. 2010. Priority water research questions as determined by UK practitioners and policy makers. *Science of The Total Environment*, **409**: 256–266.
- Browne, A., Medd, W. & Anderson, B. 2013. Developing novel approaches to tracking domestic water demand under uncertainty: A reflection on the "up scaling" of social science approaches in the United Kingdom. *Water Resources Management*, 27: 1037– 1038.
- Burt, T.P. & Ferranti, E.J.S. 2012. Changing patterns of heavy rainfall in upland areas: A case study from northern England. *International Journal of Climatology*, **32**: 518–532.
- Burt, T.P., Howden, N.J.K. & Worrall, F. 2016. The changing water cycle: Hydroclimatic extremes in the British Isles. *Wiley Interdisciplinary Reviews: Water*, **3**: 854–870.
- Burt, T.P., Howden, N.J.K., Worrall, F., Whelan, M.J. & Bieroza, M. 2011. Nitrate in United Kingdom rivers: Policy and its outcomes since 1970. *Environmental Science & Technology*, 45: 175–181.
- Bussi, G., Dadson, S.J., Prudhomme, C. & Whitehead, P.G. 2016. Modelling the future impacts of climate and land-use change on suspended sediment transport in the River Thames (UK). *Journal of Hydrology*, **542**: 357–372.
- Bussi, G., Whitehead, P.G., Gutiérrez-Cánovas, C., Ledesma, J.L.J., Ormerod, S.J. & Couture, R.M. 2018. Modelling the effects of climate and land-use change on the hydrochemistry and ecology of the River Wye (Wales). *Science of The Total Environment*, **627**: 733–743.
- Bussi, G., Whitehead, P.G., Thomas, A.R.C., Masante, D., Jones, L., Jack Cosby, B., Emmett, B.A., Malham, S.K., Prudhomme, C. & Prosser, H. 2017. Climate and land-use change impact on faecal indicator bacteria in a temperate maritime catchment (the River Conwy, Wales). *Journal of Hydrology*, **553**: 248–261.
- Butler, D. & Memon, F.A. 2006. *Water demand management*. 1st ed. London, England: IWA Publishing.
- Caissie, D. 2006. The thermal regime of rivers: A review. *Freshwater Biology*, **51**: 1389–1406.
- Callendar, G.S. 1938. The artificial production of carbon dioxide and its influence on temperature. *Quarterly Journal of the Royal Meteorological Society*, **64**: 223–240.
- Capell, R., Tetzlaff, D., Essery, R. & Soulsby, C. 2014. Projecting climate change impacts on stream flow regimes with tracer-aided runoff models Preliminary assessment of heterogeneity at the mesoscale. *Hydrological Processes*, **28**: 545–558.
- Capell, R., Tetzlaff, D. & Soulsby, C. 2013. Will catchment characteristics moderate the projected effects of climate change on flow regimes in the Scottish Highlands? *Hydrological Processes*, **27**: 687–699.
- Carless, D. & Whitehead, P.G. 2013. The potential impacts of climate change on hydropower generation in Mid Wales. *Hydrology Research*, **44**: 495–505.
- CCC. 2020. About the Committee on Climate Change. [Online]. Available at: https://www.theccc.org.uk/about/. Accessed: 5 March 2021.

- Chambers, B.M., Pradhanang, S.M. & Gold, A.J. 2017. Simulating climate change induced thermal stress in coldwater fish habitat using SWAT model. *Water*, **9**: 732.
- Chang, H., Praskievicz, S. & Parandvash, H. 2014. Sensitivity of urban water consumption to weather and climate variability at multiple temporal scales: The case of Portland, Oregon. *International Journal of Geospatial and Environmental Research*, **1**: 7.
- Chapman, S.C., Watkins, N.W. & Stainforth, D.A. 2019. Warming trends in summer heatwaves. *Geophysical Research Letters*, **46**: 1634–1640.
- Charlton, M.B. & Arnell, N.W. 2014. Assessing the impacts of climate change on river flows in England using the UKCP09 climate change projections. *Journal of Hydrology*, **519**: 1723–1738.
- Chaturvedi, D. & Misra, O.P. 2020. Simultaneous effects of the rise in temperature due to greenhouse gases and hypoxia on the dynamics of the aquatic population: A mathematical model. *Journal of Applied Mathematics and Computing*, **63**: 59–85.
- Chitrakar, S., Solemslie, B.W., Neopane, H.P. & Dahlhaug, O.G. 2020. Review on numerical techniques applied in impulse hydro turbines. *Renewable Energy*, **159**: 843–859.
- Cho, K.H., Cha, S.M., Kang, J.H., Lee, S.W., Park, Y., Kim, J.W. & Kim, J.H. 2010. Meteorological effects on the levels of fecal indicator bacteria in an urban stream: A modeling approach. *Water Research*, **44**: 2189–2202.
- Christierson, B., Vidal, J.P. & Wade, S.D. 2012. Using UKCP09 probabilistic climate information for UK water resource planning. *Journal of Hydrology*, **424–425**: 48–67.
- Chun, K.P., Wheater, H.S. & Onof, C. 2009. Streamflow estimation for six UK catchments under future climate scenarios. *Hydrology Research*, **40**: 96–112.
- Clark, K.E., Shanley, J.B., Scholl, M.A., Perdrial, N., Perdrial, J.N., Plante, A.F. & McDowell, W.H. 2017. Tropical river suspended sediment and solute dynamics in storms during an extreme drought. *Water Resources Research*, **53**: 3695–3712.
- Clerc, M. & Kennedy, J. 2002. The particle swarm-explosion, stability, and convergence in a multidimensional complex space. *IEEE Transactions on Evolutionary Computation*, **6**: 58–73.
- Cloke, H.L., Jeffers, C., Wetterhall, F., Byrne, T., Lowe, J.A. & Pappenberger, F. 2010. Climate impacts on river flow: Projections for the Medway Catchment, UK, with UKCP09 and CATCHMOD. *Hydrological Processes*, **24**: 3476–3489.
- Cloke, H.L., Wetterhall, F., He, Y., Freer, J.E. & Pappenberger, F. 2013. Modelling climate impact on floods with ensemble climate projections. *Quarterly Journal of the Royal Meteorological Society*, **139**: 282–297.
- Cobb, B.R. & Sharp, K. V. 2013. Impulse (Turgo and Pelton) turbine performance characteristics and their impact on pico-hydro installations. *Renewable Energy*, **50**: 959–964.
- Coello, C.A.C., Pulido, G.T. & Lechuga, M.S. 2004. Handling multiple objectives with particle swarm optimization. *IEEE Transactions on Evolutionary Computation*, **8**: 256–279.

- Coffey, R., Benham, B., Wolfe, M.L., Dorai-Raj, S., Bhreathnach, N., O'Flaherty, V., Cormican, M. & Cummins, E. 2016. Sensitivity of streamflow and microbial water quality to future climate and land use change in the West of Ireland. *Regional Environmental Change*, 16: 2111–2128.
- Collet, L., Harrigan, S., Prudhomme, C., Formetta, G. & Beevers, L. 2018. Future hot-spots for hydro-hazards in Great Britain: A probabilistic assessment. *Hydrology and Earth System Sciences*, **22**: 5387–5401.
- Conlan, K., Wade, T., Ormerod, S., Lane, S., Durance, I. & Dapeng, Y. 2007. *Preparing for climate change impacts on freshwater ecosystems (PRINCE)*. Bristol, England: Environment Agency.
- Cook, C. 2016. Drought planning as a proxy for water security in England. *Current Opinion in Environmental Sustainability*, **21**: 65–69.
- Cook, J., Nuccitelli, D., Green, S.A., Richardson, M., Winkler, B., Painting, R., Way, R., Jacobs, P. & Skuce, A. 2013. Quantifying the consensus on anthropogenic global warming in the scientific literature. *Environmental Research Letters*, 8: 024024.
- Cook, J., Oreskes, N., Doran, P.T., Anderegg, W.R.L., Verheggen, B., Maibach, E.W., Carlton, J.S., Lewandowsky, S., Skuce, A.G., Green, S.A., Nuccitelli, D., Jacobs, P., Richardson, M., Winkler, B., Painting, R. & Rice, K. 2016. Consensus on consensus: A synthesis of consensus estimates on human-caused global warming. *Environmental Research Letters*, 11: 048002.
- Cosgrove, W.J. & Loucks, D.P. 2015. Water management: Current and future challenges and research directions. *Water Resources Research*, **51**: 4823–4839.
- Coulthard, T.J., Lewin, J. & Macklin, M.G. 2005. Modelling differential catchment response to environmental change. *Geomorphology*, **69**: 222–241.
- Cox, B.A. & Whitehead, P.G. 2009. Impacts of climate change scenarios on dissolved oxygen in the River Thames, UK. *Hydrology Research*, **40**: 138–152.
- Crisp, D.T. & Howson, G. 1982. Effect of air temperature upon mean water temperature in streams in the north Pennines and English Lake District. *Freshwater Biology*, **12**: 359–367.
- Crossman, J., Futter, M.N., Oni, S.K., Whitehead, P.G., Jin, L., Butterfield, D., Baulch, H.M. & Dillon, P.J. 2013. Impacts of climate change on hydrology and water quality: Future proofing management strategies in the Lake Simcoe watershed, Canada. *Journal of Great Lakes Research*, **39**: 19–32.
- Dakhlalla, A.O. & Parajuli, P.B. 2020. Sensitivity of fecal coliform bacteria transport to climate change in an agricultural watershed. *Journal of Water and Climate Change*, **11**: 1250–1262.
- Dallison, R.J.H., Patil, S.D. & Williams, A.P. 2020. Influence of historical climate patterns on streamflow and water demand in Wales, UK. *Water*, **12**: 1684.
- Daniels, M.E. & Danner, E.M. 2020. The drivers of river temperatures below a large dam. *Water Resources Research*, **56**: e2019WR026751.
- da Silva, R.M., Santos, C.A.G., Moreira, M., Corte-Real, J., Silva, V.C.L. & Medeiros, I.C. 2015. Rainfall and river flow trends using Mann–Kendall and Sen's slope estimator statistical tests in the Cobres River basin. *Natural Hazards*, **77**: 1205–1221.

- DCWW. 2015. How our waste water system works. Treharris, Wales: Dŵr Cymru Welsh Water.
- DCWW. 2016a. Trading and procurement code. Treharris, Wales: Dŵr Cymru Welsh Water.
- DCWW. 2016b. Water facts. Treharris, Wales: Dŵr Cymru Welsh Water.
- DCWW. 2016c. *Helping to manage and sustain our environment*. Treharris, Wales: Dŵr Cymru Welsh Water.
- DCWW. 2018. Welsh Water 2050. Treharris, Wales: Dŵr Cymru Welsh Water.
- DCWW. 2019a. *Final water resources management plan 2019*. Treharris, Wales: Dŵr Cymru Welsh Water.
- DCWW. 2019b. Revised draft drought plan 2020. Treharris, Wales: Dŵr Cymru Welsh Water.
- Dearmont, D., McCarl, B.A. & Tolman, D.A. 1998. Costs of water treatment due to diminished water quality: A case study in Texas. *Water Resources Research*, **34**: 849–853.
- de Leeuw, J., Methven, J. & Blackburn, M. 2016. Variability and trends in England and Wales precipitation. *International Journal of Climatology*, **36**: 2823–2836.
- Delpla, I., Jung, A.V., Baures, E., Clement, M. & Thomas, O. 2009. Impacts of climate change on surface water quality in relation to drinking water production. *Environment International*, **35**: 1225–1233.
- Demertzi, K.A., Papamichail, D.M., Georgiou, P.E., Karamouzis, D.N. & Aschonitis, V.G. 2014. Assessment of rural and highly seasonal tourist activity plus drought effects on reservoir operation in a semi-arid region of Greece using the WEAP model. *Water International*, **39**: 23–34.
- Dessai, S. & Hulme, M. 2008. How do UK climate scenarios compare with recent observations? *Atmospheric Science Letters*, **9**: 189–195.
- Diaz-Nieto, J. & Wilby, R.L. 2005. A comparison of statistical downscaling and climate change factor methods: Impacts on low flows in the River Thames, United Kingdom. *Climatic Change*, **69**: 245–268.
- Dixon, H., Lawler, D.M. & Shamseldin, A.Y. 2006. Streamflow trends in western Britain. *Geophysical Research Letters*, **33**: L19406.
- Dobson, B., Coxon, G., Freer, J., Gavin, H., Mortazavi-Naeini, M. & Hall, J.W. 2020. The Spatial dynamics of droughts and water scarcity in England and Wales. *Water Resources Research*, **56**: 2020WR027187.
- Dooge, J.C.I. 1992. Sensitivity of runoff to climate change: A Hortonian approach. *Bulletin of the American Meteorological Society*, **73**: 2013–2024.
- Dore, M.H.I. 2005. Climate change and changes in global precipitation patterns: What do we know? *Environment International*, **31**: 1167–1181.
- Downing, T.E., Butterfield, R.E., Edmunds, B., Knox, J.W., Moss, S., Piper, B.S. & Weatherhead, E.K. 2003. *Climate change and demand for water*. Oxford, England: Stockholm Environment Institute.
- Dunne, T. 1978. Field studies of hillslope flow processes. In: Kirkby, M.J. (ed.). *Hillslope hydrology*. Chichester, England: John Wiley, 227–293.

- Durance, I. & Ormerod, S.J. 2007. Climate change effects on upland stream macroinvertebrates over a 25-year period. *Global Change Biology*, **13**: 942–957.
- Edwards, A.C., Cook, Y., Smart, R. & Wade, A.J. 2000. Concentrations of nitrogen and phosphorus in streams draining the mixed land-use Dee Catchment, north-east Scotland. *Journal of Applied Ecology*, **37**: 159–170.
- EEA. 2012. Corine Land Cover 2012. [Online]. European Environment Agency. Available at: https://www.eea.europa.eu/data-and-maps/data/clc-2012-raster. Accessed: 5 March 2021.
- EEA. 2017. *Copernicus land service pan-European component: CORINE land cover*. Copenhagen, Denmark: European Environment Agency.
- El-Nasr, A.A., Arnold, J.G., Feyen, J. & Berlamont, J. 2005. Modelling the hydrology of a catchment using a distributed and a semi-distributed model. *Hydrological Processes*, **19**: 573–587.
- Ellis, J.R., Milligan, S.P., Readdy, L., Taylor, N. & Brown, M.J. 2012. *Spawning and nursery grounds of selected fish species in UK waters: Science series technical report no. 147*. Lowestoft, England: Cefas.
- Erickson, T.R. & Stefan, H.G. 2000. Linear air/water temperature correlations for streams during open water periods. *Journal of Hydrologic Engineering*, **5**: 317–321.
- Esteve, P., Varela-Ortega, C., Blanco-Gutiérrez, I. & Downing, T.E. 2015. A hydro-economic model for the assessment of climate change impacts and adaptation in irrigated agriculture. *Ecological Economics*, **120**: 49–58.
- European Commission. 2004. *The European Soil Database distribution version 2.0*. [Online]. Available at: https://esdac.jrc.ec.europa.eu/content/european-soil-database-v20-vector-and-attribute-data. Accessed: 5 March 2021.
- European Commission. 2020. *European state of the climate 2019: Summary*. Brussels, Belgium: European Commission.
- FAO. 1998. *World reference base for soil resources*. Rome, Italy: Food and Agriculture Organisation of the United Nations.
- Farrar, J.F. & Vaze, P. 2000. *Wales: Changing climate, challenging choices*. Cardiff, Wales: National Assembly for Wales.
- Fatemi, S.E., Vafaie, F. & Bressers, H. 2013. Assessment of environmental flow requirement effects at an estuary. *Proceedings of the Institution of Civil Engineers Water Management*, **166**: 411–421.
- Fereidoon, M. & Koch, M. 2018. SWAT-MODSIM-PSO optimization of multi-crop planning in the Karkheh River Basin, Iran, under the impacts of climate change. *Science of The Total Environment*, 630: 502–516.
- Ficklin, D.L. & Barnhart, B.L. 2014. SWAT hydrologic model parameter uncertainty and its implications for hydroclimatic projections in snowmelt-dependent watersheds. *Journal of Hydrology*, **519**: 2081–2090.
- Fiquepron, J., Garcia, S. & Stenger, A. 2013. Land use impact on water quality: Valuing forest services in terms of the water supply sector. *Journal of Environmental Management*, **126**: 113–121.

- Flores-López, F., Galaitsi, S., Escobar, M. & Purkey, D. 2016. Modeling of Andean Páramo ecosystems' hydrological response to environmental change. *Water*, **8**: 94.
- Forster, D.L. & Murray, C. 2007. Effects of pesticide use and farming practices on water treatment costs in Maumee river basin communities. In: Hitzhusen, F.J. (ed.). *Economic valuation of river systems*. Cheltenham, England: Edward Elgar Publishing Ltd., 115–128.
- Forzieri, G., Feyen, L., Rojas, R., Flörke, M., Wimmer, F. & Bianchi, A. 2014. Ensemble projections of future streamflow droughts in Europe. *Hydrology and Earth System Sciences*, **18**: 85–108.
- Fosser, G., Kendon, E., Chan, S., Lock, A., Roberts, N. & Bush, M. 2020. Optimal configuration and resolution for the first convection-permitting ensemble of climate projections over the United Kingdom. *International Journal of Climatology*, **40**: 3585–3606.
- Fowler, H.J. & Ekström, M. 2009. Multi-model ensemble estimates of climate change impacts on UK seasonal precipitation extremes. *International Journal of Climatology*, **29**: 385– 416.
- Fowler, H.J. & Kilsby, C.G. 2007. Using regional climate model data to simulate historical and future river flows in northwest England. *Climatic Change*, **80**: 337–367.
- Fowler, H.J. & Wilby, R.L. 2010. Detecting changes in seasonal precipitation extremes using regional climate model projections: Implications for managing fluvial flood risk. *Water Resources Research*, **46**: W03525.
- Garner, G., Hannah, D.M. & Watts, G. 2017. Climate change and water in the UK: Recent scientific evidence for past and future change. *Progress in Physical Geography: Earth and Environment*, **41**: 154–170.
- Gassman, P.W., Sadeghi, A.M. & Srinivasan, R. 2014. Applications of the SWAT model special section: Overview and insights. *Journal of Environmental Quality*, **43**: 1–8.
- Gleick, P.H. & Palaniappan, M. 2010. Peak water limits to freshwater withdrawal and use. *Proceedings of the National Academy of Sciences*, **107**: 11155–11162.
- Gocic, M. & Trajkovic, S. 2013. Analysis of changes in meteorological variables using Mann-Kendall and Sen's slope estimator statistical tests in Serbia. *Global and Planetary Change*, **100**: 172–182.
- Gohar, L., Bernie, D., Good, P. & Lowe, J.A. 2018. UKCP18 derived projections of future climate over the UK. Exeter, England: Met Office Hadley Centre.
- Golmohammadi, G., Prasher, S., Madani, A. & Rudra, R. 2014. Evaluating three hydrological distributed watershed models: MIKE-SHE, APEX, SWAT. *Hydrology*, **1**: 20–39.
- Goodchild, C.W. 2003. Modelling the impact of climate change on domestic water demand. *Water and Environment Journal*, **17**: 8–12.
- Green, M. & Weatherhead, E.K. 2014. Irrigation demand modelling using the UKCP09 weather generator: Lessons learned. *Journal of Water and Climate Change*, **5**: 117–127.
- Gu, G. & Adler, R.F. 2015. Spatial patterns of global precipitation change and variability during 1901–2010. *Journal of Climate*, **28**: 4431–4453.

- Guo, D., Lintern, A., Webb, J.A., Ryu, D., Liu, S., Bende-Michl, U., Leahy, P., Wilson, P. & Western, A.W. 2019. Key factors affecting temporal variability in stream water quality. *Water Resources Research*, **55**: 112–129.
- Gupta, H. V., Kling, H., Yilmaz, K.K. & Martinez, G.F. 2009. Decomposition of the mean squared error and NSE performance criteria: Implications for improving hydrological modelling. *Journal of Hydrology*, **377**: 80–91.
- Gutzler, D.S. & Nims, J.S. 2005. Interannual variability of water demand and summer climate in Albuquerque, New Mexico. *Journal of Applied Meteorology*, **44**: 1777–1787.
- Haguma, D., Leconte, R. & Krau, S. 2017. Hydropower plant adaptation strategies for climate change impacts on hydrological regime. *Canadian Journal of Civil Engineering*, **44**: 962– 970.
- Hajani, E., Rahman, A. & Ishak, E. 2017. Trends in extreme rainfall in the state of New South Wales, Australia. *Hydrological Sciences Journal*, **62**: 2160–2174.
- Hamed, K.H. & Rao, A.R. 1998. A modified Mann-Kendall trend test for autocorrelated data. *Journal of Hydrology*, **204**: 182–196.
- Hannaford, J. 2015. Climate-driven changes in UK river flows: A review of the evidence. *Progress in Physical Geography: Earth and Environment*, **39**: 29–48.
- Hannaford, J. & Buys, G. 2012. Trends in seasonal river flow regimes in the UK. *Journal of Hydrology*, **475**: 158–174.
- Hannaford, J. & Marsh, T.J. 2008. High-flow and flood trends in a network of undisturbed catchments in the UK. *International Journal of Climatology*, **28**: 1325–1338.
- Hannah, D.M. & Garner, G. 2015. River water temperature in the United Kingdom. *Progress in Physical Geography: Earth and Environment*, **39**: 68–92.
- Hänsel, S., Medeiros, D.M., Matschullat, J., Petta, R.A. & Silva, I.D.M. 2016. Assessing homogeneity and climate variability of temperature and precipitation series in the capitals of North-Eastern Brazil. *Frontiers in Earth Science*, **4**: 1–21.
- Hanson, M.J., Keller, A., Boland, M.A. & Lazaus, W.F. 2016. The debate about farm nitrates and drinking water. *Choices*, **31**: 1–7.
- Harkness, C., Semenov, M.A., Areal, F., Senapati, N., Trnka, M., Balek, J. & Bishop, J. 2020. Adverse weather conditions for UK wheat production under climate change. *Agricultural and Forest Meteorology*, **282–283**: 107862.
- Harrigan, S., Hannaford, J., Muchan, K. & Marsh, T.J. 2018. Designation and trend analysis of the updated UK Benchmark Network of river flow stations: The UKBN2 dataset. *Hydrology Research*, **49**: 552–567.
- Harris, I., Jones, P.D., Osborn, T.J. & Lister, D.H. 2014. Updated high-resolution grids of monthly climatic observations: The CRU TS3.10 Dataset. *International Journal of Climatology*, 34: 623–642.
- Harrison, G.P. 2006. Climate change in Scotland: Impact on mini-hydro. SHP News, 23: 15–20.
- Harvey, R., Lye, L., Khan, A. & Paterson, R. 2011. The influence of air temperature on water temperature and the concentration of dissolved oxygen in Newfoundland rivers. *Canadian Water Resources Journal*, **36**: 171–192.

- Hashempour, Y., Nasseri, M., Mohseni-Bandpei, A., Motesaddi, S. & Eslamizadeh, M. 2020. Assessing vulnerability to climate change for total organic carbon in a system of drinking water supply. *Sustainable Cities and Society*, **53**: 101904.
- Hatata, A.Y., El-Saadawi, M.M. & Saad, S. 2019. A feasibility study of small hydro power for selected locations in Egypt. *Energy Strategy Reviews*, **24**: 300–313.
- Hattermann, F.F., Kundzewicz, Z.W., Huang, S., Vetter, T., Gerstengarbe, F.W. & Werner, P.
  2013. Climatological drivers of changes in flood hazard in Germany. *Acta Geophysica*, 61: 463–477.
- Hauer, C., Wagner, B., Aigner, J., Holzapfel, P., Flödl, P., Liedermann, M., Tritthart, M., Sindelar, C., Pulg, U., Klösch, M., Haimann, M., Donnum, B.O., Stickler, M. & Habersack, H. 2018. State of the art, shortcomings and future challenges for a sustainable sediment management in hydropower: A review. *Renewable and Sustainable Energy Reviews*, 98: 40–55.
- Hawkins, D.M. 1977. Testing a sequence of observations for a shift in location. *Journal of the American Statistical Association*, **72**: 180–186.
- Heathwaite, A.L. 2010. Multiple stressors on water availability at global to catchment scales: Understanding human impact on nutrient cycles to protect water quality and water availability in the long term. *Freshwater Biology*, **55**: 241–257.
- Heberling, M.T., Nietch, C.T., Thurston, H.W., Elovitz, M., Birkenhauer, K.H., Panguluri, S., Ramakrishnan, B., Heiser, E. & Neyer, T. 2015. Comparing drinking water treatment costs to source water protection costs using time series analysis. *Water Resources Research*, 51: 8741–8756.
- Helsel, D.R. & Hirsch, R.M. 2002. *Statistical methods in water resources*. Virginia, USA: United States Geological Survey.
- Henriques, C., Garnett, K., Weatherhead, E.K., Lickorish, F.A., Forrow, D. & Delgado, J. 2015. The future water environment: Using scenarios to explore the significant water management challenges in England and Wales to 2050. *Science of The Total Environment*, 512–513: 381–396.
- Herrington, P.R. 1996. *Climate change and the demand for water*. London, England: Her Majesty's Stationery Office.
- Hess, T., Knox, J., Holman, I. & Sutcliffe, C. 2020. Resilience of primary food production to a changing climate: On-farm responses to water-related risks. *Water*, **12**: 2155.
- HM Government. 2017. Water resources planning. [Online]. Available at: https://www.gov.uk/government/publications/water-resources-planning-managingsupply-and-demand#water-resources-management-writing-a-plan. Accessed: 5 March 2021.
- Hofstra, N. 2011. Quantifying the impact of climate change on enteric waterborne pathogen concentrations in surface water. *Current Opinion in Environmental Sustainability*, **3**: 471–479.
- Holden, J., Haygarth, P.M., Dunn, N., Harris, J., Harris, R.C., Humble, A., Jenkins, A., MacDonald, J., McGonigle, D.F., Meacham, T., Orr, H.G., Pearson, P.L., Ross, M., Sapiets, A. & Benton, T. 2017. Water quality and UK agriculture: Challenges and opportunities. *Wiley Interdisciplinary Reviews: Water*, 4: e1201.

- Höllermann, B., Giertz, S. & Diekkrüger, B. 2010. Benin 2025: Balancing future water availability and demand using the WEAP 'Water Evaluation and Planning' system. Water Resources Management, 24: 3591–3613.
- Hornberger, G.M., Wiberg, P.L., Raffensperger, J.P. & D'Odorico, P. 2014. The hillslope stream continuum. In: *Elements of physical hydrology*. Baltimore, USA: Johns Hopkins University Press.
- Horton, D.E., Johnson, N.C., Singh, D., Swain, D.L., Rajaratnam, B. & Diffenbaugh, N.S. 2015. Contribution of changes in atmospheric circulation patterns to extreme temperature trends. *Nature*, **522**: 465–469.
- House of Commons Environmental Audit Committee. 2018. *UK progress on reducing nitrate pollution*. London, England: House of Commons.
- House, W.A., Jickells, T.D., Edwards, A.C., Praska, K.E. & Denison, F.H. 1998. Reactions of phosphorus with sediments in fresh and marine waters. *Soil Use and Management*, **14**: 139–146.
- Huffman, G.J., Adler, R.F., Bolvin, D.T. & Gu, G. 2009. Improving the global precipitation record: GPCP Version 2.1. *Geophysical Research Letters*, **36**: L17808.
- Hutchins, M.G., Williams, R.J., Prudhomme, C., Bowes, M.J., Brown, H.E., Waylett, A.J. & Loewenthal, M. 2016. Projections of future deterioration in UK river quality are hampered by climatic uncertainty under extreme conditions. *Hydrological Sciences Journal*, **61**: 2818–2833.
- IPCC. 2007. Climate change 2007: The physical science basis. Contribution of Working Group I to the Fourth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge, England: Cambridge University Press.
- IPCC. 2013. Climate change 2013: The physical science basis. Contribution of Working Group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge, England: Cambridge University Press.
- IPCC. 2014a. Climate change 2014: Impacts, adaptation, and vulnerability. Part A: Global and sectoral aspects. Contribution of Working Group II to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge, England: Cambridge University Press.
- IPCC. 2014b. Climate change 2014: Synthesis report. Contribution of Working Groups I, II and III to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. Core writing team: Pachauri, R.K. & Meyer, L.A. (eds.). Geneva, Switzerland: IPCC.
- IPCC. 2014c. Climate change 2014: Impacts, adaptation, and vulnerability. Part B: Regional aspects. Contribution of Working Group II to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change. Cambridge, England: Cambridge University Press.
- IPCC. 2018. *Global warming of 1.5°C*. Geneva, Switzerland: Intergovernmental Panel on Climate Change.
- Isaak, D.J., Wollrab, S., Horan, D. & Chandler, G. 2012. Climate change effects on stream and river temperatures across the northwest U.S. from 1980-2009 and implications for salmonid fishes. *Climatic Change*, **113**: 499–524.

- Islam, S.U., Hay, R.W., Déry, S.J. & Booth, B.P. 2019. Modelling the impacts of climate change on riverine thermal regimes in western Canada's largest Pacific watershed. *Scientific Reports*, **9**: 11398.
- Jaiswal, R.K., Lohani, A.K. & Tiwari, H.L. 2015. Statistical analysis for change detection and trend assessment in climatological parameters. *Environmental Processes*, **2**: 729–749.
- Janes, V.J.J., Nicholas, A.P., Collins, A.L. & Quine, T.A. 2017. Analysis of fundamental physical factors influencing channel bank erosion: Results for contrasting catchments in England and Wales. *Environmental Earth Sciences*, **76**: 307.
- Jenkins, G.J., Perry, M.C. & Prior, M.J. 2008. *The climate of the United Kingdom and recent trends*. Exeter, England: Met Office Hadley Centre.
- Jenkins, K., Surminski, S., Hall, J. & Crick, F. 2017. Assessing surface water flood risk and management strategies under future climate change: Insights from an Agent-Based Model. *Science of The Total Environment*, **595**: 159–168.
- Jennings, E., Allott, N., Pierson, D.C., Schneiderman, E.M., Lenihan, D., Samuelsson, P. & Taylor, D. 2009. Impacts of climate change on phosphorus loading from a grassland catchment: Implications for future management. *Water Research*, **43**: 4316–4326.
- Jilo, N.B., Gebremariam, B., Harka, A.E., Woldemariam, G.W. & Behulu, F. 2019. Evaluation of the impacts of climate change on sediment yield from the Logiya Watershed, Lower Awash Basin, Ethiopia. *Hydrology*, 6: 81.
- Jin, J., Wang, G., Zhang, J., Yang, Q., Liu, C., Liu, Y., Bao, Z. & He, R. 2020. Impacts of climate change on hydrology in the Yellow River source region, China. *Journal of Water and Climate Change*, **11**: 916–930.
- Jin, L., Whitehead, P.G., Futter, M.N. & Lu, Z. 2012. Modelling the impacts of climate change on flow and nitrate in the River Thames: Assessing potential adaptation strategies. *Hydrology Research*, **43**: 902–916.
- Johnson, A.C., Acreman, M.C., Dunbar, M.J., Feist, S.W., Giacomello, A.M., Gozlan, R.E., Hinsley, S.A., Ibbotson, A.T., Jarvie, H.P., Jones, J.I., Longshaw, M., Maberly, S.C., Marsh, T.J., Neal, C., Newman, J.R., Nunn, M.A., Pickup, R.W., Reynard, N.S., Sullivan, C.A., Sumpter, J.P. & Williams, R.J. 2009. The British river of the future: How climate change and human activity might affect two contrasting river ecosystems in England. *Science of The Total Environment*, **407**: 4787–4798.
- Johnson, T., Butcher, J., Deb, D., Faizullabhoy, M., Hummel, P., Kittle, J., McGinnis, S., Mearns, L.O., Nover, D., Parker, A., Sarkar, S., Srinivasan, R., Tuppad, P., Warren, M., Weaver, C. & Witt, J. 2015. Modeling streamflow and water quality sensitivity to climate change and urban development in 20 U.S. watersheds. *Journal of the American Water Resources Association*, **51**: 1321–1341.
- Jones, C.S., Hill, D. & Brand, G. 2007. Use a multifaceted approach to manage high sourcewater nitrate. *Opflow*, **33**: 20–22.
- Jones, M.R., Fowler, H.J., Kilsby, C.G. & Blenkinsop, S. 2013. An assessment of changes in seasonal and annual extreme rainfall in the UK between 1961 and 2009. *International Journal of Climatology*, **33**: 1178–1194.

- Joyce, B.A., Mehta, V.K., Purkey, D.R., Dale, L.L. & Hanemann, M. 2011. Modifying agricultural water management to adapt to climate change in California's central valley. *Climatic Change*, **109**: 299–316.
- Kakouei, K., Kiesel, J., Domisch, S., Irving, K.S., Jähnig, S.C. & Kail, J. 2018. Projected effects of climate-change-induced flow alterations on stream macroinvertebrate abundances. *Ecology and Evolution*, **8**: 3393–3409.
- Kauffeldt, A., Wetterhall, F., Pappenberger, F., Salamon, P. & Thielen, J. 2016. Technical review of large-scale hydrological models for implementation in operational flood forecasting schemes on continental level. *Environmental Modelling & Software*, **75**: 68– 76.
- Kay, A.L., Bell, V.A., Guillod, B.P., Jones, R.G. & Rudd, A.C. 2018. National-scale analysis of low flow frequency: Historical trends and potential future changes. *Climatic Change*, **147**: 585–599.
- Kay, A.L., Crooks, S.M., Davies, H.N., Prudhomme, C. & Reynard, N.S. 2014a. Probabilistic impacts of climate change on flood frequency using response surfaces I: England and Wales. *Regional Environmental Change*, 14: 1215–1227.
- Kay, A.L., Crooks, S.M., Davies, H.N. & Reynard, N.S. 2014b. Probabilistic impacts of climate change on flood frequency using response surfaces II: Scotland. *Regional Environmental Change*, 14: 1243–1255.
- Kay, A.L., Watts, G., Wells, S.C. & Allen, S. 2020. The impact of climate change on UK river flows: A preliminary comparison of two generations of probabilistic climate projections. *Hydrological Processes*, **34**: 1081–1088.
- Keller, V.D.J., Tanguy, M., Prosdocimi, I., Terry, J.A., Hitt, O., Cole, S.J., Fry, M., Morris, D.G. & Dixon, H. 2015. CEH-GEAR: 1 km resolution daily and monthly areal rainfall estimates for the UK for hydrological and other applications. *Earth System Science Data*, **7**: 143–155.
- Kendall, M.G. 1975. Rank correlation methods. 3rd ed. London, England: Griffin Publishers.
- Kendon, E.J., Fosser, G., Murphy, J.M., Chan, S., Clark, R.T., Harris, G.R., Lock, A., Lowe, J.A., Martin, G., Pirret, J., Roberts, N.M., Sanderson, M.G. & Tucker, S. 2019. UKCP convectionpermitting model projections: Science report. Exeter, England: Met Office Hadley Centre.
- Kendon, M., McCarthy, M., Jevrejeva, S., Matthews, A., Sparks, T. & Garforth, J. 2020. State of the UK climate 2019. *International Journal of Climatology*, **40**: 1–69.
- Kennedy, J. & Eberhart, R.C. 1995. Particle swarm optimization. In: Proceedings of ICNN'95 -International Conference on Neural Networks. Perth, Australia: Institute of Electrical and Electronics Engineers, 1942–1948.
- Khan, A.J., Koch, M. & Tahir, A.A. 2020. Impacts of climate change on the water availability, seasonality and extremes in the Upper Indus Basin (UIB). *Sustainability*, **12**: 1283.
- Killingtveit, Å. 2019. Hydropower. In: Letcher, T.M. (ed.). *Managing global warming. An interface of technology and human issues*. Cambridge, USA: Academic Press, 265–315.
- King, A.D. & Karoly, D.J. 2017. Climate extremes in Europe at 1.5 and 2 degrees of global warming. *Environmental Research Letters*, **12**: 114031.
- Knox, J., Morris, J. & Hess, T. 2010. Identifying future risks to UK agricultural crop production. *Outlook on Agriculture*, **39**: 249–256.

- Konapala, G., Mishra, A.K., Wada, Y. & Mann, M.E. 2020. Climate change will affect global water availability through compounding changes in seasonal precipitation and evaporation. *Nature Communications*, **11**: 3044.
- Kumar, P., Masago, Y., Mishra, B., Jalilov, S., Rafiei Emam, A., Kefi, M. & Fukushi, K. 2017. Current assessment and future outlook for water resources considering climate change and a population burst: A case study of Ciliwung River, Jakarta City, Indonesia. *Water*, **9**: 410.
- Kundzewics, Z.W. & Robson, A.J. 2004. Change detection in hydrological records A review of the methodology. *Hydrological Sciences Journal*, **49**: 7–19.
- Kundzewicz, Z.W., Pińskwar, I. & Brakenridge, G.R. 2018. Changes in river flood hazard in Europe: A review. *Hydrology Research*, **49**: 294–302.
- Laizé, C.L.R. & Hannah, D.M. 2010. Modification of climate-river flow associations by basin properties. *Journal of Hydrology*, **389**: 186–204.
- Langhammer, J. & Bernsteinová, J. 2020. Which aspects of hydrological regime in mid-latitude montane basins are affected by climate change? *Water*, **12**: 2279.
- Lavers, D.A., Allan, R.P., Villarini, G., Lloyd-Hughes, B., Brayshaw, D.J. & Wade, A.J. 2013. Future changes in atmospheric rivers and their implications for winter flooding in Britain. *Environmental Research Letters*, **8**: 034010.
- Lee, K.H., Baek, S.W. & Kim, K.W. 2008. Inverse radiation analysis using repulsive particle swarm optimization algorithm. *International Journal of Heat and Mass Transfer*, **51**: 2772–2783.
- Lee, S., Yeo, I.Y., Sadeghi, A.M., McCarty, G.W., Hively, W.D., Lang, M.W. & Sharifi, A. 2018. Comparative analyses of hydrological responses of two adjacent watersheds to climate variability and change using the SWAT model. *Hydrology and Earth System Sciences*, 22: 689–708.
- Liang, X., Lettenmaier, D.P., Wood, E.F. & Burges, S.J. 1994. A simple hydrologically based model of land surface water and energy fluxes for general circulation models. *Journal of Geophysical Research*, **99**: 14415–14428.
- Lilienthal, P., Lambert, T. & Gilman, P. 2004. Computer modeling of renewable power systems. In: Cleveland, C.J. (ed.). *Encyclopedia of Energy*. Amsterdam, Netherlands: Elsevier, 633–647.
- Liu, Z., Anderson, B., Yan, K., Dong, W., Liao, H. & Shi, P. 2017. Global and regional changes in exposure to extreme heat and the relative contributions of climate and population change. *Scientific Reports*, **7**: 43909.
- Loperfido, J.V. 2014. Surface water quality in streams and rivers: Scaling and climate change. In: Ahuja, S. (ed.). *Comprehensive water quality and purification*. Amsterdam, Netherlands: Elsevier, 87–105.
- Lowe, J.A., Bernie, D., Bett, P., Bricheno, L., Brown, S.J., Calvert, D., Clark, R.T., Eagle, K.E., Edwards, T., Fosser, G., Fung, F., Gohar, L., Good, P., Gregory, J., Harris, G.R., Howard, T.P., Kaye, N., Kendon, E.J., Krijnen, J., Maisey, P., McDonald, R., McInnes, R.N., McSweeney, C.F., Mitchell, J.F.B., Murphy, J.M., Palmer, M., Roberts, C., Rostron, J.W., Sexton, D.M.H., Thornton, H.E., Tinker, J., Tucker, S., Yamazaki, K. & Belcher, S. 2018. UKCP18 science overview report. Exeter, England: Met Office Hadley Centre.

- Luterbacher, J. 2004. European seasonal and annual temperature variability, trends, and extremes since 1500. *Science*, **303**: 1499–1503.
- Lynch, A.J., Myers, B.J.E., Chu, C., Eby, L.A., Falke, J.A., Kovach, R.P., Krabbenhoft, T.J., Kwak,
   T.J., Lyons, J., Paukert, C.P. & Whitney, J.E. 2016. Climate change effects on North
   American inland fish populations and assemblages. *Fisheries*, **41**: 346–361.
- Macdonald, N., Phillips, I.D. & Mayle, G. 2010. Spatial and temporal variability of flood seasonality in Wales. *Hydrological Processes*, **24**: 1806–1820.
- Mann, H.B. 1945. Nonparametric tests against trend. *Econometrica*, **13**: 245–259.
- Martz, L.W. & Garbrecht, J. 1992. Numerical definition of drainage network and subcatchment areas from digital elevation models. *Computers & Geosciences*, **18**: 747–761.
- Marx, A., Kumar, R., Thober, S., Rakovec, O., Wanders, N., Zink, M., Wood, E.F., Pan, M., Sheffield, J. & Samaniego, L. 2018. Climate change alters low flows in Europe under global warming of 1.5, 2, and 3 °C. *Hydrology and Earth System Sciences*, **22**: 1017–1032.
- Massoud, E.C., Purdy, A.J., Miro, M.E. & Famiglietti, J.S. 2018. Projecting groundwater storage changes in California's Central Valley. *Scientific Reports*, **8**: 12917.
- Mayes, J. 2000. Changing regional climatic gradients in the United Kingdom. *The Geographical Journal*, **166**: 125–138.
- McCartney, M.P. & Menker Girma, M. 2012. Evaluating the downstream implications of planned water resource development in the Ethiopian portion of the Blue Nile River. *Water International*, **37**: 362–379.
- McDonald, R. & Shemie, D. 2014. *Urban water blueprint: Mapping conservation solutions to the global water challenge*. Washington DC, USA: The Nature Conservancy.
- McGrane, S.J. 2016. Impacts of urbanisation on hydrological and water quality dynamics, and urban water management: A review. *Hydrological Sciences Journal*, **61**: 2295–2311.
- Met Office. 2019. *UK climate projections: Headline findings*. Exeter, England: Met Office Hadley Centre.
- Met Office. 2020. UK climate averages. [Online]. Available at: http://www.metoffice.gov.uk/public/weather/climate#?region=wales. Accessed: 5 March 2021.
- Michalak, A.M. 2016. Study role of climate change in extreme threats to water quality. *Nature*, **535**: 349–350.
- Miller, J.D. & Hutchins, M. 2017. The impacts of urbanisation and climate change on urban flooding and urban water quality: A review of the evidence concerning the United Kingdom. *Journal of Hydrology: Regional Studies*, **12**: 345–362.
- Mishra, V., Ganguly, A.R., Nijssen, B. & Lettenmaier, D.P. 2015. Changes in observed climate extremes in global urban areas. *Environmental Research Letters*, **10**: 024005.
- Mo, W., Wang, H. & Jacobs, J.M. 2016. Understanding the influence of climate change on the embodied energy of water supply. *Water Research*, **95**: 220–229.

- Mockler, E.M., Deakin, J., Archbold, M., Gill, L., Daly, D. & Bruen, M. 2017. Sources of nitrogen and phosphorus emissions to Irish rivers and coastal waters: Estimates from a nutrient load apportionment framework. *Science of The Total Environment*, **601–602**: 326–339.
- Mohammadi, M., Finnan, J., Baker, C. & Sterling, M. 2020. The potential impact of climate change on oat lodging in the UK and Republic of Ireland. *Advances in Meteorology*, **2020**: 1–16.
- MOHC. 2018a. *State of the UK climate 2017: Supplementary report on climate extremes*. Exeter, England: Met Office Hadley Centre.
- MOHC. 2018b. UKCP18 regional projections on a 12km grid over the UK for 1980-2080. [Online]. Available at: https://catalogue.ceda.ac.uk/uuid/589211abeb844070a95d061c8cc7f604. Accessed: 5 March 2021.
- MOHC. 2019a. UKCP18 factsheet: Temperature. Exeter, England: Met Office Hadley Centre.
- MOHC. 2019b. UKCP18 factsheet: Precipitation. Exeter, England: Met Office Hadley Centre.
- Momblanch, A., Beevers, L., Srinivasalu, P., Kulkarni, A. & Holman, I.P. 2020. Enhancing production and flow of freshwater ecosystem services in a managed Himalayan river system under uncertain future climate. *Climatic Change*.
- Morrill, J.C., Bales, R.C. & Conklin, M.H. 2005. Estimating stream temperature from air temperature: Implications for future water quality. *Journal of Environmental Engineering*, **131**: 139–146.
- Mortazavi-Naeini, M., Bussi, G., Elliott, J.A., Hall, J. & Whitehead, P.G. 2019. Assessment of risks to public water supply from low flows and harmful water quality in a changing climate. *Water Resources Research*, **55**: 10386–10404.
- Mosheim, R. & Ribaudo, M. 2017. Costs of nitrogen runoff for rural water utilities: A shadow cost approach. *Land Economics*, **93**: 12–39.
- Moss, R.H., Edmonds, J.A., Hibbard, K.A., Manning, M.R., Rose, S.K., van Vuuren, D.P., Carter, T.R., Emori, S., Kainuma, M., Kram, T., Meehl, G.A., Mitchell, J.F.B., Nakicenovic, N., Riahi, K., Smith, S.J., Stouffer, R.J., Thomson, A.M., Weyant, J.P. & Wilbanks, T.J. 2010. The next generation of scenarios for climate change research and assessment. *Nature*, 463: 747–756.
- Mudbhatkal, A. & Amai, M. 2018. Regional climate trends and topographic influence over the Western Ghat catchments of India. *International Journal of Climatology*, **38**: 2265–2279.
- Murphy, C., Harrigan, S., Hall, J. & Wilby, R.L. 2013. Climate-driven trends in mean and high flows from a network of reference stations in Ireland. *Hydrological Sciences Journal*, **58**: 755–772.
- Murphy, J.M., Booth, B.B.B., Collins, M., Harris, G.R., Sexton, D.M.H. & Webb, M.J. 2007. A methodology for probabilistic predictions of regional climate change from perturbed physics ensembles. *Philosophical Transactions of the Royal Society A: Mathematical, Physical and Engineering Sciences*, **365**: 1993–2028.

- Murphy, J.M., Harris, G.R., Sexton, D.M.H., Kendon, E.J., Bett, P., Clark, R.T., Eagle, K.E., Fosser, G., Fung, F., Lowe, J.A., McDonald, R., McInnes, R.N., McSweeney, C.F., Mitchell, J.F.B., Rostron, J.W., Thornton, H.E., Tucker, S. & Yamazaki, K. 2018. UKCP18 land projections: Science report. Exeter, England: Met Office Hadley Centre.
- Murphy, J.M., Sexton, D.M.H., Jenkins, G.J., Boorman, P.M., Booth, B.B.B., Brown, C.C., Clark, R.T., Collins, M., Harris, G.R., Kendon, E.J., Betts, R.A., Brown, S.J., Howard, T.P., Humphrey, K.A., McCarthy, M.P., McDonald, R., Stephens, A., Wallace, C., Warren, R., Wilby, R.L. & Wood, R.A. 2009. UK climate projections science report: Climate change projections. Exeter, England: Met Office Hadley Centre.
- Murphy, T., Hanley, M., Ellis, J. & Lunt, P. 2019. Deviation between projected and observed precipitation trends greater with altitude. *Climate Research*, **79**: 77–89.
- Murray, C. 2001. A study of pesticide use, farming practices, and community drinking water treatment costs in the Maumee and Great Lakes Basins. Columbus, USA: Ohio State University.
- Mwangi, H.M., Julich, S., Patil, S.D., McDonald, M.A. & Feger, K.H. 2016. Relative contribution of land use change and climate variability on discharge of upper Mara River, Kenya. *Journal of Hydrology: Regional Studies*, **5**: 244–260.
- Naidu, B.S.K. 1999. Developing silt consciousness in the minds of hydro power engineers. In: Varma, C.V.J., Naidu, B.S.K. & Rao, A.R.G. (eds.). *Proceedings of 1st International Conference on Silting Problems in Hydropower Plants*. New Delhi, India: Central Board of Irrigation & Power, 1–36.
- Neitsch, S.L., Arnold, J.G., Kiniry, J.R. & Williams, J.R. 2011. *Soil and Water Assessment Tool theoretical decumentation: Version 2009.* Texas, USA: Texas Water Resources Institute.
- Neopane, H.P. 2010. *Sediment erosion in hydro turbines*. Trondheim, Norway: Norwegian University of Science and Technology.
- Nerantzaki, S.D., Giannakis, G. V., Nikolaidis, N.P., Zacharias, I., Karatzas, G.P. & Sibetheros,
   I.A. 2016. Assessing the impact of climate change on sediment loads in a large Mediterranean watershed. *Soil Science*, **181**: 306–314.
- Nesbitt, A., Kemp, B., Steele, C., Lovett, A. & Dorling, S. 2016. Impact of recent climate change and weather variability on the viability of UK viticulture - combining weather and climate records with producers' perspectives. *Australian Journal of Grape and Wine Research*, 22: 324–335.
- NIC. 2018. *Preparing for a drier future: England's water infrastructure needs*. London, England: National Infrastructure Committee.
- Nilawar, A.P. & Waikar, M.L. 2019. Impacts of climate change on streamflow and sediment concentration under RCP 4.5 and 8.5: A case study in Purna River Basin, India. *Science of The Total Environment*, **650**: 2685–2696.
- Nilsson, C. & Malm Renöfält, B. 2008. Linking flow regime and water quality in rivers: A challenge to adaptive catchment managment. *Ecology and Society*, **13**: 1–20.
- Novara, D. & McNabola, A. 2018. A model for the extrapolation of the characteristic curves of pumps as turbines from a datum best efficiency point. *Energy Conversion and Management*, **174**: 1–7.

- NRFA. 2020. *National River Flow Archive*. [Online]. Available at: https://nrfa.ceh.ac.uk/data/search. Accessed: 5 March 2021.
- NRW. 2019a. *Reasons for not achieving good status*. [Online]. Available at: https://waterwatchwales.naturalresourceswales.gov.uk/en/. Accessed: 5 March 2021.
- NRW. 2019b. Natural Resources Wales (NRW) Water Quality Data Archive held in WISKI (KiWQM). [Online]. Available at: https://libcat.naturalresources.wales/webview/?infile=details.glu&loid=119251&rs=113 6580&hitno=1&straight to details=TRUE&tiarray=full. Accessed: 5 March 2021.
- NRW. 2019c. *Hydropower permits*. [Online]. Available at: https://lle.gov.wales/catalogue/item/HydropowerPermits/?lang=en. Accessed: 5 March 2021.
- NRW. 2020a. Water management. [Online]. Available at: https://naturalresources.wales/about-us/what-we-do/water/resources/watermanagement. Accessed: 5 March 2021.
- NRW. 2020b. HGN 2 hydropower flow standards. Cardiff, Wales: Natural Resources Wales.
- Ockenden, M.C., Hollaway, M.J., Beven, K.J., Collins, A.L., Evans, R., Falloon, P.D., Forber, K.J., Hiscock, K.M., Kahana, R., Macleod, C.J.A., Tych, W., Villamizar, M.L., Wearing, C., Withers, P.J.A., Zhou, J.G., Barker, P.A., Burke, S., Freer, J.E., Johnes, P.J., Snell, M.A., Surridge, B.W.J. & Haygarth, P.M. 2017. Major agricultural changes required to mitigate phosphorus losses under climate change. *Nature Communications*, 8: 161.
- Ordnance Survey. 2017. OS Terrain 5: User guide and technical specification. Southampton, England: Ordnance Survey.
- Ordnance Survey. 2020. OS Terrain 5. [Online]. Available at: https://www.ordnancesurvey.co.uk/business-government/products/terrain-5. Accessed: 5 March 2021.
- Oreskes, N. 2018. The scientific consensus on climate change: How do we know we're not wrong? In: Lloyd, A. & Winsberg, E. (eds.). *Climate Modelling*. Cham, Switzerland: Palgrave Macmilan, 31–64.
- Osborn, T.J. 2011. Winter 2009/2010 temperatures and a record-breaking North Atlantic Oscillation index. *Weather*, **66**: 19–21.
- Osborn, T.J. & Hulme, M. 2002. Evidence for trends in heavy rainfall events over the UK. *Philosophical Transactions of the Royal Society of London. Series A: Mathematical, Physical and Engineering Sciences*, **360**: 1313–1325.
- Padhy, M.K. & Saini, R.P. 2008. A review on silt erosion in hydro turbines. *Renewable and Sustainable Energy Reviews*, **12**: 1974–1987.
- Padhy, M.K. & Saini, R.P. 2011. Study of silt erosion on performance of a Pelton turbine. *Energy*, **36**: 141–147.
- Paish, O. 2002. Micro-hydropower: Status and prospects. *Proceedings of the Institution of Mechanical Engineers, Part A: Journal of Power and Energy*, **216**: 31–40.
- Papadimitriou, L. V., Koutroulis, A.G., Grillakis, M.G. & Tsanis, I.K. 2016. High-end climate change impact on European runoff and low flows exploring the effects of forcing biases. *Hydrology and Earth System Sciences*, **20**: 1785–1808.

- Papalexiou, S.M., Aghakouchak, A., Trenberth, K.E. & Foufoula-Georgiou, E. 2018. Global, regional, and megacity trends in the highest temperature of the year: Diagnostics and evidence for accelerating trends. *Earth's Future*, **6**: 71–79.
- Papalexiou, S.M. & Montanari, A. 2019. Global and regional increase of precipitation extremes under global warming. *Water Resources Research*, **55**: 2018WR024067.
- Park, J.Y. & Kim, S.J. 2014. Potential impacts of climate change on the reliability of water and hydropower supply from a multipurpose dam in South Korea. *Journal of the American Water Resources Association*, **50**: 1273–1288.
- Parker, J. & Wilby, R.L. 2013. Quantifying household water demand: A review of theory and practice in the UK. *Water Resources Management*, **27**: 981–1011.
- Perkins-Kirkpatrick, S.E. & Lewis, S.C. 2020. Increasing trends in regional heatwaves. *Nature Communications*, **11**: 3357.
- Perra, E., Piras, M., Deidda, R., Paniconi, C., Mascaro, G., Vivoni, E.R., Cau, P., Marras, P.A., Ludwig, R. & Meyer, S. 2018. Multimodel assessment of climate change-induced hydrologic impacts for a Mediterranean catchment. *Hydrology and Earth System Sciences*, 22: 4125–4143.
- Perry, M. 2006. A spatial analysis of trends in the UK climate since 1914 using gridded datasets. Exeter, England: Met Office.
- Pesce, M., Critto, A., Torresan, S., Giubilato, E., Santini, M., Zirino, A., Ouyang, W. & Marcomini, A. 2018. Modelling climate change impacts on nutrients and primary production in coastal waters. *Science of The Total Environment*, **628–629**: 919–937.
- Pettitt, A.N. 1979. A non-parametric approach to the change-point problem. *Applied Statistics*, **28**: 126.
- Preud'homme, E.B. & Stefan, H.G. 1992. *Relationship between water temperatures and air temperatures for central US streams*. Minneapolis, USA: University of Minnesota.
- Price, J.I. & Heberling, M.T. 2018. The effects of source water quality on drinking water treatment costs: A review and synthesis of empirical literature. *Ecological Economics*, **151**: 195–209.
- Prudhomme, C., Crooks, S., Kay, A.L. & Reynard, N. 2013a. Climate change and river flooding: Part 1 classifying the sensitivity of British catchments. *Climatic Change*, **119**: 933–948.
- Prudhomme, C., Haxton, T., Crooks, S., Jackson, C., Barkwith, A., Williamson, J., Kelvin, J., Mackay, J., Wang, L., Young, A. & Watts, G. 2013b. Future Flows Hydrology: An ensemble of daily river flow and monthly groundwater levels for use for climate change impact assessment across Great Britain. *Earth System Science Data*, 5: 101–107.
- Prudhomme, C., Jakob, D. & Svensson, C. 2003. Uncertainty and climate change impact on the flood regime of small UK catchments. *Journal of Hydrology*, **277**: 1–23.
- Prudhomme, C., Kay, A.L., Crooks, S. & Reynard, N. 2013c. Climate change and river flooding: Part 2 sensitivity characterisation for British catchments and example vulnerability assessments. *Climatic Change*, **119**: 949–964.

- Prudhomme, C., Young, A., Watts, G., Haxton, T., Crooks, S., Williamson, J., Davies, H., Dadson, S. & Allen, S. 2012. The drying up of Britain? A national estimate of changes in seasonal river flows from 11 Regional Climate Model simulations. *Hydrological Processes*, 26: 1115–1118.
- Pulley, S. & Collins, A.L. 2019. Field-based determination of controls on runoff and fine sediment generation from lowland grazing livestock fields. *Journal of Environmental Management*, 249: 109365.
- Purkey, D.R., Joyce, B., Vicuna, S., Hanemann, M.W., Dale, L.L., Yates, D. & Dracup, J.A. 2008.
   Robust analysis of future climate change impacts on water for agriculture and other sectors: A case study in the Sacramento Valley. *Climatic Change*, 87: 109–122.
- Qi, H., Niu, C.Y., Gong, S., Ren, Y.T. & Ruan, L.M. 2015. Application of the hybrid particle swarm optimization algorithms for simultaneous estimation of multi-parameters in a transient conduction–radiation problem. *International Journal of Heat and Mass Transfer*, 83: 428– 440.
- Qiao, L., Pan, Z., Herrmann, R.B. & Hong, Y. 2014. Hydrological variability and uncertainty of lower Missouri River Basin under changing climate. *Journal of the American Water Resources Association*, **50**: 246–260.
- Rahiz, M. & New, M. 2013. 21st century drought scenarios for the UK. *Water Resources Management*, **27**: 1039–1061.
- Raskin, P., Hansen, E., Zhu, Z. & Stavisky, D. 1992. Simulation of water supply and demand in the Aral Sea region. *Water International*, **17**: 55–67.
- Rau, M., He, Y., Goodess, C. & Bárdossy, A. 2020. Statistical downscaling to project extreme hourly precipitation over the United Kingdom. *International Journal of Climatology*, **40**: 1805–1823.
- Ravansalar, M., Rajaee, T. & Ergil, M. 2016. Prediction of dissolved oxygen in River Calder by noise elimination time series using wavelet transform. *Journal of Experimental & Theoretical Artificial Intelligence*, **28**: 689–706.
- Refsgaard, J.C. & Storm, B. 1995. MIKE SHE. In: Singh, V.P. (ed.). *Computer models of watershed hydrology*. Colorado, USA: Water Resources Publications, 809–847.
- Rehana, S. & Mujumdar, P.P. 2012. Climate change induced risk in water quality control problems. *Journal of Hydrology*, **444–445**: 63–77.
- Remesan, R., Bellerby, T. & Frostick, L. 2014. Hydrological modelling using data from monthly GCMs in a regional catchment. *Hydrological Processes*, **28**: 3241–3263.
- Riahi, K., Rao, S., Krey, V., Cho, C., Chirkov, V., Fischer, G., Kindermann, G., Nakicenovic, N. & Rafaj, P. 2011. RCP 8.5: A scenario of comparatively high greenhouse gas emissions. *Climatic Change*, **109**: 33–57.
- Rio, M., Rey, D., Prudhomme, C. & Holman, I.P. 2018. Evaluation of changing surface water abstraction reliability for supplemental irrigation under climate change. *Agricultural Water Management*, **206**: 200–208.

- Ritson, J.P., Bell, M., Graham, N.J.D., Templeton, M.R., Brazier, R.E., Verhoef, A., Freeman, C. & Clark, J.M. 2014. Simulated climate change impact on summer dissolved organic carbon release from peat and surface vegetation: Implications for drinking water treatment. *Water Research*, 67: 66–76.
- Rivas-Tabares, D., Tarquis, A.M., Willaarts, B. & De Miguel, Á. 2019. An accurate evaluation of water availability in sub-arid Mediterranean watersheds through SWAT: Cega-Eresma-Adaja. *Agricultural Water Management*, **212**: 211–225.
- Robinson, E.L., Blyth, E., Clark, D.B., Comyn-Platt, E., Finch, J. & Rudd, A.C. 2017. *Climate hydrology and ecology research support system meteorology dataset for Great Britain (1961-2015) [CHESS-met]*. Wallingford, England: Centre for Ecology & Hydrology.
- Robinson, M., Cognard-Plancq, A.L., Cosandey, C., David, J., Durand, P., Führer, H.W., Hall, R., Hendriques, M.O., Marc, V., McCarthy, R., McDonnell, M., Martin, C., Nisbet, T., O'Dea, P., Rodgers, M. & Zollner, A. 2003. Studies of the impact of forests on peak flows and baseflows: A European perspective. *Forest Ecology and Management*, **186**: 85–97.
- Rockström, J., Steffen, W., Noone, K., Persson, Å., Chapin, F.S., Lambin, E.F., Lenton, T.M., Scheffer, M., Folke, C., Schellnhuber, H.J., Nykvist, B., de Wit, C.A., Hughes, T., van der Leeuw, S., Rodhe, H., Sörlin, S., Snyder, P.K., Costanza, R., Svedin, U., Falkenmark, M., Karlberg, L., Corell, R.W., Fabry, V.J., Hansen, J., Walker, B., Liverman, D., Richardson, K., Crutzen, P. & Foley, J.A. 2009. A safe operating space for humanity. *Nature*, 461: 472–475.
- Rostron, J.W., Sexton, D.M.H., McSweeney, C.F., Yamazaki, K., Andrews, T., Furtado, K., Ringer, M.A. & Tsushima, Y. 2020. The impact of performance filtering on climate feedbacks in a perturbed parameter ensemble. *Climate Dynamics*, **55**: 521–551.
- Rudd, A.C., Kay, A.L. & Bell, V.A. 2019. National-scale analysis of future river flow and soil moisture droughts: Potential changes in drought characteristics. *Climatic Change*, **156**: 323–340.
- Rumsby, B.T. & Macklin, M.G. 1994. Channel and floodplain response to recent abrupt climate change: The Tyne Basin, Northern England. *Earth Surface Processes and Landforms*, **19**: 499–515.
- Sample, J.E., Duncan, N., Ferguson, M. & Cooksley, S. 2015. Scotland's hydropower: Current capacity, future potential and the possible impacts of climate change. *Renewable and Sustainable Energy Reviews*, **52**: 111–122.
- Sanderson, M.G., Wiltshire, A.J. & Betts, R.A. 2012. Projected changes in water availability in the United Kingdom. *Water Resources Research*, **48**: W08512.
- Sankarasubramanian, A., Vogel, R.M. & Limbrunner, J.F. 2001. Climate elasticity of streamflow in the United States. *Water Resources Research*, **37**: 1771–1781.
- Sawicz, K., Wagener, T., Sivapalan, M., Troch, P.A. & Carrillo, G. 2011. Catchment classification: Empirical analysis of hydrologic similarity based on catchment function in the eastern USA. *Hydrology and Earth System Sciences*, **15**: 2895–2911.
- Saxton, K.E. & Rawls, W.J. 2006. Soil water characteristic estimates by texture and organic matter for hydrologic solutions. *Soil Science Society of America Journal*, **70**: 1569–1578.

- Sayers, P., Horritt, M., Penning-Rowsell, E. & McKenzie, A. 2015. *Climate change risk assessment 2017: Projections of future flood risk in the UK*. London, England: Committee on Climate Change.
- Sayers, P., Horritt, M., Penning-Rowsell, E., McKenzie, A. & Thompson, D. 2016. The analysis of future flood risk in the UK using the Future Flood Explorer. *E3S Web of Conferences*, **7**: 21005.
- Schaake, J.S. 1990. From climate to flow. In: Waggoner, P.E. (ed.). *Climate change and US water resources*. New York, USA: John Wiley, 177–206.
- Schneider, C., Laizé, C.L.R., Acreman, M.C. & Flörke, M. 2013. How will climate change modify river flow regimes in Europe? *Hydrology and Earth System Sciences*, **17**: 325–339.
- Sen, P.K. 1968. Estimates of the regression coefficient based on Kendall's Tau. *Journal of the American Statistical Association*, **63**: 1379–1389.
- Sharma, S.K., Kansal, M.L. & Tyagi, A. 2016. Integrated water management plan for Shimla City in India using geospatial techniques. *Water Supply*, **16**: 641–652.
- Shrestha, S., Bajracharya, A.R. & Babel, M.S. 2016. Assessment of risks due to climate change for the Upper Tamakoshi Hydropower Project in Nepal. *Climate Risk Management*, **14**: 27–41.
- Singh, V. & Goyal, M.K. 2017. Unsteady high velocity flood flows and the development of rating curves in a Himalayan basin under climate change scenarios. *Journal of Hydrologic Engineering*, **22**: 04017023.
- Slater, L.J., Khouakhi, A. & Wilby, R.L. 2019. River channel conveyance capacity adjusts to modes of climate variability. *Scientific Reports*, **9**: 12619.
- Slater, L.J. & Singer, M.B. 2013. Imprint of climate and climate change in alluvial riverbeds: Continental United States, 1950-2011. *Geology*, **41**: 595–598.
- Slavíková, L., Malý, V., Rost, M., Petružela, L. & Vojáček, O. 2013. Impacts of climate variables on residential water consumption in the Czech Republic. *Water Resources Management*, 27: 365–379.
- Slingo, J. 2014. The recent storms and floods in the UK. Exeter, England: Met Office.
- Staddon, C. & Scott, C.A. 2018. Putting water security to work: Addressing global challenges. *Water International*, **43**: 1017–1025.
- Stagl, J. & Hattermann, F. 2016. Impacts of climate change on riverine ecosystems: Alterations of ecologically relevant flow dynamics in the Danube River and its major tributaries. *Water*, **8**: 566.
- Stahl, K., Hisdal, H., Hannaford, J., Tallaksen, L.M., Lanen, H.A.J. van, Sauquet, E., Demuth, S., Fendekova, M. & Jódar, J. 2010. Streamflow trends in Europe: Evidence from a dataset of near-natural catchments. *Hydrology and Earth System Sciences*, **14**: 2367–2382.
- Suggitt, A., Maclean, I., Critchlow, R., Beale, C., Rowcroft, P. & White, C. 2015. Aggregate assessment of climate change impacts on the goods and benefits provided by the UK's natural assets. London, England: AECOM Limited.

- Suliman, A.H.A., Jajarmizadeh, M., Harun, S. & Mat Darus, I.Z. 2015. Comparison of semidistributed, GIS-based hydrological models for the prediction of streamflow in a large catchment. *Water Resources Management*, **29**: 3095–3110.
- Sultana, R. & Choi, M. 2018. Sensitivity of streamflow response in the snow-dominated Sierra Nevada watershed using projected CMIP5 data. *Journal of Hydrologic Engineering*, **23**: 05018015.
- SWWRA. 1965. *Licence 22/60/3/0035 to abstract water*. Llanelly, Wales: South West Wales River Authority.
- Tait, S.J., Ashley, R.M., Cashman, A., Blanksby, J. & Saul, A.J. 2008. Sewer system operation into the 21st century, study of selected responses from a UK perspective. *Urban Water Journal*, **5**: 79–88.
- Tena, T.M., Mwaanga, P. & Nguvulu, A. 2019. Impact of land use/land cover change on hydrological components in Chongwe River Catchment. *Sustainability*, **11**: 6415.
- Tetzlaff, D., Waldron, S., Brewer, M.J. & Soulsby, C. 2007. Assessing nested hydrological and hydrochemical behaviour of a mesoscale catchment using continuous tracer data. *Journal of Hydrology*, **336**: 430–443.
- Teutschbein, C., Grabs, T., Karlsen, R.H., Laudon, H. & Bishop, K. 2015. Hydrological response to changing climate conditions: Spatial streamflow variability in the boreal region. *Water Resources Research*, **51**: 9425–9446.
- Thober, S., Kumar, R., Wanders, N., Marx, A., Pan, M., Rakovec, O., Samaniego, L., Sheffield, J., Wood, E.F. & Zink, M. 2018. Multi-model ensemble projections of European river floods and high flows at 1.5, 2, and 3 degrees global warming. *Environmental Research Letters*, **13**: 014003.
- Thompson, J.R. 2012. Modelling the impacts of climate change on upland catchments in southwest Scotland using MIKE SHE and the UKCP09 probabilistic projections. *Hydrology Research*, **43**: 507–530.
- Thompson, J.R., Iravani, H., Clilverd, H.M., Sayer, C.D., Heppell, C.M. & Axmacher, J.C. 2017. Simulation of the hydrological impacts of climate change on a restored floodplain. *Hydrological Sciences Journal*, **62**: 2482–2510.
- Thompson, L.C., Escobar, M.I., Mosser, C.M., Purkey, D.R., Yates, D. & Moyle, P.B. 2012. Water management adaptations to prevent loss of spring-run Chinook Salmon in California under climate change. *Journal of Water Resources Planning and Management*, **138**: 465– 478.
- Toure, A., Diekkrüger, B., Mariko, A. & Cissé, A. 2017. Assessment of groundwater resources in the context of climate change and population growth: Case of the Klela Basin in southern Mali. *Climate*, **5**: 45.
- Tramblay, Y., Villarini, G. & Zhang, W. 2020. Observed changes in flood hazard in Africa. *Environmental Research Letters*, **15**: 1040b5.
- Trenberth, K. 2011. Changes in precipitation with climate change. *Climate Research*, **47**: 123–138.
- Tu, J. 2009. Combined impact of climate and land use changes on streamflow and water quality in eastern Massachusetts, USA. *Journal of Hydrology*, **379**: 268–283.

- UK CEH. 2012. *National changes in river flow*. [Online]. Available at: https://www.ceh.ac.uk/national-changes-river-flow#overview. Accessed: 5 March 2021.
- UN. 2015. *Transforming our world: The 2030 agenda for sustainable development*. New York, USA: United Nations.
- United Utilities. 2018. Final drought plan. Warrington, England: United Utilities.
- Valdivia-Garcia, M., Weir, P., Graham, D.W. & Werner, D. 2019. Predicted impact of climate change on Trihalomethanes formation in drinking water treatment. *Scientific Reports*, **9**: 9967.
- Van Vuuren, D.P., Edmonds, J., Kainuma, M., Riahi, K., Thomson, A., Hibbard, K., Hurtt, G.C., Kram, T., Krey, V., Lamarque, J.F., Masui, T., Meinshausen, M., Nakicenovic, N., Smith, S.J. & Rose, S.K. 2011. The Representative Concentration Pathways: An overview. *Climatic Change*, **109**: 5–31.
- Veettil, A.V. & Mishra, A.K. 2016. Water security assessment using blue and green water footprint concepts. *Journal of Hydrology*, **542**: 589–602.
- Vercruysse, K. & Grabowski, R.C. 2019. Temporal variation in suspended sediment transport: Linking sediment sources and hydro-meteorological drivers. *Earth Surface Processes and Landforms*, **44**: 2587–2599.
- Visser-Quinn, A., Beevers, L., Collet, L., Formetta, G., Smith, K., Wanders, N., Thober, S., Pan, M. & Kumar, R. 2019. Spatio-temporal analysis of compound hydro-hazard extremes across the UK. *Advances in Water Resources*, **130**: 77–90.
- Vonk, E., Xu, Y.P., Booij, M.J., Zhang, X. & M. Augustijn, D.C. 2014. Adapting multireservoir operation to shifting patterns of water supply and demand. *Water Resources Management*, 28: 625–643.
- Vörösmarty, C.J. 2000. Global water resources: Vulnerability from climate change and population growth. *Science*, **289**: 284–288.
- Vörösmarty, C.J., McIntyre, P.B., Gessner, M.O., Dudgeon, D., Prusevich, A., Green, P., Glidden, S., Bunn, S.E., Sullivan, C.A., Liermann, C.R. & Davies, P.M. 2010. Global threats to human water security and river biodiversity. *Nature*, **467**: 555–561.
- Wang, X., Li, Z. & Li, M. 2018. Impacts of climate change on stream flow and water quality in a drinking water source area, Northern China. *Environmental Earth Sciences*, **77**: 410.
- Warren, A.J. & Holman, I.P. 2012. Evaluating the effects of climate change on the water resources for the city of Birmingham, UK. *Water and Environment Journal*, **26**: 361–370.
- Watts, G. & Anderson, M. 2016. *Water climate change impacts report card 2016 edition*. London, England: Living with Environmental Change Network.
- Watts, G., Battarbee, R.W., Bloomfield, J.P., Crossman, J., Daccache, A., Durance, I., Elliott, J.A., Garner, G., Hannaford, J., Hannah, D.M., Hess, T., Jackson, C., Kay, A.L., Kernan, M., Knox, J.W., Mackay, J., Monteith, D.T., Ormerod, S.J., Rance, J., Stuart, M.E., Wade, A.J., Wade, S.D., Weatherhead, K., Whitehead, P.G. & Wilby, R.L. 2015a. Climate change and water in the UK Past changes and future prospects. *Progress in Physical Geography: Earth and Environment*, **39**: 6–28.

- Watts, G., Jenkins, A., Hess, T., Humble, A., Olbert, C., Kay, M., Pope, V., Stannard, T., Storey, M., Meacham, T., Benton, T. & Noble, A. 2015b. *Agriculture's impacts on water availability*. Wiltshire, England: Global Food Security.
- Webb, B.W., Clack, P.D. & Walling, D.E. 2003. Water-air temperature relationships in a Devon river system and the role of flow. *Hydrological Processes*, **17**: 3069–3084.
- Weigelhofer, G., Hein, T. & Bondar-Kunze, E. 2018. Phosphorus and nitrogen dynamics in riverine systems: Human impacts and management options. In: Schmutz, S. & Sendzimir, J. (eds.). *Riverine Ecosystem Management*. Cham, Switzerland: Springer International Publishing, 187–202.
- Weiler, M. & McDonnell, J.J. 2006. Testing nutrient flushing hypotheses at the hillslope scale: A virtual experiment approach. *Journal of Hydrology*, **319**: 339–356.
- Welsh Government. 2013. *Preparing for a changing climate. Part 1: Starting*. Cardiff, Wales: Welsh Government.
- Welsh Government. 2015a. *Woodlands for Wales indicators 2014-15. Statistics for Wales.* Cardiff, Wales: Welsh Government.
- Welsh Government. 2015b. Water strategy for Wales. Cardiff, Wales: Welsh Government.
- Welsh Government. 2019a. Prosperity for all: A climate conscious Wales. A climate change adaptation plan for Wales. Cardiff, Wales: Welsh Government.
- Welsh Government. 2019b. *Energy generation in Wales: 2018*. Cardiff, Wales: Welsh Government.
- Welsh Government. 2019c. June 2019 survey of agriculture and horticulture: Results for Wales. Cardiff, Wales: Welsh Government.
- Welsh Government. 2020. *Welcome to Wales: Priorities for the visitor economy 2020-2025*. Cardiff, Wales: Welsh Government.
- West, H., Quinn, N. & Horswell, M. 2019. Regional rainfall response to the North Atlantic Oscillation (NAO) across Great Britain. *Hydrology Research*, **50**: 1549–1563.
- Wheater, H.S. & Evans, E. 2009. Land use, water management and future flood risk. *Land Use Policy*, **26**: S251–S264.
- Whitehead, P., Butterfield, D. & Wade, A.J. 2008. *Potential impacts of climate change on river water quality*. Bristol, England: Environment Agency.
- Whitehead, P.G., Crossman, J., Balana, B.B., Futter, M.N., Comber, S., Jin, L., Skuras, D., Wade, A.J., Bowes, M.J. & Read, D.S. 2013. A cost-effectiveness analysis of water security and water quality: Impacts of climate and land-use change on the River Thames system. *Philosophical Transactions of the Royal Society A: Mathematical, Physical and Engineering Sciences*, **371**: 20120413.
- Whitehead, P.G., Wade, A.J. & Butterfield, D. 2009a. Potential impacts of climate change on water quality and ecology in six UK rivers. *Hydrology Research*, **40**: 113–122.
- Whitehead, P.G., Wilby, R.L., Battarbee, R.W., Kernan, M. & Wade, A.J. 2009b. A review of the potential impacts of climate change on surface water quality. *Hydrological Sciences Journal*, **54**: 101–123.

- Whitehead, P.G. & Williams, R.J. 1982. A dynamic nitrogen balance model for river systems. In: *Proceedings of the Exeter Symposium*. Wallingford, England: IAHS Publications, 89–89.
- Wilby, R.L. & Harris, I. 2006. A framework for assessing uncertainties in climate change impacts: Low-flow scenarios for the River Thames, UK. *Water Resources Research*, **42**: W02419.
- Wilby, R.L., Whitehead, P.G., Wade, A.J., Butterfield, D., Davis, R.J. & Watts, G. 2006. Integrated modelling of climate change impacts on water resources and quality in a lowland catchment: River Kennet, UK. *Journal of Hydrology*, **330**: 204–220.
- Wilkes, G., Edge, T.A., Gannon, V.P.J., Jokinen, C., Lyautey, E., Neumann, N.F., Ruecker, N., Scott, A., Sunohara, M., Topp, E. & Lapen, D.R. 2011. Associations among pathogenic bacteria, parasites, and environmental and land use factors in multiple mixed-use watersheds. *Water Research*, **45**: 5807–5825.
- Williams, J.R. & Izaurralde, R.C. 2010. The APEX Model. In: Singh, V.P. & Frevert, D.K. (eds.). *Watershed Models*. Boca Raton, USA: Taylor & Francis Group, 437–482.
- WMO. 2017. *WMO guidelines on the calculation of climate normals*. Geneva, Switzerland: World Meteorological Organization.
- WMO. 2020. *WMO statement on the state of the global climate in 2019*. Geneva, Switzerland: World Meteorological Organization.
- Worster, R.C. & Denny, D.F. 1955. Hydraulic transport of solid material in pipes. *Proceedings* of the Institution of Mechanical Engineers, **169**: 563–586.
- Xenochristou, M., Kapelan, Z. & Hutton, C. 2020. Using smart demand-metering data and customer characteristics to investigate influence of weather on water consumption in the UK. Journal of Water Resources Planning and Management, **146**: 04019073.
- Xia, X., Gui, L. & Zhan, Z.H. 2018. A multi-swarm particle swarm optimization algorithm based on dynamical topology and purposeful detecting. *Applied Soft Computing*, **67**: 126–140.
- Xu, X., Wang, Y.C., Kalcic, M., Muenich, R.L., Yang, Y.C.E. & Scavia, D. 2019. Evaluating the impact of climate change on fluvial flood risk in a mixed-use watershed. *Environmental Modelling & Software*, **122**: 104031.
- Yang, D., Herath, S. & Musiake, K. 2000. Comparison of different distributed hydrological models for characterization of catchment spatial variability. *Hydrological Processes*, 14: 403–416.
- Yang, M., Xiao, W., Zhao, Y., Li, X., Huang, Y., Lu, F., Hou, B. & Li, B. 2018. Assessment of potential climate change effects on the rice yield and water footprint in the Nanliujiang catchment, China. *Sustainability*, **10**: 242.
- Yang, X., Tan, L., He, R., Fu, G., Ye, J., Liu, Q. & Wang, G. 2017. Stochastic sensitivity analysis of nitrogen pollution to climate change in a river basin with complex pollution sources. *Environmental Science and Pollution Research*, 24: 26545–26561.
- Yates, D., Sieber, J., Purkey, D. & Huber-Lee, A. 2005. WEAP21 A demand-, priority-, and preference-driven water planning model. Part 1: Model characteristics. *Water International*, **30**: 487–500.

- Yuan, Z., Xu, J. & Wang, Y. 2019. Historical and future changes of blue water and green water resources in the Yangtze River source region, China. *Theoretical and Applied Climatology*, 138: 1035–1047.
- Zeiger, S. & Hubbart, J.A. 2016. Quantifying suspended sediment flux in a mixed-land-use urbanizing watershed using a nested-scale study design. *Science of The Total Environment*, **542**: 315–323.
- Zhang, X., Zwiers, F.W., Hegerl, G.C., Lambert, F.H., Gillett, N.P., Solomon, S., Stott, P.A. & Nozawa, T. 2007. Detection of human influence on twentieth-century precipitation trends. *Nature*, **448**: 461–465.
- Židonis, A., Benzon, D.S. & Aggidis, G.A. 2015. Development of hydro impulse turbines and new opportunities. *Renewable and Sustainable Energy Reviews*, **51**: 1624–1635.

### **APPENDICES**



### **Appendix 1: SWAT model calibration charts**

**Figure A1.1.** Observed streamflow (blue) compared to the best simulation (red) for the Clwyd catchment, following calibration for the period 1<sup>st</sup> January 1985 to 31<sup>st</sup> December 1998; KGE = 0.810.



**Figure A1.2.** Observed streamflow (blue) compared to the best simulation (red) for the Conwy catchment, following calibration for the period 1<sup>st</sup> January 1985 to 31<sup>st</sup> December 1998; KGE = 0.770.



**Figure A1.3.** Observed streamflow (blue) compared to the best simulation (red) for the Dyfi catchment, following calibration for the period  $1^{st}$  January 1985 to  $31^{st}$  December 1998; KGE = 0.788.
## Appendices



**Figure A1.4.** Observed streamflow (blue) compared to the best simulation (red) for the Teifi catchment, following calibration for the period  $1^{st}$  January 1985 to  $31^{st}$  December 1998; KGE = 0.851.



**Figure A1.5.** Observed streamflow (blue) compared to the best simulation (red) for the Tywi catchment, following calibration for the period 1<sup>st</sup> January 1985 to 31<sup>st</sup> December 1998; KGE = 0.841.



## Appendix 2: Ensemble projections for seasonal and annual streamflow

**Figure A2.6.** Seasonal and annual future streamflow projections for the Clwyd based on a 5-year moving mean; coloured lines represent individual model runs from the 12-member ensemble, black line is the ensemble mean.



**Figure A2.7.** Seasonal and annual future streamflow projections for the Conwy based on a 5-year moving mean; coloured lines represent individual model runs from the 12-member ensemble, black line is the ensemble mean.



**Figure A2.8.** Seasonal and annual future streamflow projections for the Dyfi based on a 5-year moving mean; coloured lines represent individual model runs from the 12-member ensemble, black line is the ensemble mean.



**Figure A2.9.** Seasonal and annual future streamflow projections for the Teifi based on a 5-year moving mean; coloured lines represent individual model runs from the 12-member ensemble, black line is the ensemble mean.



**Figure A2.10.** Seasonal and annual future streamflow projections for the Tywi based on a 5-year moving mean; coloured lines represent individual model runs from the 12-member ensemble, black line is the ensemble mean.